SECURITIES AND EXCHANGE COMMISSION

FORM 10-Q

Quarterly report pursuant to sections 13 or 15(d)

Filing Date: **2016-07-29** | Period of Report: **2016-06-30** SEC Accession No. 0001136261-16-000571

(HTML Version on secdatabase.com)

FILER

8X8 INC /DE/

CIK:1023731| IRS No.: 770142404 | State of Incorp.:DE | Fiscal Year End: 0331

Type: 10-Q | Act: 34 | File No.: 000-21783 | Film No.: 161791756 SIC: 4813 Telephone communications (no radiotelephone)

Mailing Address 2125 O'NEL DRIVE SAN JOSE CA 95131 Business Address 2125 O'NEL DRIVE SAN JOSE CA 95131 4087271885

UNITED STATES SECURITIES AND EXCHANGE COMMISSION Washington D.C. 20549

Washington, D.C. 20549	9
FORM 10-Q	_
[X] QUARTERLY REPORT PURSUANT TO SECTION 13 OR EXCHANGE ACT OF 1934	15(d) OF THE SECURITIES
For the quarterly period ended Ju	ne 30, 2016
OR	,
[] TRANSITION REPORT PURSUANT TO SECTION 13 OR : EXCHANGE ACT OF 1934	15(d) OF THE SECURITIES
For the transition period fromto	
<u>Commission file number 000-</u>	<u>-21783</u>
8x8, INC.	
(Exact name of Registrant as Specified in its Charter)	
Delaware	77-0142404
(State or Other Jurisdiction of Incorporation or Organization)	(I.R.S. Employer Identification Number)
2125 O'Nel Drive	
San Jose, CA 95131	
(Address of Principal Executive Offices) (408) 727-1885	
(Registrant's Telephone Number, including Area Code) Indicate by check mark whether the registrant (1) has filed all reports required to Exchange Act of 1934 during the preceding 12 months (or for such shorter period to has been subject to such filing requirements for the past 90 days. ☒ YES ☐ Indicate by check mark whether the registrant has submitted electronically and Interactive Data File required to be submitted and posted pursuant to Rule 405 of R	that the registrant was required to file reports), and (2 NO posted on its corporate Web site, if any, every

reporting company. See definitions of "large accelerated filer," "accelerated filer" and "smaller reporting company" in Rule 12b-2 of the Exchange Act. (Check one):

Large accelerated filer

Accelerated filer

Non-accelerated filer

Do not check if a smaller reporting company

Smaller reporting company

Indicate by check mark whether the registrant is a shell company (as defined in Rule 12b-2 of the Exchange Act). YES

Indicate by check mark whether the registrant is a large accelerated filer, an accelerated filer, a non-accelerated filer or a smaller

NO \boxtimes The number of shares of the Registrant's Common Stock outstanding as of July 25, 2016 was 89,680,865.

preceding 12 months (or for such shorter period that the registrant was required to submit and post such files).

NO □

Х

TABLE OF CONTENTS

PART I. FINANCIAL INFORMATION	Page No.
Item 1. Financial Statements:	
Condensed Consolidated Balance Sheets at June 30, 2016 and March 31, 2016	<u>2</u>
Condensed Consolidated Statements of Operations for the three months ended June 30, 2016 and 2015	<u>3</u>
Condensed Consolidated Statements of Comprehensive Income (Loss) for the three months ended June 30, 2016 and 2015	4
Condensed Consolidated Statements of Cash Flows for the three months ended June 30, 2016 and 2015	<u>5</u>
Notes to Unaudited Condensed Consolidated Financial Statements	<u>6</u>
Item 2. Management's Discussion and Analysis of Financial Condition and Results of Operations	<u>17</u>
Item 3. Quantitative and Qualitative Disclosures About Market Risk	<u>22</u>
Item 4. Controls and Procedures	<u>23</u>
PART II. OTHER INFORMATION	
Item 1. Legal Proceedings	<u>23</u>
Item 1A. Risk Factors	<u>24</u>
Item 2. Unregistered Sales of Equity Securities and Use of Proceeds	<u>24</u>
Item 5. Other Information	<u>24</u>
Item 6. Exhibits	<u>25</u>
Signature 1	<u>26</u>

Part I -- FINANCIAL INFORMATION ITEM 1. FINANCIAL STATEMENTS

8X8, Inc. CONDENSED CONSOLIDATED BALANCE SHEETS (In thousands, unaudited)

	June 30, 2016	March 31, 2016	
ASSETS		_	
Current assets:			
Cash and cash equivalents	\$27,466	\$33,576	
Short-term investments	139,607	129,274	
Accounts receivable, net	11,538	11,070	
Inventory	463	520	
Deferred cost of goods sold	717	634	
Deferred income taxes	-	5,382	
Other current assets	5,096	5,444	
Total current assets	184,887	185,900	
Property and equipment, net	13,015	12,375	
Intangible assets, net	19,531	21,464	
Goodwill	45,931	47,420	
Non-current deferred income taxes	48,532	43,189	
Other assets	3,751	3,104	
Total assets	\$ <u>315,647</u>	\$313,452	
LIABILITIES AND STOCKHOLDERS' EQUITY			
Current liabilities:			
Accounts payable	\$12,649	\$10,954	
Accrued compensation	9,631	10,063	
Accrued warranty	340	326	
Accrued taxes	5,390	5,200	
Accrued outside commissions	2,070	2,186	
Deferred revenue	2,120	1,925	
Other accrued liabilities	3,226	4,080	
Total current liabilities	35,426	34,734	
Non-current liabilities	3,107	3,258	
Non-current deferred revenue	134	154	
Total liabilities	38,667	38,146	
Commitments and contingencies (Note 6)			
Stockholders' equity:			
Common stock	89	89	
Additional paid-in capital	394,100	389,260	
Accumulated other comprehensive loss	(6,822)	(4,184)	
Accumulated deficit	(110,387)	(109,859)	
Total stockholders' equity	276,980	275,306	
Total liabilities and stockholders' equity	\$315,647	\$313,452	
The common successful and stockholders equity	C.1 1:4 1 1	1 1:1 / 1:	

8X8, Inc.

CONDENSED CONSOLIDATED STATEMENTS OF OPERATIONS

(In thousands, except per share amounts; unaudited)

Three Months Ended

	1	June 30,	
	2016	2015	
Service revenue	\$55,296	\$44,168	
Product revenue	4,745	3,724	
Total revenue	60,041	47,892	
Operating expenses:			
Cost of service revenue	10,235	8,459	
Cost of product revenue	5,505	4,382	
Research and development	6,710	5,080	
Sales and marketing	31,691	23,824	
General and administrative	6,801	6,068	
Total operating expenses	60,942	47,813	
Income (loss) from operations	(901)	79	
Other income, net	410	234	
Income (loss) before provision for income taxes	(491)	313	
Provision for income taxes	37	785	
Net loss	\$ <u>(528)</u>	\$ <u>(472)</u>	
Net loss per share:			
Basic	\$(0.01)	\$(0.01)	
Diluted	\$(0.01)	\$(0.01)	
Weighted average number of shares:			
Basic	89,434	88,233	
Diluted	89,434	88,233	

8X8, Inc. CONDENSED CONSOLIDATED STATEMENTS OF COMPREHENSIVE INCOME (LOSS) (In thousands, unaudited)

Three	Months	Ended
-------	--------	-------

	June 30,		
	2016	2015	
Net loss	\$ (528)	\$ (472)	
Other comprehensive income (loss), net of tax			
Unrealized gain (loss) on investments in securities	146	(48)	
Foreign currency translation adjustment	(2,784)	1,478	
Comprehensive income (loss)	\$ (3,166)	\$ 958	

8X8, Inc. CONDENSED CONSOLIDATED STATEMENTS OF CASH FLOWS (In thousands, unaudited)

	June 30,	
	2016	2015
Cash flows from operating activities:		
Net loss	\$ (528)	\$ (472)
Adjustments to reconcile net loss to net cash		
provided by operating activities:		
Depreciation	1,471	993
Amortization of intangible assets	960	546
Amortization of capitalized software	146	456
Net accretion of discount and amortization of		
premium on marketable securities	100	236
Stock-based compensation expense	5,051	3,022
Deferred income tax (benefit) provision	(44)	476
Other	190	74
Changes in assets and liabilities:		
Accounts receivable, net	(1,043)	(612)
Inventory	53	88
Other current and noncurrent assets	(508)	(470)
Deferred cost of goods sold	(120)	(53)
Accounts payable	1,137	933
Accrued compensation	(354)	725
Accrued warranty	14	3
Accrued taxes	240	492
Deferred revenue	211	(704)
Accrued outside commissions	(116)	199
Other current and noncurrent liabilities	(324)	(1,272)
Net cash provided by operating activities	6,536	4,660
Cash flows from investing activities:		
Purchases of property and equipment	(1,604)	(1,073)
Cost of capitalized software	(707)	(471)
Purchase of businesses, net of cash acquired	- -	(23,434)
Proceeds from maturity of investments	17,025	7,820
Sales of investments - available for sale	15,324	22,620
Purchase of investments - available for sale	(42,625)	(34,409)
Net cash used in investing activities	(12,587)	(28,947)
Cash flows from financing activities:		
Capital lease payments	(182)	(54)
Payment of contingent consideration and escrow	(200)	-
Repurchase of common stock	(629)	(25)
Proceeds from issuance of common stock under employee stock plans	1,039	336
Net cash provided by financing activities	28	257
Effect of exchange rate changes on cash	(87)	218
Net decrease in cash and cash equivalents	(6,110)	(23,812)
and account in our and our equivalents	(0,110)	(23,012)
Cash and cash equivalents, beginning of the period	33,576	53,110
Cash and cash equivalents, end of the period	\$ 27,466	\$ 29,298
Supplemental cash flow information		
Income taxes paid	\$ 87	\$ 52
Interest paid	7	6

8X8, Inc. NOTES TO UNAUDITED CONDENSED CONSOLIDATED

FINANCIAL STATEMENTS

1. DESCRIPTION OF BUSINESS AND SIGNIFICANT ACCOUNTING POLICIES

DESCRIPTION OF BUSINESS

8x8, Inc. (8x8 or the Company) is a provider of cloud-based, enterprise-class software solutions that transform the way businesses communicate and collaborate globally. The Company's integrated, "pure-cloud" offering combines global voice, conferencing, messaging and video with integrated workflows and big data analytics on a single platform to enable increased team productivity, better customer engagement and real-time insights into business performance.

BASIS OF PRESENTATION

The Company's fiscal year ends on March 31 of each calendar year. Each reference to a fiscal year in these notes to the consolidated financial statements refers to the fiscal year ended March 31 of the calendar year indicated (for example, fiscal 2017 refers to the fiscal year ended March 31, 2017).

The accompanying interim condensed consolidated financial statements are unaudited and have been prepared on substantially the same basis as our annual consolidated financial statements for the fiscal year ended March 31, 2016. In the opinion of the Company's management, these interim condensed consolidated financial statements reflect all adjustments (consisting only of normal recurring adjustments) considered necessary for a fair statement of our financial position, results of operations and cash flows for the periods presented. The preparation of financial statements in conformity with generally accepted accounting principles requires management to make estimates and assumptions that affect the reported amounts of assets and liabilities and disclosure of contingent assets and liabilities at the date of the condensed consolidated financial statements and the reported amounts of revenue and expenses during the reporting periods. Actual results could differ from these estimates.

The March 31, 2016 year-end condensed consolidated balance sheet data in this document were derived from audited consolidated financial statements and does not include all of the disclosures required by U.S. generally accepted accounting principles. These condensed consolidated financial statements should be read in conjunction with the Company's audited consolidated financial statements as of and for the fiscal year ended March 31, 2016 and notes thereto included in the Company's fiscal 2016 Annual Report on Form 10-K.

The results of operations and cash flows for the interim periods included in these condensed consolidated financial statements are not necessarily indicative of the results to be expected for any future period or the entire fiscal year.

PRINCIPLES OF CONSOLIDATION

The consolidated financial statements include the accounts of 8x8 and its subsidiaries. All material intercompany accounts and transactions have been eliminated.

SIGNIFICANT ACCOUNTING POLICIES

The significant accounting policies used in preparation of these condensed consolidated financial statements are disclosed in our Annual Report on Form 10-K for the fiscal year ended March 31, 2016 filed with the SEC on May 31, 2016, and there have been no changes to the Company's significant accounting policies during the three months ended June 30, 2016, except as described in the "Recently Adopted Accounting Pronouncements" section below.

RECENTLY ADOPTED ACCOUNTING PRONOUNCEMENTS

In April 2015, the Financial Accounting Standards Board (FASB) issued Accounting Standards Update (ASU) 2015-5, "Intangibles-Goodwill and Other-Internal-Use Software (Subtopic 350-40): Customer's Accounting for Fees Paid in a Cloud Computing Arrangement." This update provides guidance in evaluating whether a cloud computing arrangement includes a software license. If a cloud computing arrangement of the arrangement should be accounted for as an acquisition of a software license. If a cloud computing arrangement does not include a software license, the customer should account for the arrangement as a service contract. The guidance does not change generally accepted accounting principles for a customer's accounting for service contracts. This update is effective for annual periods, including interim periods within those annual periods, beginning after December 15, 2015. Therefore, the Company has prospectively adopted this new standard on April 1, 2016. The adoption of this standard did not have a material impact on our consolidated financial statements.

In November 2015, the FASB issued ASU 2015-17, *Income Taxes - Balance Sheet Classification of Deferred Taxes (Topic 740)*. This ASU requires all deferred tax liabilities and assets to be presented in the balance sheet as noncurrent. As permitted, the Company early adopted this standard prospectively during the quarter ended June 30, 2016. The adoption of this standard resulted in reclassifying current deferred income tax assets to noncurrent deferred income tax assets and current deferred income tax liabilities. No prior periods were retrospectively adjusted.

RECENT ACCOUNTING PRONOUNCEMENTS

In May 2014, the FASB issued ASU No. 2014-09, *Revenue from Contracts with Customers*, which requires an entity to recognize the amount of revenue to which it expects to be entitled for the transfer of promised goods or services to customers. The ASU will replace most existing revenue recognition guidance in U.S. GAAP when it becomes effective. The new standard will become effective for the Company on April 1, 2018. The standard permits the use of either the retrospective or cumulative effect transition method. The Company has not yet selected a transition method.

In April 2016, the FASB issued ASU No. 2016-10, *Revenue from Contracts with Customers (Topic 606) - Identifying Performance Obligations and Licensing*. The ASU amends the guidance in ASU 2014-09 related to identifying performance obligations and accounting for licenses of intellectual property.

In May 2016, the FASB issued ASU No. 2016-12, *Revenue from Contracts with Customers (Topic 606): Narrow-Scope Improvements and Practical Expedients*. The amendment provides clarifying guidance in certain narrow areas and adds some practical expedients. The Company is currently assessing the impact of this pronouncement and its amendments to its consolidated financial statements.

2. FAIR VALUE MEASUREMENTS

Cash, cash equivalents, and available-for-sale investments, and contingent consideration were (in thousands):

As of June 30, 2016	Amortized Costs	Gross Unrealized Gain	Gross Unrealized Loss	Estimated Fair Value	Cash and Cash Equivalents	Short-Term Investments
Cash	\$ 22,440	\$-	\$-	\$22,440	\$ 22,440	\$-
Level 1:	\$ 22,440	Φ-	3 -	\$ 22,440	\$ 22,440	φ-
Money market funds	5,026	_	_	5,026	5,026	_
Mutual funds	2,000	_	(175)	1,825	-	1,825
Subtotal	29,466		(175)	29,291	27,466	1,825
	29,400	-	(173)	29,291	27,400	1,023
Level 2:	15.660	0		15 ((0		15.660
Commercial paper Corporate debt	15,660 88.620	8 143	- (10)	15,668	-	15,668
•	,-		(10)	88,753	-	88,753
Municipal securities Asset backed securities	1,001	-	(1)	1,000	-	1,000
	27,349	32	(1)	27,380	-	27,380
Mortgage backed securities	1,989	-	(14)	1,975	-	1,975
Agency bond	2,000 1,000	5 1	-	2,005 1,001	-	2,005 1,001
International government securities		· 				
Subtotal	137,619	189	(26)	137,782		137,782
Total assets	\$ <u>167,085</u>	\$ 189	\$ <u>(201)</u>	\$ <u>167,073</u>	\$ 27,466	\$ 139,607
Level 3:						
Contingent consideration	\$	\$	\$ -	\$ 216	\$ -	\$
Total liabilities	\$ -	\$ -	\$-	\$216	\$ -	\$ -
As of March 31, 2016	Amortized Costs	Unrealized Gain	Unrealized Loss	Estimated Fair Value	Cash Equivalents	Short-Term Investments
Cash	\$ 18,596	\$-	\$-	\$ 18,596	\$ 18,596	\$ -
Level 1:	Ψ 10,000	Ψ	Ψ	\$ 10,000	ψ 10,2 × 0	Ψ
Money market funds	14,980	_	_	14,980	14,980	_
Mutual funds	2,000	_	(187)	1,813	-	1,813
Subtotal	35,576	-	(187)	35,389	33,576	1,813
Level 2:	50,070		(107)	30,500	33,670	1,015
Commercial paper	6,794	2	_	6,796	_	6,796
Corporate debt	85,164	78	(28)	85,214	_	85,214
Municipal securities	1,007	-	(1)	1,006	_	1,006
Asset backed securities	24,614	7	(11)	24,610	_	24,610
Mortgage backed securities	2,045	-	(17)	2,028	_	2,028
Agency bond	6,805	1	-	6,806	-	6,806
Agency bond	· · · · · · · · · · · · · · · · · · ·		_	1,001	-	1,001
	1,000	1	=			
International government securities		- <u>1</u> 89	(57)	127.461	-	127.461
International government securities Subtotal	127,429	1 89 \$ 89	(57) \$ (244)	127,461 \$ 162,850	- \$ 33 576	127,461 \$ 129 274
International government securities Subtotal Total assets		1 89 \$ 89	(57) \$(244)	127,461 \$ 162,850	\$ <u>33,576</u>	127,461 \$ 129,274
International government securities Subtotal Total assets Level 3:	127,429 \$ 163,005	\$ 89	\$ (244)	\$ 162,850		\$ 129,274
International government securities Subtotal	127,429	-			\$\frac{1}{33,576}\$\$ \$\frac{1}{5}\$ \$\frac{1}{5}\$	

Contractual maturities of investments as of June 30, 2016 are set forth below (in thousands):

Estilliateu
Fair Value
\$79,772
59,835
\$139,607

Contingent Consideration and Escrow Liability

The Company's contingent consideration liability and escrow liability, included in other accrued liabilities and noncurrent liabilities on the consolidated balance sheets, was associated with the Quality Software Corporation (QSC) acquisition made in the first quarter of fiscal 2016. Amounts held in escrow were measure at fair value using present value computations. The contingent consideration was measured at fair value using a probability weighted average of the potential payment outcomes that would occur should certain contract milestones be reached. There is no market data available to use in valuing the contingent consideration; therefore, the Company developed its own assumptions related to the achievement of the milestones to evaluate the fair value of the liability. As such, the contingent consideration is classified within Level 3 as described below.

The items are classified as Level 3 within the valuation hierarchy, consisting of contingent consideration and escrow liability related to the QSC acquisition, were valued based on an estimate of the probability of success of the milestones being achieved and present value computations, respectively. The table below presents a roll-forward of the contingent consideration and escrow liability valued using a Level 3 input (in thousands):

		Three Months Ended June 30, 2016	Year Ended March 31, 2016
Balance at beginning of period	\$3	341	\$-
Purchase price contingent consideration	-		541
Contingent consideration payments	<u>(</u>	125)	(200)
Balance at end of period	\$2	216	\$ <u>341</u>
3. INVENTORIES	_		
	_	June 30, 2016	March 31, 2016
Inventory (in thousands)			
Work-in-process	\$3	;	\$76
Finished goods	4	160	444
Total	9	163	\$ <u>520</u>

4. INTANGIBLE ASSETS AND GOODWILL

The carrying value of intangible assets consisted of the following (in thousands):

		June 30, 2016			March 31, 2016			
	Gross Carrying Amount	Accumulated Amortization	Net Carrying Amount	Gross Carrying Amount	Accumulated Amortization	Net Carrying Amount		
Technology	\$ 18,033	\$ (5,215)	\$ 12,818	\$ 18,640	\$ (4,622)	\$ 14,018		
Customer relationships	9,716	(5,215)	4,501	9,993	(4,847)	5,146		
Trade names/domains	2,117	-	2,117	2,205	-	2,205		
In-process research and development	95	_	95	95	_	95		
Total acquired identifiable intangible assets	\$ 29,961	\$ (10,430)	\$ <u>19,531</u>	\$ 30,933	\$ (9,469)	\$ 21,464		

At June 30, 2016, annual amortization of intangible assets, based upon our existing intangible assets and current useful lives, is estimated to be the following (in thousands):

	Amount
Remaining 2017	\$2,758
2018	3,411
2019	3,162
2020	3,162
2021	2,796
Thereafter	2,030
Total	\$ <u>17,319</u>

The following table provides a summary of the changes in the carrying amounts of goodwill by reporting segment (in thousands):

	Americas	Europe	Total
Balance as of March 31, 2016	\$25,729	\$21,691	\$47,420
Foreign currency translation	_	(1,489)	(1,489)
Balance as of June 30, 2016	\$25,729	\$20,202	\$45,931

5. RESEARCH, DEVELOPMENT AND SOFTWARE COSTS

In the first three months of fiscal 2017 and 2016, the Company expensed all research and development costs in accordance with ASC 985-20, Costs of Software to be Sold, Leased or Marketed (ASC 985-20).

The Company accounts for computer software developed or obtained for internal use in accordance with ASC 350-40, *Internal Use Software* (ASC 350-40). Capitalized costs are classified as either long-term assets or property and equipment on the consolidated balance sheets.

Other Long-Term Assets

In the first three months of fiscal 2017, the Company capitalized \$0.7 million as other long-term assets. In the first three months of fiscal 2016, the Company capitalized \$0.5 million as other long-term assets. At June 30, 2016 and March 31, 2016, total completed capitalized software development costs included in other long-term assets was approximately \$1.7 million and \$0, respectively. At June 30, 2016 and March 31, 2016, accumulated amortization costs related to capitalized software in other long term assets was approximately \$0.1 million and \$0, respectively.

Property and Equipment

In the first three months of fiscal 2017, the Company capitalized \$0.3 million as property and equipment. The Company did not capitalize any costs in property and equipment in accordance with ASC 350-40 in the first three months of fiscal 2016. At June 30, 2016 and March 31, 2016, total completed capitalized software costs included in property and equipment was approximately \$1.2 million. At June 30, 2016 and March 31, 2016, accumulated amortization costs related to capitalized software in property and equipment was approximately \$0.3 million and \$0.2 million, respectively.

6. COMMITMENTS AND CONTINGENCIES

Leases

The Company leases its headquarters facility in San Jose, California under an operating lease agreement that expires in October 2019. The lease is an industrial net lease with monthly base rent of \$130,821 for the first 15 months with a 3% increase each year thereafter, and requires us to pay property taxes, utilities and normal maintenance costs. The Company also leases facilities for office space under non-cancelable operating leases for its various domestic and international locations.

The Company has entered into a series of noncancelable capital lease agreements for office equipment bearing interest at various rates. Assets under capital lease at June 30, 2016 totaled \$1.6 million with accumulated amortization of \$0.6 million.

Guarantees

Indemnifications

In the normal course of business, the Company may agree to indemnify other parties, including customers, lessors and parties to other transactions with the Company, with respect to certain matters such as breaches of representations or covenants or intellectual property infringement or other claims made by third parties. These agreements may limit the time within which an indemnification claim can be made and the amount of the claim. In addition, the Company has entered into indemnification agreements with its officers and directors. It is not possible to determine the maximum potential amount of the Company's exposure under these indemnification agreements due to the limited history of prior indemnification claims and the unique facts and circumstances involved in each particular agreement. Historically, payments made by the Company under these agreements have not had a material impact on the Company's operating results, financial position or cash flows. Under some of these agreements, however, the Company's potential indemnification liability might not have a contractual limit.

Product Warranties

The Company accrues for the estimated costs that may be incurred under its product warranties upon revenue recognition. Changes in the Company's product warranty liability, which is included in cost of product revenues in the consolidated statements of income (loss), were as follows (in thousands):

Thusa Montha Endad

			ns Ended
		June 3	30,
	2016	_	2015
Balance at beginning of period	\$ 326	\$	339
Accruals for warranties	114		98
Settlements	(87)		(83)
Adjustments	(13)	_	(12)
Balance at end of period	\$ 340	\$	342

Minimum Third Party Customer Support Commitments

In the third quarter of 2010, the Company amended its contract with one of its third party customer support vendors containing a minimum monthly commitment of approximately \$0.4 million effective April 1, 2010. The agreement requires a 150-day notice to terminate. At June 30, 2016, the total remaining obligation upon a termination of the contract was \$2.2 million.

Minimum Third Party Network Service Provider Commitments

The Company has entered into contracts with multiple vendors for third party network service which expire on various dates in fiscal 2017 through 2020. At June 30, 2016, future minimum annual payments under these third party network service contracts were as follows (in thousands):

Year ending March 31:

Remaining 2017	\$1,961
2018	981
2019	32
2020	8
Total minimum payments	\$ <u>2,982</u>

Legal Proceedings

The Company, from time to time, is involved in various legal claims or litigation, including patent infringement claims that can arise in the normal course of the Company's operations. Pending or future litigation could be costly, could cause the diversion of management's attention and could upon resolution, have a material adverse effect on the Company's business, results of operations, financial condition and cash flows.

On February 22, 2011, the Company was named a defendant in Bear Creek Technologies, Inc. v. 8x8, Inc. et al., along with 20 other defendants. On August 17, 2011, the suit was dismissed without prejudice as to the Company under Rule 21 of the Federal Rules of Civil Procedure. On August 17, 2011, Bear Creek Technologies, Inc. refiled its suit against the Company in the United States District Court for the District of Delaware. Further, on November 28, 2012, the U.S. Patent & Trademark Office initiated a Reexamination proceeding with a Reexamination Declaration explaining that there is a substantial new question of patentability, based on four separate grounds and affecting each claim of the patent which is the basis for the complaint filed against us. On March 26, 2013, the USPTO issued a first Office Action in the Reexamination, with all claims of the '722 patent being rejected on each of the four separate grounds raised in the Request for Reexamination. On July 10, 2013, the Company filed an informational pleading in support of and joining a motion to stay the proceeding in the District Court; the District Court granted the motion on July 17, 2013, based on the possibility that at least one of the USPTO rejections will be upheld and considering the USPTO's conclusion that Bear Creek's patent suffers from a defective claim for priority. On March 24, 2014, the USPTO issued another Office Action in which the rejections of the claims were maintained. On August 15, 2014, the USPTO issued a Right of Appeal Notice, as the USPTO maintained all rejections of the patent claims.

On September 15, 2014, Bear Creek Technologies, Inc. filed a Notice of Appeal of this decision with the Patent Trial and Appeal Board. The case is currently on appeal. The Company believes that it has meritorious defenses to these claims and is presenting a vigorous defense, but the Company cannot estimate potential liability in this case at this early stage of litigation.

On April 16, 2015, the Company was named as a defendant in Slocumb Law Firm v. 8x8, Inc. The Slocumb Law Firm has alleged that it purchased certain business services from the Company that did not perform as advertised or expected, and has asserted causes of actions for fraud, breach of contract, violations of the Alabama Deceptive Trade Practices Act and negligence. On May 7, 2015, the Company filed a motion with the U.S. District Court for the Middle District of Alabama, seeking an order compelling the Slocumb Law Firm to arbitrate its claims against the Company in Santa Clara County, California pursuant to a clause mandating arbitration of disputes set forth in the terms and conditions to which Slocumb Law Firm agreed in connection with its purchase of business services from the Company. No briefing schedule or hearing date for the motion has been set as of this time. Discovery has not yet commenced in the case. The Company intends to vigorously defend against Slocumb Law Firm's claims.

State and Municipal Taxes

From time to time, the Company has received inquiries from a number of state and municipal taxing agencies with respect to the remittance of sales, use, telecommunications, excise, and income taxes. Several jurisdictions currently are conducting tax audits of the Company's records. The Company collects or has accrued for taxes that it believes are required to be remitted. The amounts that have been remitted have historically been within the accruals established by the Company.

7. STOCK-BASED COMPENSATION

The following table summarizes stock-based compensation expense (in thousands):

Three Months Ended

Weighted

	June 30,	
	2016	2015
Cost of service revenue	\$ 360	\$ 219
Cost of product revenue	-	-
Research and development	887	531
Sales and marketing	1,915	1,197
General and administrative	1,889	1,075
Total stock-based compensation expense related to employee		
stock options and employee stock purchases, pre-tax	5,051	3,022
Tax benefit	<u>-</u>	<u>-</u>
Stock-based compensation expense related to employee		
stock options and employee stock purchases, net of tax	\$ 5,051	\$ 3,022
	TT *, 4 .: *,	

Stock Options, Stock Purchase Right and Restricted Stock Unit Activity

Stock Option activity under all the Company's stock option plans for the three months ended June 30, 2016, is summarized as follows:

	Number of Shares	Average Exercise Price Per Share
Outstanding at March 31, 2016	4,793,266	\$6.29
Granted	53,976	11.31
Exercised	(191,855)	2.18
Canceled/Forfeited	(1)	2.58
Outstanding at June 30, 2016	4,655,386	\$6.52
Vested and expected to vest at June 30, 2016	4,655,386	\$6.52
Exercisable at June 30, 2016	3,014,307	\$5.45

Stock Purchase Right activity for the three months ended June 30, 2016 is summarized as follows:

	Number of	Weighted Average Grant-Date Fair Market	Weighted Average Remaining Contractual
	Shares	Value	Term (in Years)
Balance at March 31, 2016	82,171	\$6.30	0.76
Granted	-	-	
Vested	(9,475)	6.23	
Forfeited	_		
Balance at June 30, 2016	72,696	\$6.31 13	0.45

Restricted Stock Unit activity for the three months ended June 30, 2016 is summarized as follows:

	Number of Shares	Weighted Average Grant Date Fair Value	Weighted Average Remaining Contractual Term (in Years)
Balance at March 31, 2016	4,544,799	\$8.08	1.67
Granted	300,942	11.40	
Vested	(311,949)	8.02	
Forfeited	(101,193)	8.48	
Balance at June 30, 2016	4,432,599	\$8.32	1.53

The following table summarizes stock options outstanding and exercisable at June 30, 2016:

	Options Outstanding			Options Exercisa	ble		
		Weighted Average	Weighted Average			Weighted Average	
	Shares	Exercise Price Per Share	Remaining Contractual Life (Years)	Aggregate Intrinsic Value	Shares	Exercise Price Per Share	Aggregate Intrinsic Value
\$0.55 to \$1.27	933,625	\$1.12	1.5	\$12,596,169	933,625	\$1.12	\$12,596,169
\$1.39 to \$6.86	1,477,028	\$5.83	6.8	12,968,088	1,050,590	\$5.43	9,643,718
\$7.52 to \$8.93	973,650	\$8.28	8.8	6,165,143	271,615	\$8.22	1,735,944
\$9.21 to \$9.74	983,835	\$9.62	7.2	4,905,183	666,290	\$9.63	3,319,570
\$10.50 to \$11.31	287,248	\$11.02	8.5	1,031,611	92,187	\$11.10	323,576
	4,655,386	_		\$37,666,194	3,014,307	_	\$27,618,977

As of June 30, 2016, there was \$32.9 million of unamortized stock-based compensation expense related to unvested stock options and awards which is expected to be recognized over a weighted average period of 2.22 years.

Unamortized stock-based compensation expense related to shares issued as part of an acquisition was approximately \$1.8 million, which will be recognized over a weighted average period of 2.92 years.

Assumptions Used to Calculate Stock-Based Compensation Expense

The fair value of each of the Company's option grants has been estimated on the date of grant using the Black-Scholes pricing model with the following assumptions:

	Three Months Ended June 30,	
	2016	2015
Expected volatility	51%	53%
Expected dividend yield	=	-
Risk-free interest rate	1.40%	1.59%
Weighted average expected option term	5.75 years	5.25 years
Weighted average fair value of options granted	\$5.45	\$4.17
	14	

Stock Repurchases

In February 2015, the Company's board of directors authorized the Company to purchase up to \$20.0 million of its common stock from time to time until February 29, 2016 (the "2015 Repurchase Plan"). Share repurchases, if any, will be funded with available cash. Repurchases under the 2015 Repurchase Plan may be made through open market purchases at prevailing market prices or in privately negotiated transactions. The timing, volume and nature of share repurchases are subject to market prices and conditions, applicable securities laws and other factors, and are at the discretion of the Company's management. Share repurchases under the 2015 Repurchase Plan may be commenced, suspended or discontinued at any time. This tranche of shares authorized for repurchase expired in February 2016.

In October 2015, the Company's board of directors authorized the Company to purchase an additional \$15.0 million of its common stock from time to time until October 20, 2016 under the 2015 Repurchase Plan. There were no stock repurchases for the period ended June 30, 2016. The remaining authorized repurchase amount at June 30, 2016 was \$15.0 million.

8. INCOME TAXES

For the three months ended June 30, 2016, the Company recorded a provision for income taxes of \$37,000, which was primarily attributable to income from operations. For the three months ended June 30, 2015, the Company recorded a provision for income taxes of \$0.8 million which was primarily attributable to income from operations.

The effective tax rate is calculated by dividing the income tax provision by net income before income tax expense.

At March 31, 2016, the Company had a liability for unrecognized tax benefits of \$2.9 million, all of which, if recognized, would decrease the company's effective tax rate. The Company does not believe that there has been any significant change in the unrecognized tax benefits in the three-month period ended June 30, 2016, and does not expect the remaining unrecognized tax benefit to change materially in the next 12 months. To the extent that the remaining unrecognized tax benefits are ultimately recognized, they will have an impact on the effective tax rate in future periods.

The Company is subject to taxation in the U.S., California and various other states and foreign jurisdictions in which it has or had a subsidiary or branch operations or it is collecting sales tax. All tax returns from fiscal 2013 to fiscal 2016 may be subject to examination by the Internal Revenue Service, California and various other states. As of July 22, 2016, there were no active federal or state income tax audits. Returns filed in foreign jurisdictions may be subject to examination for the fiscal years 2011 to 2016.

9. NET INCOME (LOSS) PER SHARE

The following is a reconciliation of the weighted average number of common shares outstanding used in calculating basic and diluted net income (loss) per share (in thousands, except share and per share data):

Three Months Ended

	1111	June 30,
	2016	2015
Numerator:		
Net loss available to common stockholders	\$ (528)	\$ (472)
Denominator:		
Common shares	89,434	88,233
Denominator for basic calculation	89,434	88,233
Employee stock options	-	-
Stock purchase rights	<u>-</u>	<u> </u>
Denominator for diluted calculation	89,434	88,233
Net loss per share		
Basic	\$ (0.01)	\$ (0.01)
Diluted	\$ (0.01)	\$ (0.01)

The following shares attributable to outstanding stock options and restricted stock purchase rights were excluded from the calculation of diluted earnings per share because their inclusion would have been antidilutive (in thousands):

		Months Ended une 30,
	2016	2015
Employee stock options	1,528	2,447
Stock purchase rights	1,602	70
Total anti-dilutive employee stock-based securities	3,130	2,517

10. SEGMENT REPORTING

ASC 280, Segment Reporting, establishes annual and interim reporting standards for an enterprise's business segments and related disclosures about its products, services, geographic areas and major customers. Under ASC 280, the method for determining what information to report is based upon the way management organizes the operating segments within the Company for making operating decisions and assessing financial performance.

The Company manages its operations primarily on a geographic basis. The Chief Executive Officer, the Chief Financial Officer, and the Chief Technology Officer or the Company's Chief Operating Decision Makers (CODMs), evaluate performance of the Company and make decisions regarding allocation of resources based on geographic results. The Company's reportable segments are the Americas and Europe. The Americas segment is primarily North America. The Europe segment is primarily the United Kingdom. Each operating segment provides similar products and services.

The Company's CODMs evaluate the performance of its operating segments based on revenues and net income. Revenues are attributed to each segment based on the ordering location of the customer or ship to location. The Company allocates corporate overhead costs such as research and development, sales and marketing, general and administrative, amortization expense, stock-based compensation expense, and commitment and contingencies to the Americas segment.

The Company's revenue distribution by geographic region (based upon the destination of shipments and the customer's service address) was as follows:

Three Months Ended

Three Months Ended

	June 50,			
	2016	2015		
Americas (principally US)	88%	88%		
Europe	11%	9%		
Asia Pacific	1%	3%		
	100%	100%		

Geographic area data is based upon the location of the property and equipment and is as follows (in thousands):

	2016	2016
Americas (principally US)	\$ 9,643	\$ 9,165
Europe	3,126	2,642
Asia-Pacific	246	568
Total	\$ 13,015	\$ 12,375

The following table provides financial information by segment for the three month period ending June 30, 2016 and 2015 (in thousands):

	i nree Months Ended			
	June 30,			
	2016		2015	
Americas (principally US):				
Net revenue	\$ 53,398	\$	43,588	
Net income	\$ 1,466	\$	251	
Europe:				
Net revenue	\$ 6,643	\$	4,304	
Net loss	\$ (1,994)	\$	(723)	

ITEM 2. MANAGEMENT'S DISCUSSION AND ANALYSIS OF FINANCIAL CONDITION AND RESULTS OF OPERATIONS

FORWARD-LOOKING STATEMENTS

This Management Discussion and Analysis of Financial Condition and Results of Operations contains forward-looking statements within the meaning of Section 27A of the Securities Act and Section 21E of the Exchange Act. Any statements contained herein that are not statements of historical fact may be deemed to be forward-looking statements. For example, words such as "may," "will," "should," "estimates," "predicts," "potential," "continue," "strategy," "believes," "anticipates," "plans," "expects," "intends," and similar expressions are intended to identify forward-looking statements. You should not place undue reliance on these forward-looking statements. Actual results and trends may differ materially from historical results or those projected in any such forward-looking statements depending on a variety of factors. These factors include, but are not limited to, customer acceptance and demand for our cloud communications and collaboration services, the quality and reliability of our services, the prices for our services, customer renewal rates, customer acquisition costs, our ability to compete effectively in the hosted telecommunications and cloud-based computing services business, actions by our competitors, including price reductions for their competitive services, our ability to provide cost-effective and timely service and support to larger distributed enterprises, the impact of risks associated with our international operations, potential federal and state regulatory actions, compliance costs, potential warranty claims and product defects, our need for and the availability of adequate working capital, our ability to innovate technologically, the timely supply of products

by our contract manufacturers, our management's ability to execute its plans, strategies and objectives for future operations, including the execution of integration plans, and to realize the expected benefits of our acquisitions, and potential future intellectual property infringement claims and other litigation that could adversely affect our business and operating results. All forward-looking statements included in this report are based on information available to us on the date hereof, and we assume no obligation to update any such forward-looking statements. In addition to the factors discussed elsewhere in this Form 10-Q, see the Risk Factors discussion in Item 1A of our 2016 Form 10-K. The forward-looking statements included in this Form 10-Q are made only as of the date of this report, and we undertake no obligation to update the forward-looking statements to reflect subsequent events or circumstances.

BUSINESS OVERVIEW

We provide cloud-based, enterprise-class software solutions that transform the way businesses communicate and collaborate globally. Delivered through a SaaS (Software as a Service) business model, our solutions are at the forefront of a disruptive technology shift that is occurring in business communications where enterprises are increasingly replacing costly and unwieldy on-premises communications equipment with agile, cloud-based software services delivered over the public Internet.

Our integrated, "pure-cloud" offering combines global voice, conferencing, messaging and video with integrated workflows and big data analytics on a single platform to enable increased team productivity, better customer engagement and real-time insights into business performance. Through a combination of open API's (application program interface) and pre-built integrations, our solutions seamlessly leverage critical customer context from internal data systems and industry-leading Customer Relationship Management (CRM) systems, including cloud-based solutions from Salesforce.com, NetSuite, and Zendesk.

Powered by internally owned and managed technologies, our cloud communications and contact center offerings are uniquely positioned to serve businesses of all sizes and can easily scale to large, globally distributed enterprise customers. Our turnkey solution spans the breadth of communications and collaboration needs, is provided with industry-leading reliability at an affordable cost and is quick and easy-to deploy through our proprietary deployment methodology. This allows customers to focus on their business instead of managing the complexities of disparate communications and collaboration platforms and the integration of these platforms with other cloud-based business applications.

Our fiscal year ends on March 31 of each calendar year. Each reference to a fiscal year in this report refers to the fiscal year ending March 31 of the calendar year indicated (for example, fiscal 2017 refers to the fiscal year ending March 31, 2017).

SUMMARY AND OUTLOOK

In the first quarter of fiscal 2017, our service revenue from mid-market and enterprise customers grew 44% year-over year and represented 52% of total service revenue. New monthly recurring revenue (MRR) bookings from mid-market and enterprise customers and by our channel sales teams increased 62% year-over-year, reflecting strong demand for our services in our target market segments. Also, average monthly service revenue per business customer (ARPU) increased 13% to a record \$399, compared with \$353 in the same period last year. Our ability to offer a broad range of cloud- based mission critical communications services is bringing us larger deals where we continue to displace incumbent, premises-based systems.

Building upon our Global Reach® initiative, on July 19, 2016, we announced plans to expand our international footprint with new investments in Europe, Asia Pacific, and Latin America (LATAM) to support growing demand from global and distributed mid-market and enterprise customers. These include the addition of three new co-located data centers in Singapore, The Netherlands and Brazil, full localization capabilities for an additional six languages - French (two dialects), German, Dutch, Spanish, Italian and Portuguese - and new support centers in the Philippines and Romania delivering 24 x 7 follow-the-sun support. With this expansion, 8x8 will have a total of 12 data centers in eight regions of the world and will be one of the first global cloud communications providers with a presence in LATAM

We continued the advancement of our technology and service offerings in the first quarter of fiscal 2017 with the release of our new Virtual Office Meeting service - a high definition video conferencing and collaboration solution that enables secure, continuous communication from any device, anywhere in the world. This enables instant, seamless collaboration from a single platform with the ability to share content from cloud applications such as Dropbox, Box, iCloud and Google Drive.

CRITICAL ACCOUNTING POLICIES & ESTIMATES

The discussion and analysis of our financial condition and results of operations are based upon our condensed consolidated financial statements, which have been prepared in accordance with accounting principles generally accepted in the United States. The preparation of these financial statements requires us to make estimates and judgments that affect the reported amounts of assets, liabilities, revenue and expenses, and related disclosure of assets and liabilities. On an on-going basis, we evaluate our critical accounting policies and estimates. We base our estimates on historical experience and on various other assumptions that we believe to be reasonable under the circumstances, the results of which form the basis for making judgments about the carrying values of assets and liabilities that are not readily apparent from other sources. Actual results may differ from these estimates under different assumptions or conditions.

RECENTLY ADOPTED ACCOUNTING PRONOUNCEMENTS

See Item 1 of Part I, "Financial Statements - Note 1 - Basis of Presentation - Recent Adopted Accounting Pronouncements."

RECENT ACCOUNTING PRONOUNCEMENTS

See Item 1 of Part I, "Financial Statements - Note 1 - Basis of Presentation - Recent Accounting Pronouncements."

SELECTED OPERATING STATISTICS

We periodically review certain key business metrics, within the context of our articulated performance goals, in order to evaluate the effectiveness of our operational strategies, allocate resources and maximize the financial performance of our business. The selected operating statistics include the following:

Salastad Onovating Statistics

	Selected Operating Statistics					
	June 30,	March 31,	Dec. 31,	Sept. 30,	June 30,	
	2016	2016	2015	2015	2015	
Business customers average monthly						
service revenue per customer (1)	\$ 399	\$ 385	\$ 369	\$ 360	\$ 353	
Monthly business service revenue churn (2)(3)	0.5%	0.4%	1.2%	0.7%	1.0%	
Overall service margin	81%	81%	80%	80%	81%	
Overall product margin	-16%	-18%	-21%	-15%	-18%	
Overall gross margin	74%	72%	72%	73%	73%	

⁽¹⁾ Business customer average monthly service revenue per customer is service revenue from business customers in the period divided by the number of months in the period divided by the simple average number of business customers during the period.

RESULTS OF OPERATIONS

The following discussion should be read in conjunction with our condensed consolidated financial statements and the notes thereto.

	J	une 30,	Dollar	Percent
Service revenue	2016	2015	Change	Change
		thousands)	_	
Three months ended	\$ 55,296	\$ 44,168	\$ 11,128	25.2%
Percentage of total revenue	92.1%	92.2%		

Service revenue consists primarily of revenue attributable to the provision of our 8x8 cloud communication and collaboration services. We expect that cloud software solutions service revenues will continue to comprise nearly all of our service revenues for the foreseeable future.

8x8 service revenues increased in the first quarter of fiscal 2017 compared with the first quarter of the previous fiscal year primarily due to an increase in our business customer subscriber base (net of customer churn), and an increase in the average monthly service revenue per customer. Average monthly service revenue per customer increased from \$353 at June 30, 2015 to \$399 at June 30, 2016. We expect growth in the number of business customers and average monthly service revenue per customer to continue in fiscal 2017.

We translate revenue denominated in foreign currency into U.S. dollars for our financial statements. If the exchange rate for British Pound Sterling, or GBP, to U.S. dollars, or USD, persists at current levels, or declines further, our revenues will be adversely impacted due to the foreign currency translation adjustment. We cannot ascertain how much impact this will have in fiscal 2017, but at the current exchange rate, we estimate a negative impact on revenues of approximately \$0.6 million per quarter in comparison with the exchange rate for GBP to USD effective in the first quarter of fiscal 2017.

⁽²⁾ Business customer service revenue churn is calculated by dividing the service revenue lost from business customers (after the expiration of 30-day trial) during the period by the simple average of business customer service revenue during the same period and dividing the result by the number of months in the period.

⁽³⁾ Excludes DXI business customer service revenue churn for all periods presented.

	J	une 30,	Dollar	Percent
Product revenue	2016	2015	Change	Change
Three months ended	\$ 4,745	\$ 3,724	\$ 1,021	27.4%
Percentage of total revenue	7.9%	7.8%		

Product revenue consists primarily of revenue from sales of IP telephones in conjunction with our 8x8 cloud communication service. Product revenue increased for the three months ended June 30, 2016 primarily due to an increase in equipment sales to business customers.

No customer represented greater than 10% of the Company's total revenues for the three months ended June 30, 2016 or 2015.

	Ju	ine 30,	Dollar	Percent
Cost of service revenue	2016	2015	Change	Change
	(d			
Three months ended	\$ 10,235	\$ 8,459	\$ 1,776	21.0%
Percentage of service revenue	18.5%	19.2%		

The cost of service revenue primarily consists of costs associated with network operations and related personnel, communication origination and termination services provided party carriers, technology licenses, and royalty expenses.

Cost of service revenue for the three months ended June 30, 2016 increased over the comparable period in the prior fiscal year primarily due to a \$0.9 million increase in third party network services expenses, a \$0.4 million increase in amortization expense, a \$0.2 million increase in payroll and related expenses, a \$0.2 million increase in depreciation expense, and a \$0.1 million increase in license and fee expenses.

		June 30,	Dollar	Percent	
Cost of product revenue	2016	2015	Change	Change	
	(dollar amounts in thousands)				
Three months ended	\$ 5,505	\$ 4,382	\$ 1,123	25.6%	
Percentage of product revenue	116.0%	117.7%			

The cost of product revenue consists primarily of IP Telephones, estimated warranty obligations and direct and indirect costs associated with product purchasing, scheduling, shipping and handling. The amount of revenue allocated to product revenue based on relative selling price under our customer subscription agreements is less than the cost of the IP phone equipment.

The cost of product revenue for the three months ended June 30, 2016 increased over the comparable period in the prior fiscal year primarily due to an increase in equipment shipped to customers. The decrease in negative margin is due to less discounting of equipment in the current period.

	June 30,			Dollar	Percent
Research and development		2016 2015 Change		Change	
	(dollar amounts in thousands)				
Three months ended	\$ 6,710	\$	5,080	\$ 1,630	32.1%
Percentage of total revenue	11.29	6	10.6%		

Historically, our research and development expenses have consisted primarily of personnel, system prototype design, and equipment costs necessary for us to conduct our development and engineering efforts. During the three months ended June 30, 2016, we expensed all research and development costs as they were incurred in accordance with ASC 985-20.

The research and development expenses for the three months ended June 30, 2016 increased over the comparable period in the prior fiscal year primarily due to a \$1.5 million increase in payroll and related costs, a \$0.4 million increase in stock-based compensation costs, a \$0.3 million increase in consulting, temporary personnel, and outside service expenses, offset by a \$0.7 million of payroll and related costs, consulting and outside services capitalized in accordance with ASC 350-40. Also, for the three months ended June 30, 2016, the DXI acquisition and our Romanian subsidiary increased total research and development costs by \$0.5 million and \$0.6 million, respectively, compared to the three months ended June 30, 2015. We expect research and development expenses to increase for the foreseeable future as we continue to invest in our DXI unit and in the formation of our research and development team in Romania.

	Jui	Dollar	Percent	
Sales and marketing	2016	2015	Change	Change
	(do			
Three months ended	\$ 31,691	\$ 23,824	\$ 7,867	33.0%
Percentage of total revenue	52.8%	49.7%		

Sales and marketing expenses consist primarily of personnel and related overhead costs for sales, marketing, and customer service which includes deployment engineering. Such costs also include outsourced customer service call center operations, sales commissions, as well as trade show, advertising and other marketing and promotional expenses.

Sales and marketing expenses for the first quarter of fiscal 2017 increased over the same quarter in the prior fiscal year primarily due to a \$4.6 million increase in payroll and related costs, a \$0.7 increase in stock-based compensation costs, a \$0.6 million increase in indirect channel commission expenses, a \$0.3 million increase in travel costs, and a \$0.3 million increase in recruiting expenses. Also, for the three months ended June 30, 2016, the DXI acquisition increased total sales and marketing costs by \$1.1 million, compared to the three months ended June 30, 2015. We expect sales and marketing expenses to increase for the foreseeable future as we continue to invest in our US sales and marketing efforts and our DXI unit.

	June 30,			Percent
General and administrative	2016	2015	Change	Change
	(dollar amounts in thousands)			
Three months ended \$	6,801 \$	6,068	\$ 733	12.1%
Percentage of total revenue	11.3%	12.7%		

General and administrative expenses consist primarily of personnel and related overhead costs for finance, human resources and general management.

General and administrative expenses for the first quarter of fiscal 2017 increased over the same quarter in the prior fiscal year primarily due to a \$0.8 million increase in stock-based compensation costs, a \$0.3 million increase in facility lease expenses, and a \$0.1 million increase in recruiting expenses, offset by a \$0.5 million decrease in legal fees, which were primarily related to business acquisitions costs that occurred during the first fiscal quarter of 2015.

		June 30,	Dollar	Percent
Other income, net	2016	2015	Change	Change
Three months ended	\$ 410	\$ 234	\$ 176	75.2%
Percentage of total revenue	0.7%	0.5%		

Other income, net, primarily consisted of interest income earned on our cash, cash equivalents and investments and amortization or accretion of investments in fiscal 2017 and 2016.

		June 30,	Dollar	Percent
Provision (benefit) for income tax	2016	2015	Change	Change
		(dollar amounts in the	housands)	<u> </u>
Three months ended	\$ 37	\$ 785	\$ (748)	-95.3%
Percentage of income				
before provision for income taxes	-7 5%	250.8%		

For the three months ended June 30, 2016, we recorded a provision for income taxes of \$37,000, all of which related to net income (loss) from operations. For the three months ended June 30, 2015, we recorded a provision for income taxes of \$0.8 million, all of which related to net income (loss) from operations.

The effective tax rate set forth in the previous table is calculated by dividing the income tax provision by net income before income tax expense. We estimate our annual effective tax rate at the end of each quarter. In estimating the annual effective tax rate, we, in consultation with our tax advisors, consider, among other things, annual pre-tax income, permanent tax differences, the geographic mix of pre-tax income and the application and interpretations of existing tax laws.

We estimate our annual effective rate at the end of each quarterly period, and we record the tax effect of certain discrete items, which are unusual or occur infrequently, in the interim period in which they occur, including changes in judgment about deferred tax valuation allowances. The determination of the effective tax rate reflects tax expense and benefit generated in certain domestic and foreign jurisdictions. However, jurisdictions with a year-to-date loss where no tax benefit can be recognized are excluded from the annual effective tax rate.

Liquidity and Capital Resources

As of June 30, 2016, we had approximately \$167.1 million in cash, cash equivalents and short-term investments.

Net cash provided by operating activities for the three months ended June 30, 2016 was approximately \$6.5 million, compared with \$4.7 million for the three months ended June 30, 2015. Cash provided by operating activities has historically been affected by the amount of net income (loss), sales of subscriptions, changes in working capital accounts particularly in deferred revenue due to timing of annual plan renewals, add-backs of non-cash expense items such as the use of deferred tax assets, depreciation and amortization and the expense associated with stock-based awards.

The net cash used in investing activities for the three months ended June 30, 2016 was \$12.6 million, during which we purchased approximately \$10.3 million of short term investments, net of sales and maturities of short term investments, and we spent approximately \$1.6 million on the purchase of property and equipment. Net cash used in investing activities was approximately \$28.9 million, during the three months ended June 30, 2015, during which we spent approximately \$1.1 million on the purchase of property and equipment, spent approximately \$23.4 million on acquisitions of two businesses, and purchased approximately \$4.0 million of short term investments, net of sales and maturities of short term investments.

Net cash provided by financing activities for the three months ended June 30, 2016 was approximately \$28,000, which primarily resulted from \$1.0 million of cash received from the issuance of common stock under our employee stock purchase plan, reduced by \$0.6 million of repurchases of our common stock related to shares withheld for payroll taxes, \$0.2 million of payments of contingent consideration and escrow, and \$0.2 million of payments on capital leases. Net cash provided by financing activities for the three months ended June 30, 2015 was approximately \$0.3 million, which primarily resulted from cash received from the issuance of common stock under our employee stock purchase plan.

Contractual Obligations

Except as set forth below, there were no significant changes in our commitments under contractual obligations, as disclosed in the Company's Annual Report on Form 10-K, for the year ended June 30, 2016.

In June 2016, we entered into a new lease in London UK for our DXI location for approximately 16,000 square feet under an operating lease that expires in June 2026. We received an 18 month rent holiday from rent payments. After the rent holiday, the lease has a base monthly rent of approximately \$98,000, and requires us to pay service charges and normal maintenance costs. The lease contains a break clause, which allows us to end the lease in June 2022, subject to certain conditions.

ITEM 3. QUANTITATIVE AND QUALITATIVE DISCLOSURES ABOUT MARKET RISK

Foreign Currency

Our financial market risk consists primarily of risks associated with international operations and related foreign currencies. We derive a portion of our revenue from customers in Europe and Asia. In order to reduce the risk from fluctuation in foreign exchange rates, the vast majority of our sales are denominated in U.S. dollars. In addition, almost all of our arrangements with our contract manufacturers are denominated in U.S. dollars. We have not entered into any currency hedging activities.

We translate revenue denominated in foreign currency into U.S. dollars for our financial statements. During periods of a strengthening dollar, our reported European revenue is reduced because foreign currencies translate into fewer U.S. dollars. However, our UK segments are currently in a net loss position. Therefore, during periods of a strengthening dollar, our net loss from our UK segment would be reduced as well.

To date, our exposure to exchange rate volatility has not been significant. However, the June 2016 vote on a referendum to exit the European Union decision has resulted in a steep decline in the exchange rate for GBP to USD. The impact of Brexit to our results of operations for the period ended June 30, 2016 was not material. However, there can be no assurance that there will not be a material impact in the future.

Investments

The primary objective of our investment activities is to preserve principal while maximizing income without significantly increasing risk. Some of the securities in which we invest may be subject to market risk. This means that a change in prevailing interest rates may cause the principal amount of the investment to fluctuate. To minimize this risk, we may maintain our portfolio of cash equivalents and investments in a variety of securities, including commercial paper, money market funds, debt securities and certificates of deposit. The risk associated with fluctuating interest rates is limited to our investment portfolio and we do not believe that a 10% change in interest rates would have a significant impact on our interest income.

ITEM 4. CONTROLS AND PROCEDURES

Evaluation of Effectiveness of Disclosure Controls and Procedures

We maintain disclosure controls and procedures as defined in Rule 13a-15(e) and 15d-15(e) under the Securities Exchange Act of 1934 (Disclosure Controls) that are designed to ensure that information we are required to disclose in reports filed or submitted under the Securities and Exchange Act of 1934 is accumulated and communicated to management, including our principal executive and principal financial officers, as appropriate, to allow timely decisions regarding required disclosure, and that such information is recorded, processed, summarized and reported within the time periods specified in Securities and Exchange Commission rules and forms. As of the end of the period covered by this Quarterly Report on Form 10-Q, under the supervision of our Chief Executive Officer and our Chief Financial Officer, we evaluated the effectiveness of our Disclosure Controls. Based on this evaluation, our Chief Executive Officer and our Chief Financial Officer have concluded that our Disclosure Controls were effective as of June 30, 2016.

Limitations on the Effectiveness of Controls

Our management, including the Chief Executive Officer and Chief Financial Officer, do not expect that our Disclosure Controls or internal control over financial reporting will prevent all errors and all fraud. A control system, no matter how well designed and operated, can provide only reasonable, not absolute, assurance that the control system's objectives will be met. Further, the design of a control system must reflect the fact that there are resource constraints, and the benefits of controls must be considered relative to their costs. Because of the inherent limitations in all control systems, no evaluation of controls can provide absolute assurance that all control issues and instances of fraud, if any, have been detected.

Changes in Internal Control over Financial Reporting

During the first quarter of fiscal 2016, there were no changes in our internal control over financial reporting that have materially affected, or are reasonably likely to materially affect, our internal control over financial reporting.

PART II -- OTHER INFORMATION

ITEM 1. Legal Proceedings

Descriptions of our legal proceedings are contained in Part I, Item 1, Financial Statements - Notes to Condensed Consolidated Financial Statements - "Note 6".

ITEM 1A. Risk Factors

We face many significant risks in our business, some of which are unknown to us and not presently foreseen. These risks could have a material adverse impact on our business, financial condition and results of operations in the future. We have disclosed a number of material risks under Part I, Item 1A of our annual report on Form 10-K for the fiscal year ended March 31, 2016, which we filed with the Securities and Exchange Commission on May 31, 2016. Except as presented below, there have been no material changes from the risk factors described in our Form 10-K.

Because our long-term growth strategy involves further expansion outside the United States, our business will be susceptible to risks associated with international operations.

In addition, on June 23, 2016, the UK held a referendum in which a majority of voters voted to exit the European Union (Brexit). The result of the Brexit vote adversely impacted global markets and foreign currencies. In particular, the value of the Pound Sterling has sharply declined as compared to the U.S. Dollar and other currencies. This volatility in foreign currencies is expected to continue as the UK negotiates and executes its exit from the European Union but it is uncertain over what time period this will occur. A significantly weaker Pound Sterling compared to the U.S. Dollar could materially reduce our revenues after taking into account foreign currency translation adjustments.

ITEM 2. Unregistered Sales of Equity Securities and Use of Proceeds

	Total Number	Average	Total Number of Shares Purchased	Approximate Dollar Value of Shares that
	of Shares	Price Paid	as Part of Publicly	May Yet be Purchased
	Purchased	Per Share	Announced Program	Under the Program ⁽¹⁾
April 1 - April 30, 2016	-	\$-	-	\$15,000,000
May 1 - May 31, 2016	-	-	-	15,000,000
June 1 - June 30, 2016	-	<u>-</u> -	-	15,000,000
Total	-	<u> </u> \$-	<u>-</u>	

ITEM 5. OTHER INFORMATION

See Item 2 of Part II, "Contractual Obligations", regarding the new DXI operating lease for office space in London, UK.

ITEM 6. Exhibits

Exhibit Number		
10.33	Lease between AG Commercial St. I.B.V. as Landlord, DXI Limited as Tenant, and 8x8 Inc. as Guarantor.	
31.1	Certification of Chief Executive Officer pursuant to Rules 13a-14 and 15d-14 under the Securities Exchange Act of 1934, as adopted pursuant to Section 302 of the Sarbanes-Oxley Act of 2002.	
31.2	Certification of Chief Financial Officer pursuant to Rules 13a-14 and 15d-14 under the Securities Exchange Act of 1934, as adopted pursuant to Section 302 of the Sarbanes-Oxley Act of 2002.	
32.1	Certification of Chief Executive Officer pursuant to 18 U.S.C. Section 1350, as adopted pursuant to Section 906 of the Sarbanes-Oxley Act of 2002.	
32.2	Certification of Chief Financial Officer pursuant to 18 U.S.C. Section 1350, as adopted pursuant to Section 906 of the Sarbanes-Oxley Act of 2002.	
101.INS	XBRL Instance Document	
101.SCH	XBRL Taxonomy Extension Schema	
101.CAL	XBRL Taxonomy Extension Calculation Linkbase	
101.DEF	XBRL Taxonomy Extension Definition Linkbase	
101.LAB	XBRL Taxonomy Extension Label Linkbase	
101.PRE	XBRL Taxonomy Extension Presentation Linkbase	

SIGNATURE

Pursuant to the requirements of the Securities Exchange Act of 1934, the Registrant has duly caused this report to be signed on its behalf by the undersigned thereunto duly authorized.

Date: July 28, 2016 8X8, INC. (Registrant)

By: /s/ MARYELLEN GENOVESE

MaryEllen Genovese Chief Financial Officer (Principal Financial and Chief Accounting Officer and Duly Authorized Officer)

DATED:

REAL ESTATE/084033-00006/ABC/JWXC L_LIVE_EMEA1:33711827v3

Lease

between

AG Commercial St. I.B.V.

as Landlord

DXI Limited

as Tenant

8x8, Inc.

as Tenant's Guarantor

and

One Commercial Street Management Company Limited

as Management Company

relating to

Third Floor, Relay Building, 1 Commercial Street, London E1

Simmons & Simmons

Simmons & Simmons LLP CityPoint One Ropemaker Street London EC2Y 9SS United Kingdom T +44 20 7628 2020 F +44 20 7628 2070 DX Box No 12

CONTENTS

PART	1 : INTERPRETATION OF THIS LEASE	1	
1.	Interpretation	1	
PART	2 : CREATION OF THE LETTING AND RIGHTS AND RESERVATIONS	9	
2.	Letting and term	9	
3.	Rights and reservations	11	
PART	3: RENTS	13	
4.	Rents	13	
5.	Rent review	15	
6.	Determination of disputes	17	
PART	4 : INSURANCE	18	
7.	Insurance obligations	18	
PART	5 : TENANT'S COVENANTS	22	
8.	Financial obligations	22	
9.	Repair and redecoration	24	
10.	Alterations	25	
11.	Use of the Premises	27	
12.	Assignment	29	
13.	Underletting	31	
14.	Sharing occupation, etc.	33	
15.	Charging	33	
16.	Notification of dispositions	33	
17.	Legislation	34	
18.	Third party rights	35	
19.	Title matters	35	
20.	End of the Term	36	
PART	6 : LANDLORD'S COVENANTS AND MANAGEMENT COMPANY'S COVENANTS	38	
21.	Landlord's obligations	38	
	İ		

PART	T 7 : SUPERIOR LEASES	39	
22.	Superior Leases	39	
PART	T 8 : SERVICES AND SERVICE CHARGE	40	
23.	Services and Service Charge	40	
PART	T 9 : GUARANTOR'S OBLIGATIONS	44	
24.	Tenant's Guarantor	44	
PART	T 10 : GENERAL PROVISIONS	44	
25.	Tenant's option to renew	45	
26.	Contractual rights of third parties	47	
27.	Third party disputes	47	
28.	Assignment of reversion	47	
29.	Notices	47	
30.	Law and jurisdiction	48	
31.	Execution and delivery	48	
SCHE	EDULE 1 : THE PREMISES	49	
PART	T 1 : DEFINITION OF THE PREMISES	49	
1.	Identification of the Premises	49	
2.	Areas included in the Premises	49	
3.	Areas excluded from the Premises	49	
PART	T 2 : THE SERVICES SYSTEMS	50	
SCHE	EDULE 2 : RIGHTS GRANTED	51	
1.	Conduits	51	
2.	Rights of access	51	
3.	Support	51	
4.	Signage	51	
5.	Basement Facilities	52	
SCHE	EDULE 3 : RIGHTS RESERVED	53	
1.	Conduits and Services	53	
	ii		

2.	Building and Adjoining Premises	53		
3.	Entry on to the Premises	54		
4.	Support	54		
5.	Fire escapes	55		
6.	Superior Leases	55		
SCH	EDULE 4 : TITLE MATTERS	56		
SCH	SCHEDULE 5 : GUARANTEE PROVISIONS 57			
1.	Defined terms	57		
2.	Effect of the Guarantee	58		
3.	Postponement of rights	58		
4.	Event of Default	59		
SCH	EDULE 6 : PROVISIONS FOR UNDERLETTING	61		
PART	1 : TERMS OF THE UNDERLEASE	61		
1.	General terms	61		
2.	Fine or premium	61		
3.	Rents	61		
4.	Underlease term	61		
5.	Tenant's Covenants	61		
6.	Assignment and charging	61		
7.	Subletting	62		
8.	Sharing of occupation	62		
PART	2 : UNDERTENANT'S COVENANTS	63		
1.	Defined terms	63		
2.	To observe terms of this Lease, etc	63		
SCH	EDULE 7 : SERVICES AND ITEMS OF EXPENDITURE	64		
PART	1 : THE BUILDING SERVICES	64		
PART	2 : THE OFFICE SERVICES	65		
PART	3 : THE CHP SERVICES	66		
	iii			

PART 4 : ITEMS OF EXPENDITURE	66
SCHEDULE 8 : PROVISIONS REFERRED TO IN CLAUSE 25	69
APPENDIX 1 : BASE BUILD SPECIFICATION	69

ίV

LR1. Date of lease	22 June 2016	
LR2. Title number(s)	LR2.1 Landlord's title number(s) EGL232876, EGL464570, AGL249526 and AGL315382	
	LR2.2 Other title numbers AGL249525, AGL240234 and AGL240244	
LR3. Parties to this lease Give full names and addresses of each of the parties. For UK incorporated companies and limited liability partnerships, also give the registered number including any prefix. For overseas companies, also give the territory of incorporation and, if appropriate, the registered number in the United Kingdom including any prefix.	Landlord AG COMMERCIAL ST. I.B.V. incorporated and registered in Netherlands (company registration number 000029897165) whose registered office is at Prinsengracht 919, 1017KD Amsterdam	
	Tenant DXI LIMITED registered in England and Wales (company registration number 7840563) whose registered office is 21 New Street, London EC2M 4HR.	
	Other parties Tenant's Guarantor 8X8, INC. a Delaware corporation registered with corporation number 2676673 whose business office is at 2125 O'Nel Drive, San Jose, California 95131. Management Company ONE COMMERCIAL STREET MANAGEMENT COMPANY LIMITED registered in England and Wales (company registration number 07949192) whose registered office is First Floor-AG, 23 Saville Row, London W1S 2ET	
LR4. Property	In the case of a conflict between this clause and the remainder of this lease then, for the purposes of registration, this clause shall prevail. The premises described in part 1 of schedule 1	

LR5. Prescribed statements etc.	LR5.1 Statements prescribed under rules 179 (dispositions in favour of a charity), 180 (dispositions by a charity) or 196 (leases under the Leasehold Reform, Housing and Urban Development Act 1993) of the Land Registration Rules 2003. Not applicable
	LR5.2 This lease is made under, or by reference to, provisions of: Not applicable
LR6. Term for which the Property is lease	10 years from and including the Term Commencement Date to and including 21 June 2026.
LR7. Premium	None.
LR8. Prohibitions or restrictions on disposing of this lease	This lease contains a provision that prohibits or restricts dispositions.
LR9. Rights of acquisition etc.	LR9.1 Tenant's contractual rights to renew this lease, to acquire the reversion or another lease of the Property, or to acquire an interest in other land Tenant's option to renew at clause 25 and Schedule 8.
	LR9.2 Tenant's covenant to (or offer to) surrender this lease None
	LR9.3 Landlord's contractual rights to acquire this lease None
LR10. Restrictive covenants given in this lease by the Landlord in respect of land other than the Property.	None
LR11. Easements	LR11.1 Easements granted by this lease for the benefit of the Property schedule 2
	LR11.2 Easements granted or reserved by this lease over the Property for the benefit of other property schedule 3

LR12. Estate rentcharge burdening the Property	None
LR13. Application for standard form of restriction	None.
LR14. Declaration of trust where there is more than one person comprising the Tenant	Not applicable.

PARTICULARS

Date 22 June 2016

The Landlord referred to in clause LR3 and the person from time to Landlord

time entitled to the reversion immediately expectant on termination of

the term.

The Tenant referred to in clause LR3 or the person who is from time Tenant

to time the Tenant under this Lease.

The Tenant's Guarantor referred to in clause LR3 or if none is referred to Tenant's Guarantor means any guarantor or guarantors of the Tenant's obligations under this Lease from time to time and where the Tenant's Guarantor is an individual includes the Tenant's Guarantor's

personal representatives.

Management Company The Management Company referred to in clause LR3.

The use of the Premises for office purposes within Class B1 of the **Authorised Use**

Use Classes Order.

Break Date 22 June 2022.

Tenant's Guarantor

Guarantor Notice

Principal Rent

Notice

The building known as Relay Building, 1 Commercial Street, London **Building**

E1 described in more detail in clause 1.1.

A statutory declaration dated 20 June 2016 in a form complying with Declaration

the requirements of Schedule 2 to the Order in response to the Notice.

A statutory declaration dated 20 June 2016 in a form complying **Guarantor Declaration** with the requirements of Schedule 2 to the Order in response to the

Guarantor Notice.

A notice to the Tenant's Guarantor dated 17 June 2016 in a form

complying with the requirements of Schedule 1 to the Order in relation to the tenancy to be entered into by the Tenant's Guarantor pursuant

to paragraph 4.2 of Schedule 5 of this Lease.

A notice dated 17 June 2016 in a form complying with the

requirements of Schedule 1 to the Order in relation to the tenancy

created by this Lease.

The Regulatory Reform (Business Tenancies) (England and Wales) Order

Order 2003.

Premises The Property referred to in clause LR4

> From and including the Term Commencement Date to and including 21 December 2017, the rent of one peppercorn per annum (if

> demanded); from and including the Rent Commencement Date, the rent of eight hundred and seventy five thousand two hundred and

> fifteen pounds (£875,215) per annum subject to review under clause

5.

Rent Commencement Date 22 December 2017.

Review Date 22 June 2021.

Term The term of years referred to in clause LR6.

Term Commencement Date 22 June 2016.

This Lease creates a "new tenancy" for the purposes of the 1995 Act.

THIS LEASE

is dated on the date set out in the Particulars and made

BETWEEN:

- (1) The Landlord;
- (2) The Tenant;
- (3) The Tenant's Guarantor; and
- (4) The Management Company.

THE PARTIES AGREE THAT

•

PART 1: INTERPRETATION OF THIS LEASE

1

Interpretation

1.1 Defined terms

In this Lease, unless the contrary intention appears:

"1954 Act"

means the Landlord and Tenant Act 1954.

"1995 Act" means the Landlord and Tenant (Covenants) Act 1995.

"AC Energy" means the quantities of electricity and gas consumed from time to time by those elements of the Services Systems the function of which is the generation of hot water and chilled water for the operation of the air conditioning systems within the Building.

"Adjoining Premises" means any land, buildings or structures near or adjoining the Building in or over which the Landlord or any other person owns an estate or interest from time to time.

"Alterations" means any alterations, additions or other works to the Premises.

"Authorised Guarantee Agreement"

means an authorised guarantee agreement for the purposes of s.16 of the 1995 Act.

"Base Specification"

means the works detailed in the specification annexed to this Lease.

"Basement Car Park Lease"

means the lease of basement premises, Ibis London City Hotel, 5 Commercial Street, London E1 6BF dated 12 September 2011 made between (1) Mangrove Securities Limited and (2) DK Holdings Limited which is registered at the Land Registry with title number AGL240244 and includes any documents supplemental to it.

"<u>Building</u>" means the building of which the Premises form part which is shown for identification edged blue on Plan A and:
(A) includes all Conduits from time to time within or leading to the Building which serve it and which do not form part of the public mains; and

(B) includes the Services Systems and all other landlord's fixtures forming part of the Building from time to time; and

- (C) includes all additions, alterations and other works made to the Building from time to time (but only including the Alterations to the extent that the Tenant has given the Landlord full details of the same pursuant to clause 11.5(G)); and
- (D) excludes any tenant's fixtures forming part of the Building from time to time.
- "Building Services" means the services set out in Part 1 of Schedule 7.
- "Business Hours" means the hours of 7 am to 7 pm Mondays to Fridays, excluding in each case all usual bank or public holidays, or such longer hours on those days and such hours on such other days as the Landlord may from time to time determine.
- "CDM Regulations" means the Construction (Design and Management) Regulations 2015.
- "CHP Services" means the services set out in Part 3 of Schedule 7.
- "Conduits" means all conducting media and ancillary apparatus used for the passage or transmission of Utilities or used by the Service Systems.
- "CRC" means the carbon reduction commitment and trading scheme established by the CRC Order.
- "CRC Costs" means the aggregate of the costs incurred by the Landlord (or by any undertaking which is treated for the purposes of the CRC Order as being a member of the same group as the Landlord) whether directly or indirectly, wholly or in connection with or in relation to its obligations under the CRC Order, including without limitation registering and maintaining its registration as a participant in CRC; complying with the reporting requirements of CRC; and acquiring, financing, administering, and surrendering or recycling allowances in accordance with its obligations under the CRC Order.

 "CRC Net Costs" means the excess of the CRC Costs over such amount as is received by the Landlord (or by any undertaking which is treated for the purposes of the CRC Order as being a member of the same group as the Landlord) for the surrender or recycling of allowances pursuant to the CRC Order in any Accounting Period (as defined in clause 23.1).

 "CRC Order" means the CRC Energy Efficiency Scheme Order 2010.

"Current Guarantor"

means someone who, immediately before a proposed assignment, is either a guarantor of the Tenant's obligations under this Lease or a guarantor of the obligations given by a former tenant of this Lease under an Authorised Guarantee Agreement;

"DEC"

means Display Energy Certificate as defined in the EPC Regulations;

"Energy Centre" means the energy centre from which Heat is supplied (inter alia) to the Premises

"Energy Charge" means the cost of that proportion of AC Energy attributable to the provision of hot water and chilled water to the air conditioning systems serving the Premises as evidenced by meters installed for the purpose of measuring such proportion or as reasonably determined by the Landlord or the Management Company based on the tariff to be set by the Landlord or the Management Company such cost to include:

- (A) the standing charges, meter rents and the costs of metered units and all other existing and future levies, costs and taxes payable in respect of that provision of that AC Energy; or
- (B) the reasonable and proper administrative costs in connection with that provision of energy.
- "Energy Documents" means all or any (as relevant) of the EPC, the DEC and the Recommendation Report;
- "EPC" means Energy Performance Certificate as defined in the EPC Regulations;
- <u>"EPC Policy"</u> means the Landlord's reasonable policy (if any) for energy efficiency for the Property as revised from time to time:
- <u>"EPC Rating"</u> means the operational rating asset rating or energy efficiency rating accorded to the Property in the EPC (in accordance with the Regulations);
- <u>"EPC Regulations"</u> means the Energy Performance of Buildings (Certificates and Inspections) (England and Wales) Regulations 2007 (as amended and supplemented from time to time);
- "Event of Insolvency" means any of the following:
- (A) the Tenant is deemed to be unable to pay its debts under s.123 Insolvency Act 1986 (if a company) or s.268 Insolvency Act 1986 (if an individual); or
- (B) proceedings are taken against the Tenant or the Tenant makes any agreement, arrangement, scheme or composition with its creditors, including a voluntary arrangement under part I (if a company) or part VIII (if an individual) of the Insolvency Act 1986; or
- (C) if the Tenant is a company:
- (1) it has an administration order made against it under the Insolvency Act 1986 or a petition is presented or made; or
- (2) notice of intention to appoint an administrator is given; or
- (3) an administrator is appointed; or
- (4) a resolution is passed approving the presentation of any such petition, the making of any such application or appointment of an administrator; or
- (D) if the Tenant is an individual, a bankruptcy petition against him is presented to the Court, he has a bankruptcy order made against him or he is otherwise adjudged to be bankrupt; or
- (E) any step is taken to enforce security over or to obtain possession of the Premises of the Tenant including the appointment of an interim receiver, receiver, administrative receiver (in the case of a debenture to which the Tenant is a party created before 15 September 2003); or
- (F) if the Tenant is a company, any step is taken by any person, including the Tenant, with a view to the winding up, whether solvent or insolvent, of the Tenant, or the Tenant ceases to carry on all or a material part of its business, except for and followed by a reconstruction, amalgamation, reorganisation, merger or consolidation of the Tenant while solvent; or

- (G) in respect of a Tenant incorporated or resident in a jurisdiction outside England and Wales, any event or circumstance occurs which under the laws of that jurisdiction has an analogous or equivalent effect to any of the Events of Insolvency defined in this definition: or
- (H) any of the Events of Insolvency defined in this definition occur in relation to any guarantor of the Tenant.

"Expiry of the Term"

: the date of the expiration (but not of any sooner determination) of the Term;

"Fit-Out Guide"

means the Cushman & Wakefield Guide to Sustainable Office Fit-Out for the Building (October 2015) a copy of which was supplied to the Tenant before it became bound to enter into this Lease and/or any other reasonable fit-out regulations made by the Landlord or the Management Company at any time.

"Ground Floor Corridor Lease"

means the lease of ground floor, Ibis London City Hotel, 5 Commercial Street, London E1 6BF dated 12 September 2011 made between (1) Mangrove Securities Limited and (2) Julius Properties Limited which is registered at the Land Registry with title number AGL240234 and includes any documents supplemental to it.

"Group Company" means, in relation to any company, another company which is a member of the same group of companies as that company within the meaning of s.42 of the 1954 Act.

"<u>Hazardous Material</u>" means any substance, whether in solid, liquid or gaseous form, which is or may become a pollutant or which is hazardous, toxic, radioactive, noxious, corrosive or caustic.

"Health and Safety File" means any health and safety file required by the CDM Regulations.

"Heat" means heat generated from the Heat Installations to be supplied to the Premises and other parts of the Building.

"<u>Heat Charge</u>" means the costs (excluding the Standing Charge) of providing Heat to the Premises as evidenced by meters installed for the purpose of measuring such proportion or as reasonably determined by the Landlord or the Management Company based on the tariff to be set by the Landlord or the Management Company.

"<u>Heat Installations</u>" means the Energy Centre together with the network of pipes, wires and other ancillary plant and equipment that transfer Heat from the Energy Centre to the Building (including the Premises) together with all connected meters and monitoring equipment.

"Insurance Event" means any damage or destruction of the Building by any of the Insured Risks which at the date of such damage or destruction is covered by any policy of insurance maintained by the Landlord under this Lease.

"Insurance Excess" means such amounts, if any, which are stated in any insurance policy maintained by the Landlord as being not payable to the insured in respect of the first part of any loss resulting from the happening of any of the Insured Risks.

- "Insurance Rent" means the sums payable by the Tenant to the Landlord under clause 7.4.
- "Insured Risks" means, so long as cover is available from time to time with reputable insurance offices in the United Kingdom or through underwriters at Lloyd's on reasonable commercial terms and subject to the Insurance Excesses, exclusions and conditions of such cover, the following risks: fire; subterranean fire; lightning; storm; tempest; flood; explosion; bursting or overflowing of water tanks, apparatus or pipes or the escape of water from any of them; aircraft or other aerial devices or articles dropped or falling from any of them; riot; civil commotion; malicious damage; impact; earthquake; Terrorist Activity; heave; subsidence and such other risks which the Landlord reasonably requires from time to time.
- "Landlord's Covenants" means the obligations in this Lease to be complied with by the Landlord.
- "Lease" means this lease and any document which is supplemental to it.
- "Lettable Areas" means those parts of the Building let or designed to be let to occupational tenants excluding:
- (A) any parts used or designed to be used by persons responsible for the provision of public utilities in connection with the carrying out of their statutory duties; and
- (B) accommodation from time to time reserved in the Building for staff engaged in or employed in connection with the provision of the Services.
- "<u>LUL Lease</u>" means the lease of airspace and structures above Aldgate East Underground Station, Whitechapel High Street, London dated 13 January 2012 made between (1) London Underground Limited and (2) Redrow Homes Limited which is registered at the Land Registry with title number AGL249526 and includes any documents supplemental to it.
- "Main Structure" means:
- (A) all structural or loadbearing walls, columns, slabs, joists and beams forming part of the Building, its foundations and roofs and all external parts of the Building the repair and maintenance of which are not the responsibility of tenants or other occupiers of the Building; and
- (B) all party structures, boundary walls and railings, if any, within the boundaries of the Building; and
- (C) all parts of the Building reserved for the housing of the Services Systems or otherwise required or used for the provision of the Services; and
- (D) the Services Systems; and
- (E) the basement car park and storage area (including the areas demised by the Ground Floor Corridor Lease and the Basement Car Park Lease).
- "Management Company's Covenants" means the obligations in this Lease to be complied with by the Management Company.
- "Office Common Parts" means the entrance halls, corridors, staircases, landings, lift lobbies, lifts, passages, lavatories, storerooms, service areas, loading areas, car parking areas, storage areas and other areas in or forming part of the office parts of the Building from time to time intended for the common use of more than one office tenant or office occupier but for the avoidance of doubt this excludes the Main Structure;

- "Office Services" means the services set out in Part 2 of Schedule 7.
- "Outgoings" means any rates, taxes, charges, and outgoings from time to time assessed or charged on the Premises or payable by the owner or occupier of them and includes a proper proportion, to be determined by the Landlord acting reasonably, of any amounts assessed, charged or payable in respect of the Building.
- "Particulars" means the Particulars at the front of this Lease.
- "<u>Plans</u>" means the plans annexed to this Lease and marked "<u>Plan A</u>" and "<u>Plan B</u>" and references to individual Plans are to the Plans so marked.
- "Planning Acts" means the planning Acts defined in s.336 Town and Country Planning Act 1990 together with the Planning and Compensation Act 1991 and any other Statute relating to town and country planning.
- "Prescribed Rate" means four per cent per annum above the base rate for the time being of Barclays Bank PLC or such other clearing bank as the Landlord may nominate or, if the clearing banks cease to publish a base rate, four per cent per annum above such reasonably comparable rate of interest as the Landlord may determine.
- "Quarter Days" means 25 March, 24 June, 29 September and 25 December.
- "Rating Proposal" means any proposal made to agree the rateable value of or amend the non-domestic rating list in respect of the Premises or the Building.
- "Recommendation Report" means the recommendation report accompanying the EPC issued in accordance with the EPC Regulations.
- "Rents" mean the rents payable under clause 4.1.
- "Residential Common Parts" means the entrance halls, corridors, staircases, landings, lift lobbies, lifts, passages, lavatories, storerooms, service areas, loading areas, car parking areas, storage areas and other areas in or forming part of the residential parts of the Building from time to time intended for the common use of more than one residential tenant or residential occupier or which are not designed to be let to occupational tenants but for the avoidance of doubt this excludes the Main Structure;
- "Retained Parts" means all parts of the Building which do not comprise;
- (A) Lettable Areas;
- (B) the Office Common Parts; and
- (C) the Residential Common Parts;
- (D) and includes the Main Structure.
- "Service Charge" means the sums payable by the Tenant under clause 23.3.
- "Services" means the Building Services, the Office Services and the CHP Services.

"Services Systems" means all electrical and mechanical apparatus, plant, machinery and equipment installed in the Building from time to time, including those listed in Part 2 of schedule 1, used for the provision of services and facilities within or to the Building but excludes any installed by the Tenant or any other tenant or occupiers of the Building which are tenant's fixtures. "Standing Charge" means the administrative charges for billing of the Heat Charge and a provision for bad debt for non-payment of the Heat Charge.

"Statute" means every Act of Parliament, including any named in this Lease, in force during the Term together with all other legislation having effect in England and Wales.

"Superior Landlords" means the landlords for the time being of the Superior Leases.

"Superior Leases" means the LUL Lease, the TFL Leases, the Basement Car Park Lease and the Ground Floor Corridor Lease.

"Tenant's Covenants" means the obligations in this Lease to be complied with by the Tenant.

"Terrorist Activity"

means any act of any person or persons acting on behalf of or in connection with any organisation, including any association or combination of persons, which carries out activities directed towards the overthrowing or influencing by force or violence of Her Majesty's Government in the United Kingdom or any other government de jure or de facto.

"TFL Leases" means:

- (A) the lease of airspace and structures above 1 Commercial Street, London E1 6BF dated 13 January 2012 made between (1) Transport for London and (2) Redrow Homes Limited which is registered at the Land Registry with title number AGL249525 and includes any documents supplemental to it; and
- (B) the supplemental lease of airspace and structures above 1 Commercial Street, London E1 6BF dated 23 June 2014 made between (1) Transport for London and (2) Redrow Homes Limited which is registered at the Land Registry with title number AGL315382 and includes any documents supplemental to it at the date of this Lease.
- "<u>Uninsured Risk"</u> means an Insured Risk against which insurance is not or ceases to be obtainable for such risks on normal commercial terms in the London insurance market at reasonable commercial rates generally available in the London insurance market for a property of the type, size and location as the Building or an Insured Risk not covered by any policy of insurance maintained by the Landlord under this Lease.

"Use Classes Order" means the Town and Country Planning (Use Classes) Order 1987 as at the date of this Lease.

"<u>Utilities</u>" means the drainage of surface water and sewage and the supply or transmission of electricity, gas, telecommunications, water or any other services or supplies made to or consumed in the Building.

"Value Added Tax" includes any future tax of a like nature.

1.2 The Particulars

The Particulars form part of this Lease and words and expressions defined in the Particulars shall be treated as defined terms in this Lease.

1.3 Construction

In this Lease, unless the contrary intention appears:

- (A) references to Statute include references to:
 - 1. that Statute as amended or re-enacted or as other Statutes modify its application from time to time; and
- any subordinate legislation made or to be made under that Statute; and
 - (B) references to clauses or schedules are references to clauses in or schedules to this Lease and references to paragraphs are references to paragraphs in the schedule in which those references are made; and
 - (C) references to the singular include the plural and vice versa; and
 - (D) references to the parties include their successors in title; and
 - (E) references to persons include individuals, companies, firms, partnerships, government bodies or agencies and corporations sole and aggregate; and
 - (F) references to the masculine gender include the feminine and the neuter genders and vice versa; and
 - (G) references to an indemnity mean an indemnity against all actions, claims, demands and proceedings made against the Landlord and all costs, expenses, liabilities and losses incurred directly or indirectly by the Landlord and "indemnify" and "indemnified" shall be construed in the same way; and
 - (H) references to the Premises, the Building and Adjoining Premises include any part of them; and
 - (I) references to the end of the Term include the determination of the Term before the end of the Term; and
 - (J) where the Tenant requires the consent of the Landlord to any Alterations, any change of the Authorised Use or any assignment or any underletting, such consent shall not be effective unless given by way of a formal licence executed as a deed; and
 - (K) any reference to the date of assignment shall mean the date of the deed of assignment or transfer of this Lease and any covenants given to the Landlord on any assignment of this Lease shall take effect from such date; and

- (L) any obligation on the Tenant includes an obligation on the Tenant to ensure that any person deriving title under the Tenant and its and their agents, employees, licensees and any other person under its or their control comply with that obligation and any reference to an act or default of the Tenant includes the act or default of those persons; and
- (M) any obligation on the Tenant not to do an act or thing includes an obligation not to permit or allow that act or thing to be done; and
- (N) any obligations entered into by more than one person in this Lease are entered into jointly and severally; and
- (O) the headings shall not affect the interpretation of this Lease; and
- (P) if any provision in this Lease is held to be illegal, void, invalid or unenforceable for any reason, the legality, validity and enforceability of the remainder of this Lease shall not be affected.

PART 2: CREATION OF THE LETTING AND RIGHTS AND RESERVATIONS

2. Letting and term

2.1 Creation of the Term

The Landlord lets the Premises to the Tenant for the Term reserving the Rents.

2.2 Re-entry

The Landlord shall be entitled to re-enter the Premises or any part of them and by so doing end this Lease if:

- (A) the Rents or any part of them remain unpaid twenty one days after becoming payable, whether formally demanded or not; or
- (B) the Tenant does not comply with the Tenant's Covenants; or
- (C) there is an Event of Insolvency; or
- (D) any process of distress, execution or similar process is levied against any of the assets and undertaking of the Tenant; or
- (E) the Tenant or any guarantor of the Tenant, if a company, changes the liability of its shareholders from limited to unlimited or vice versa.

2.3 Termination on destruction

- (A) If an Insurance Event damages or destroys the whole or substantially the whole of the Premises or renders them inaccessible and:
 - 1. they have not been reinstated and made accessible within three (3) years and six (6) months of its occurrence then if the Landlord or the Tenant serves not less than six months' prior written notice on the other, this Lease shall end on the date the notice takes effect: or
 - 2. they have not been reinstated and made accessible within four years of its occurrence then if the Landlord or the Tenant serves written notice on the other, this Lease shall end on the date the notice takes effect; unless in either case clauses 2.3(B) or 2.3(B) apply, and the Landlord shall be entitled to retain for its own benefit all insurance moneys received or receivable under any policy of insurance maintained by it
- (B) If any insurance moneys have been withheld in whole or in part due to the act or default of the Tenant, any notice served by the Tenant under clause 2.3(A)(1) shall be ineffective unless the Tenant has complied with clause 7.11.
- (C) If the Premises have been reinstated and made accessible by the date any notice served by the Tenant under clause 2.3(A)(1) expires, this Lease shall not end.

Any dispute about the operation of this clause 2.3 shall be submitted at the request of the Landlord or the Tenant to the decision of a single arbitrator under the Arbitration Act 1996.

2.4 Break clause

- (A) The Tenant may end this Lease on the Break Date by serving written notice of at least nine (9) months on the Landlord. This Lease shall then end on the Break Date if:
 - 1. there are no outstanding substantial breaches of the Tenant's Covenants on the Break Date of which the Landlord has given notice to the Tenant and given the Tenant a reasonable opportunity to remedy that breach; and
 - 2. the Tenant has paid all the Rents due under this Lease up to and including the Break Date; and
 - 3. on the date the Tenant serves notice on the Landlord under this clause 2.4 the Tenant pays to the Landlord in addition to any other sums due under this Lease the sum of Four Hundred and Thirty Seven Thousand Six Hundred and Seven Pounds and Fifth Pence (£437,607.50) (plus VAT if applicable); and
 - 4. on the Break Date the Tenant returns the Premises to the Landlord free of occupation and with no subsisting subleases
 - unless the Landlord in its absolute discretion elects in writing to waive these conditions to the ending of this Lease.

- (B) If the Lease ends on the Break Date and the Tenant has paid any Principal Rent and/or Insurance Rent in relation to the period falling after the Break Date then, within 14 days after the Break Date, the Landlord shall refund such sums to the Tenant (the apportionment of which shall be from and including the date after the Break Date).
- (C) Where the title to this Lease is registered at Land Registry, on or before the Break Date the Tenant shall provide the Landlord with evidence which will satisfy the Chief Land Registrar that the title to this Lease can be closed.

2.5 Exclusion of Sections 24 to 28 Landlord and Tenant Act 1954

In relation to the 1954 Act:

- (A) the Tenant confirms that before it entered into the tenancy created by this Lease or became contractually bound to do so:
 - 1. the Landlord served on the Tenant the Notice; and
 - 2. the Tenant, or a person duly authorised by the Tenant, made the Declaration; and
 - 3. where the Declaration was made by a person other than the Tenant, the declarant was duly authorised by the Tenant to make the Declaration on the Tenant's behalf; and
- (B) the Landlord and Tenant agree to exclude the provisions of ss24 to 28 (inclusive) of the 1954 Act in relation to the tenancy created by this Lease.
- (C) the Tenant's Guarantor confirms that before this Lease was entered into:
 - 1. the Landlord served on the Tenant's Guarantor the Guarantor Notice applying to the tenancy to be entered into by the Tenant's Guarantor pursuant to paragraph 4.2 of Schedule 5 of this Lease; and
 - 2. the Tenant's Guarantor, or a person duly authorised by the Tenant's Guarantor, made the Guarantor Declaration; and
 - 3. where the Guarantor Declaration was made by a person other than the Tenant's Guarantor, the declarant was duly authorised by the Tenant's Guarantor to make the Guarantor Declaration on the Tenant's Guarantor behalf; and
- (D) the Landlord and the Tenant's Guarantor agree to exclude the provisions of ss24 to 28 (inclusive) of the 1954 Act in relation to the tenancy to be entered into by the Tenant's Guarantor pursuant to paragraph 4.2 of Schedule 5 of this Lease.

2.6 Effect of termination

When this Lease ends it shall be without prejudice to any outstanding liabilities of any party to any other party.

3. Rights and reservations

3.1 Rights granted

The Landlord lets the Premises together with the rights set out in schedule 2:

- (A) so far as the Landlord has the power to grant them; and
- (B) for the benefit of the Tenant and any person deriving title under the Tenant; and
- (C) in common with the Landlord and all others authorised by it, unless otherwise stated in schedule 2; and
- (D) subject to the right of the Landlord to interrupt, modify or end these rights without any liability to the Tenant if it grants the Tenant such alternative rights as may be necessary for the proper use and enjoyment of the Premises.

3.2 Reservations in the Superior Leases

The Premises are let subject to the rights, for the benefit of the Superior Landlords and all others authorised by it, over the Premises set out in the Second Schedule to the TFL Lease and in the Second Schedule to the LUL Lease.

3.3 Rights reserved

The Landlord reserves throughout the Term to itself, the Management Company, any superior landlord and all others authorised by it the rights over the Premises set out in schedule 3 which may be exercised without any liability to the Tenant beyond that set out in schedule 3 and the Tenant shall not prevent or interfere with the exercise of these rights.

3.4 Title matters

The Premises are let subject to the title matters set out in schedule 4 and all other easements, covenants, privileges and rights enjoyed over or against the Building so far as any of them are still subsisting and capable of taking effect.

3.5 No implied or prescriptive rights

- (A) section 62 Law of Property Act 1925 shall not apply to this Lease; and
- (B) the Tenant shall not be entitled to the benefit of or to claim or enforce against the Landlord or any other person any covenant, right, agreement, privilege or easement in respect of the Premises, the Building or any Adjoining Premises except those expressly granted in clause 3.1; and
- (C) any other right from time to time enjoyed by the Tenant in respect of the Premises, the Building or any Adjoining Premises shall be enjoyed by the consent of the Landlord, terminable at any time by notice in writing to the Tenant and without any liability to the Tenant.

3.6 No benefit of covenants and conditions

Nothing in this Lease shall give the Tenant any right to claim or enforce against the Landlord or any other person or to receive the benefit of any covenant, term or condition in any lease (other than this Lease), deed or document relating to the Premises, the Building or any Adjoining Premises.

3.7 Use of the Building and Adjoining Premises

Nothing in this Lease shall limit or affect the rights of:

- (A) the Landlord in relation to the remainder of the Building; or
- (B) the Landlord or any other person in relation to any Adjoining Premises

to use or otherwise deal with the remainder of the Building and any Adjoining Premises in such manner and for any purpose as it wishes provided that there is no material detriment to the Tenant's use or enjoyment of the Premises in accordance with the terms of this Lease.

PART 3: RENTS

4. Rents

4.1 Rents payable

During the Term the Tenant shall pay to the Landlord and (where required) to the Management Company by way of rent without any abatement, counterclaim, deduction, reduction or set-off whatsoever unless required to do so by any Statute:

- (A) the Principal Rent by equal quarterly payments in advance on the Quarter Days which, if the Landlord requires, shall be paid by banker's standing order; and
- (B) the Insurance Rent, which shall be payable within fourteen days of demand; and
- (C) the Service Charge and the other sums referred to in clause 23 at the times and in the manner set out in clause 23; and
- (D) the Heat Charge, which shall be payable within fourteen days of demand; and
- (E) the Standing Charge, which shall be payable within fourteen days of demand; and
- (F) the Energy Charge, which shall be payable within fourteen days of demand; and
- (G) any other sums expressed to be payable as additional rent under this Lease, which shall be payable within fourteen days of demand.

4.2 First payment of Principal Rent

Principal Rent for the period from and including the Rent Commencement Date to the next Quarter Day shall be paid on the Rent Commencement Date.

4.3 Payment of Insurance Rent and Service Charge

Insurance Rent and Service Charge shall be payable from and including the Term Commencement Date.

4.4 Value Added Tax

Each sum payable by the Tenant under this Lease shall be treated as being exclusive of Value Added Tax. The Tenant shall pay as additional rent any Value Added Tax properly demanded by the Landlord or the Management Company (and upon receipt of payment of such VAT the Landlord or the Management Company (as applicable) shall provide a valid VAT invoice to the Tenant):

- (A) on the Rents; and
- (B) on any supply made by the Landlord or the Management Company under this Lease; and

(C) on any supply made to the Landlord or the Management Company or any other person that the Tenant covenants to reimburse, but only to the extent that the person to whom the supply was made is unable to recover the Value Added Tax.

4.5 Overpayment of Value Added Tax

If any amount paid by the Tenant to the Landlord or the Management Company in respect of Value Added Tax was not properly chargeable:

- (A) if the prescribed VAT accounting period of the Landlord or the Management Company in which the Value Added Tax was charged has not ended, the Landlord or the Management Company shall repay the Valued Added Tax promptly to the Tenant; and
- (B) if the prescribed VAT accounting period of the Landlord or the Management Company in which the Value Added Tax was charged has ended, the Landlord or the Management Company shall use reasonable endeavours, to obtain the repayment of the Value Added Tax as soon as possible. Upon receipt of the repayment, the Landlord or the Management Company shall repay the amount recovered promptly to the Tenant.

4.6 Interest on late payments

The Tenant shall pay as additional rent interest on any sums due to the Landlord or the Management Company which are not paid on the due date for payment. Interest shall be calculated at the Prescribed Rate, both before and after any judgment, from the due date for payment to the date on which payment is made. It shall accrue on a daily basis and be compounded on the Quarter Days.

4.7 Due date for payment

If the Tenant breaches any of the Tenant's Covenants, the due date for payment of any sums incurred by the Landlord or the Management Company which the Tenant covenants to reimburse as a result of that breach shall be deemed to be the date upon which the Landlord or the Management Company first incurred them.

4.8 Suspension of rent

The following provisions shall apply following an Insurance Event which renders the Premises unfit for occupation and use or inaccessible:

(A) the Principal Rent and the Service Charge or a proper proportion according to the extent of the damage or destruction, shall be suspended except to the extent that any insurance moneys are withheld due to the act or default of the Tenant; and (B) the period of suspension shall be from the date of the Insurance Event until the earlier of:

- 1. the date when the Premises are reinstated and made accessible or:
- 2. the date on which the period covered by the Landlord's loss of rent insurance expires; and
- (C) any dispute about the operation of this clause 4.8 shall be submitted at the request of the Landlord or the Tenant to the decision of a single arbitrator under the Arbitration Act 1996.

4.9 Waiver of breaches

If there is any breach of the Tenant's Covenants which the Landlord reasonably believes to be material and it refrains from demanding or accepting payment of the Rents or any other sums due under this Lease, the Tenant shall pay interest to the Landlord on those amounts until the date on which the Landlord accepts payment of them and otherwise on the basis set out in clause 4.6.

5. Rent review

5.1 Defined terms

In this clause 5:

"Assumptions"

means the following matters to be assumed at the Review Date:

- (A) if the Building has been damaged or destroyed, it has been reinstated; and
- (B) the Building and any Conduits within or serving it are in a good state of repair and condition and the Building is accessible; and
- (C) the Services Systems are in good working order and the Building has the benefit of all the Services; and
- (D) the Premises are accessible and ready to be fitted out by a willing tenant with the Landlord having provided the Base Specification at its own cost and expense; and
- (E) the Premises may lawfully be used for the Authorised Use; and
- (F) all licences and consents required are in force and available to a willing tenant on the Review Date; and
- (G) all the Tenant's Covenants have been complied with; and
- (H) there is no Statute limiting the amount of rent payable in respect of the Premises or non-residential property generally. "Disregards"

means the following matters to be disregarded at the Review Date:

- (A) any effect on the rent of the goodwill of the business carried on in the Premises and of the occupation of the Premises by the Tenant, any permitted undertenant or other lawful occupier; and
- (B) any Alterations carried out at the sole cost of the Tenant or any permitted undertenant which would increase the rent payable unless they were made pursuant to an obligation to the Landlord under this Lease or in breach of its terms; and

- (C) any Alterations which would reduce the rent payable; and
- (D) any effect on the rent of any obligation to reinstate Alterations; and
- "Market Rent" means the rent which would reasonably be expected to become payable under a Notional Lease after any Rent Concession under that lease has expired or has been received by the willing tenant and on the basis that the Assumptions and Disregards apply.
- "Notional Lease" means a lease of the whole of the Premises granted:
- (A) in the open market; and
- (B) on the Review Date with vacant possession; and
- (C) by a willing landlord to a willing tenant; and
- (D) without the Landlord receiving a fine or premium for the grant of the lease; and
- (E) for a term equal to ten years commencing on the Review Date, with a rent review on the fifth anniversary of the Review Date: and
- (F) having as the Authorised Use office use within Class B1 of the Use Classes Order; and
- (G) otherwise on the terms of this Lease, as at the Review Date, including this clause 5, but excluding clause 2.4 and also excluding the amount of the Principal Rent payable and any rent free period or period of reduced or concessionary rent, capital payment or any other inducement of whatever nature allowed or given to the Tenant at the commencement of or otherwise in relation to this Lease.

"Rent Concession"

means any rent free period or period of reduced or concessionary rent, in each case commencing on the Review Date, or any capital payment or other inducement which, in any such case, might reasonably be expected to be allowed or given to reflect the period reasonably required by a willing tenant to fit out the Premises in accordance with the practice of the open market on the Review Date.

"Revised Rent"

means the amount of Principal Rent payable from and including the Review Date.

"Valuer"

means an independent chartered surveyor of at least ten years' standing who is experienced in the letting and valuation of property similar to the Premises and in the same area as them who shall be appointed and act in accordance with clause 6.

5.2 Basis of the review

On the Review Date the Principal Rent shall be reviewed to the higher of the Principal Rent reserved before the Review Date and the Market Rent.

5.3 Agreeing the Revised Rent

In the absence of agreement between the Landlord and the Tenant, a Valuer shall determine the Revised Rent, but no Valuer may be appointed before the date which is two months before the Review Date and the Valuer shall not determine the Revised Rent before the Review Date.

5.4 Rent pending determination

Until the Revised Rent is agreed or determined, the Tenant shall continue to pay the Principal Rent reserved before the Review Date.

5.5 Payment of Revised Rent

Within fourteen days of the Revised Rent being agreed or determined, the Tenant shall pay to the Landlord any increase in the Principal Rent together with interest on such increase at two per cent below the Prescribed Rate calculated from and including each Quarter Day on which each instalment of the increased rent would have been payable had the Revised Rent been agreed or determined on or before the Rent Review Date.

5.6 Rent review memorandum

Following the agreement or determination of the Revised Rent, the Landlord and the Tenant and any guarantor of the Tenant shall sign a memorandum recording that the Revised Rent is the Principal Rent payable from and including the Review Date. The parties shall be responsible for their own costs and expenses for this.

5.7 Statutory restrictions

If any Statute prevents the Principal Rent being reviewed on the Review Date or restricts the Landlord's right to receive the whole or any part of the Principal Rent following the rent review, when that Statute ceases to have that effect the Landlord may require the Principal Rent to be reviewed before the next Review Date or (if there is no next Review Date) before the end of the Term by serving not less than one month's prior written notice on the Tenant. The Principal Rent shall then be reviewed on the date when that notice expires on the terms of this clause 5 and that date shall be a Review Date.

5.8 Time not of the essence

Nothing in this Lease nor any act or default of the parties in relation to any review of the Principal Rent shall make time of the essence in relation to the operation of this clause 5.

6. <u>Determination of disputes</u>

6.1 Appointment of the Valuer

The Valuer shall be appointed by agreement between the Landlord and the Tenant or, in the absence of agreement, by the President for the time being of the Royal Institution of Chartered Surveyors on the application of either the Landlord or the Tenant. If he is unable or unwilling to make the appointment, it shall be made by the Vice President or the next senior officer willing and able to do so. If there is no such officer available, it shall be made by an officer of such other professional body of surveyors nominated by the Landlord.

6.2 Basis of the appointment

The Valuer shall act as a single arbitrator under the Arbitration Act 1996 unless the Landlord and Tenant agree that he shall act as an expert at the time he is appointed.

6.3 Appointment as an arbitrator

If the Valuer acts as an arbitrator:

- (A) all submissions made or evidence supplied to him shall be in writing unless the Landlord notifies the Tenant and the Valuer in writing within 21 days of his appointment that this requirement shall not apply; and
- (B) the date of his award shall be deemed to be the date on which he serves a copy of the award on the Landlord and the Tenant; and
- (C) he shall not be entitled to order the rectification, setting aside or cancellation of this Lease or any other deed or document; and
- (D) he shall not be entitled to direct that the recoverable costs of the arbitration, or any part of it, be limited to a specified amount; and
- (E) he shall not be entitled to require that security be provided in respect of the costs of the arbitration.

6.4 Appointment of an expert

If the Valuer acts as an expert:

- (A) he shall give the Landlord and the Tenant the opportunity both to make written representations to him and to comment on each other's' representations; and
- (B) another Valuer may replace him if he dies, becomes unwilling to act or incapable of acting or it becomes apparent that for any reason he will be unable to determine the matter referred to him within a reasonable time after his appointment; and
- (C) his decision shall be final and binding on the Landlord and the Tenant, save as to matters of law; and
- (D) his fees, including those of his appointment, shall be shared equally between the Landlord and the Tenant, unless he determines otherwise.

6.5 Time not of the essence

Time shall not be of the essence in relation to the operation of this clause 6.

PART 4: INSURANCE

7. Insurance obligations

7.1 Landlord's obligations

The Landlord shall insure in an insurance office of good repute and through such agency as the Landlord may from time to time determine or through underwriters at Lloyd's:

(A) the Building against damage or destruction by the Insured Risks in its full reinstatement cost including demolition, site clearance, hoarding and shoring up and all surveyor's, architect's, legal and other professional fees and statutory fees together in each case with Value Added Tax; and

- (B) four years' loss of the yearly rents, including the Principal Rent, in respect of the Building taking into account the Landlord's reasonable estimate of any increase in those rents upon any rent review or new letting; and
- (C) its public liability and employer's liability in respect of the Building.

7.2 Services Systems

The Landlord's obligations under clause 7.1 shall not include any obligation to insure the Services Systems against sudden and unforeseen damage and breakdown.

7.3 General insurance provisions

The Landlord's obligations under clause 7.1 shall be subject always to:

- (A) cover being normally available with insurance offices of good repute in the United Kingdom or through underwriters at Lloyd's on reasonable commercial terms; and
- (B) such Insurance Excesses, exclusions and conditions in the United Kingdom insurance market in which the insurance is placed applicable to the area in which the Building is situated or the type of building insured including any exclusions in respect of Terrorist Activity: and
- (C) any policy of insurance not being made void or voidable by any act or default of the Tenant; and
- (D) the Tenant notifying the Landlord of the reinstatement value of any Alterations in accordance with clause 10.4(H).

7.4 Insurance Rent

The Tenant shall pay to the Landlord a due proportion, to be fairly and properly determined by the Landlord, of:

- (A) the costs incurred by the Landlord in complying with its obligations in clause 7.1; and
- (B) the costs incurred by the Landlord in valuing the Building at reasonable intervals for insurance purposes (in any event not more than once every three years in the case of formal insurance valuation or annually in the case of desktop insurance valuations); and
- (C) any Insurance Excess; and
- (D) the Landlord's costs of preparing, making and settling any insurance claim under any policy of insurance maintained by the Landlord; and
- (E) all other sums payable by the Tenant to the Landlord under this clause 7.

7.5 Reinstatement of the Building

Following an Insurance Event, except to the extent that the insurance moneys are properly withheld due to the act or default of the Tenant and not made good by the Tenant under clause 7.11, the Landlord shall:

- (A) use all reasonable endeavours to obtain any planning permissions and other approvals required for the rebuilding and reinstatement of the Building, but without any obligation to appeal against a refusal to grant planning permission or any other approval; and
- (B) subject to those planning permissions and approvals being obtained, procure that all insurance moneys received from the insurer by virtue of clause 7.1(A) are laid out towards rebuilding or reinstating the Building with all convenient speed.

7.6 Manner of reinstatement

When rebuilding or reinstating the Building, the Landlord may use materials of a different quality, type or specification and may make reasonable changes in the original design, layout or specification of the Building so long as the extent of the Premises is not materially altered.

7.7 Impossibility of reinstatement

If the Landlord is unable to rebuild or reinstate the Building for any reason beyond its control, it shall not be under any obligation to do so and shall be entitled to retain for its own benefit all insurance moneys received or receivable under any policies of insurance maintained by it.

7.8 Commission and agency fees

The Landlord shall be entitled to retain for its own benefit any agency fee or other commission paid or allowed by the insurers.

7.9 Certificate of insurance

At the written request and cost of the Tenant, but not more than twice a year, the Landlord shall give the Tenant full details of the insurance of the Building.

7.10 Increased insurance costs

The Tenant shall not do or bring anything upon the Premises or any other part of the Building which may increase the premium payable for any policy of insurance effected by the Landlord under this clause 7. If the Tenant breaches this obligation, it shall pay the amount of any increased premium to the Landlord.

7.11 Invalidation of insurance

The Tenant shall not do or bring anything upon the Premises or any other part of the Building which may invalidate any policy of insurance effected by the Landlord under this clause 7. If any insurance moneys become irrecoverable due to the act or default of the Tenant, it shall pay the amount irrecoverable to the Landlord.

7.12 Compliance with insurer's requirements

The Tenant shall comply with the requirements and recommendations of the insurers of the Building so far as those requirements and recommendations have been communicated to the Tenant and relate to the Premises or the use of the Common Parts.

7.13 Notification

The Tenant shall promptly give notice to the Landlord if:

- (A) the Tenant becomes aware of an Insurance Event; or
- (B) the Tenant becomes aware of any matter which might invalidate or give the insurers of the Building grounds for avoiding any policy of insurance relating to the Building or which might increase the premium payable for any insurance of the Building maintained by the Landlord.

7.14 Vacating Premises

Following any Insurance Event, affecting the Premises the Tenant shall, if reasonably requested to do so by the Landlord, vacate those parts of the Premises which have been damaged or destroyed or which the Landlord reasonably requires access to in order to enable the Landlord to comply with its obligations in clause 7.5.

7.15 Tenant's insurance

The Tenant shall not insure the Landlord's interest in the Building against any of the Insured Risks or any other risk that the Landlord covenants to insure against under this clause 7. If the Tenant breaches this clause 7.15, it shall hold the benefit of any insurance moneys which it receives in respect of such interest on trust for the Landlord and shall pay such sums to the Landlord or, if the Landlord so requests, lay out such sums in reinstating the damage in respect of which those sums were paid.

7.16 Tenant's third party liability

The Tenant shall insure its public liability and employer's liability in respect of the Premises.

7.17 Exclusion of liability

The Landlord shall be under no liability to the Tenant or to any other person for any damage caused by or arising from any of the Insured Risks or Uninsured Risks except to comply with its obligations in this clause 7.

7.18 Uninsured Risks

This clause 7.18 shall apply from the date on which any Insured Risk becomes an Uninsured Risk but only in relation to the Uninsured Risk.

- (A) References to an Insured Risk becoming an Uninsured Risk shall, without limitation, include the application by insurers of an exclusion, condition or limitation to an Insured Risk to the extent to which such risk thereby is or becomes an Uninsured Risk.
- (B) If during the Term, the Building or the Premises, or a substantial part of them, shall be damaged or destroyed by an Uninsured Risk so as to make the Premises or a substantial part of them unfit for occupation and use or inaccessible ("<u>Uninsured Damage</u>") the Principal Rent and the Service Charge, or a proper proportion of them according to the extent of the damage, shall be suspended until the earlier of the date on which:
- (1) the Premises are reinstated and made accessible; or
- (2) the date on which this Lease is terminated pursuant to clause 7.18(E).

- (C) Within 12 months of the Uninsured Damage the Landlord shall give written notice to the Tenant ("<u>Election Notice</u>") stating whether or not it proposes to rebuild or reinstate the Premises or procure such rebuilding or reinstatement.
- (D) If the Election Notice states that the Landlord does propose to rebuild or reinstate the Premises (or procure the same) then the Uninsured Damage shall be deemed to be damage or destruction caused by an Insured Risk and the Landlord shall comply with the relevant provisions of clause 7.5 at its own cost to the extent that such clauses apply to the Premises.
- (E) If the Election Notice states that the Landlord does not propose to rebuild or reinstate the Premises or if no Election Notice is served within the period of 12 months referred to in clause 7.18(C) either party may determine this Lease by serving upon the other not less than one month's notice in writing.
- (F) Clauses 7.18(B) and (E) shall not apply if an Insured Risk shall have become an Uninsured Risk owing to the act or default of the Tenant or any person deriving title under the Tenant or their respective agents, employees, licensees or contractors.

PART 5: TENANT'S COVENANTS

8. Financial obligations

8.1 Payment of Outgoings

The Tenant shall pay and indemnify the Landlord against all Outgoings except for:

- (A) any tax, other than Value Added Tax, payable by the Landlord on the Rents; and
- (B) any tax on any dealing by the Landlord with its interest in the Premises or the Building.

8.2 Loss of rating relief

If the Tenant claims any rating relief during the Term for empty premises, it shall pay to the Landlord on demand an amount equal to any rating relief which the Landlord is unable to claim after the Term has ended as a result of the Tenant having made such a claim.

8.3 Rating Proposals

The following provisions shall apply in respect of any Rating Proposal:

(A) the Tenant shall notify the Landlord as soon as it is aware of any Rating Proposal made by a third party and shall not agree to any Rating Proposal or make any Rating Proposal itself without the prior written consent of the Landlord; and (B) unless the Tenant can demonstrate to the reasonable satisfaction of the Landlord that the Rating Proposal would increase the Outgoings the Tenant shall:

- 1. not object to any Rating Proposal made by the Landlord or to the outcome of any Rating Proposal; and
- 2. co-operate with the Landlord in making any Rating Proposal and give the Landlord such information as the Landlord reasonably requests; and
- 3. sign all consents, authorisations or other documents as the Landlord reasonably requests to give full effect to the Rating Proposal or its outcome.

8.4 Payment for Utilities

Unless already included in the Heat Charge, the Standing Charge and/or the Energy Charge, the Tenant shall pay for all Utilities used by the Tenant and any related meter rents, installation charges and connection charges.

8.5 To pay costs

The Tenant shall pay by way of additional rent on a full indemnity basis all costs, charges and expenses properly incurred by or on behalf of the Landlord and the Management Company (together with their own administrative and management expenses) in relation to:

- (A) the preparation and service of any notice or any proceedings under ss146 or 147 of the Law of Property Act 1925 or the Leasehold Property (Repairs) Act 1938, or any steps taken to effect peaceable re-entry of the Premises, whether or not forfeiture is avoided other than by relief granted by the court; and
- (B) the recovery of any arrears of the Rent or other sums due to the Landlord or the Management Company under this Lease, whether by action or otherwise and whether or not including the preparation and service of any notice pursuant to s.81 of the Tribunals, Courts and Enforcement Act 2007 and/or s.17 of the 1995 Act; and
- (C) the preparation and service of a schedule of dilapidations at any time during or within three months after the end of the Term; and
- (D) any action taken by the Landlord or the Management Company to ensure that the Tenant makes good any breach of the Tenant's Covenants; and
- (E) the application for consent to any matter for which the consent or approval of the Landlord is required under this Lease (but excluding any costs incurred in relation to the Landlord's consideration and/or approval of any initial fitting out works at the Premises proposed by the Tenant following the grant of this Lease) whether such consent is granted or not for any reason unless the Landlord acts unlawfully in refusing consent. Any costs payable under this clause 8.5(E) shall be payable to the extent only that they are proper and reasonable.

8.6 Indemnity

The Tenant shall keep the Landlord and the Management Company indemnified against any breach of the Tenant's Covenants or any act or default of the Tenant in relation to the Premises or the Building.

9. Repair and redecoration

9.1 Upkeep of the Premises

Subject to clause 9.2, the Tenant shall:

- (A) keep the Premises in good and substantial repair and condition; and
- (B) replace any landlord's fixtures and any carpets and other floor coverings in the Premises with new articles of a similar kind and quality if they become incapable of repair or, in the case of plant and equipment, if they cease to operate properly or in any event in the last three months of the Term; and
- (C) keep any tenant's fixtures in good and substantial repair and condition; and
- (D) keep the whole of the Premises properly cleaned with all the internal surfaces of all windows and glass being cleaned as often as reasonably necessary; and
- (E) redecorate the Premises internally in the fifth year of the Term and within the three months before the end of the Term.

9.2 Insurance Events

Clause 9.1 shall not apply to any damage to or destruction of the Premises which is an Insurance Event except to the extent that the insurance moneys are withheld due to the act or default of the Tenant.

9.3 Uninsured Damage

Clause 9.1 shall not apply to any Uninsured Damage (as defined in clause 7.18(B)) except to the extent that the relevant Insured Risk has become an Uninsured Risk due to any act or default of the Tenant or any person deriving title under the Tenant or any of their respective agents, employees, licensees or contractors.

9.4 Manner of redecoration

The Tenant shall carry out all redecoration with appropriate materials in a good, and workmanlike manner and in accordance with modern practice in colours and with materials previously approved in writing by the Landlord, such approval not to be unreasonably withheld or delayed. In the final three months of the Term, the Tenant shall use such colours (if different from existing) and materials as the Landlord reasonably requires.

9.5 Making good disrepair

The Tenant shall carry out any works required to remedy any breach of the Tenant's Covenants relating to the repair and condition of the Premises within three months after the service of any notice specifying the works required or immediately in case of emergency. In default, the Landlord with its contractors may enter and remain upon the Premises to carry out those works itself and all costs incurred by the Landlord shall be a debt payable on demand to the Landlord by the Tenant on a full indemnity basis.

9.6 Defective premises

The Tenant shall do everything necessary to comply with the Defective Premises Act 1972 and in particular shall:

- (A) promptly give notice to the Landlord upon becoming aware of any defect in the Premises or the Building which may give rise to a duty of care on the Landlord under that Act; and
- (B) not do any act or thing which might breach any duty of care on the Landlord under that Act; and
- (C) display and maintain on the Premises any notices which the Landlord may reasonably require.

10. Alterations

10.1 Prohibited alterations

The Tenant shall not carry out any alterations or additions to the Premises except as permitted in this clause 10.

10.2 Non-structural alterations

- (A) Subject to clause 11.2(B), the Tenant may carry out internal non-structural alterations and additions to the Premises without the prior written consent of the Landlord;
- (B) The Tenant may not carry out non-structural alterations or additions to the Premises which will result in adjustments to the Services Systems without the Landlord's prior written consent, which will not be unreasonably withheld or delayed provided that as a condition to the grant of such consent the Landlord may require the Tenant to:
 - 1. (subject to the Landlord proceeding diligently with the same) permit the Landlord to carry out the elements of the proposed alterations and/or additions which will result in adjustments to the Services Systems at the Tenant's reasonable expense; or
 - 2. enter into such covenants with the Landlord as the Landlord may reasonably require for the execution and supervision of the alterations and/or additions and such covenants with the Landlord as the Landlord may reasonably require for the reinstatement of the Premises before the end of the Term.

10.3 Electrical systems

The Tenant may make alterations to the electrical systems in the Premises which exclusively serve them without the consent of the Landlord if they are carried out in accordance with the standards prescribed by the electricity supply authority and the Institute of Electrical Engineers.

10.4 General obligations

In respect of all Alterations, the Tenant shall at its own cost and expense:

- (A) provide the Landlord in triplicate with plans, drawings and specifications showing:
 - 1. the layout of the Premises before the proposed Alterations are carried out; and
 - 2. the proposed Alterations; and
 - 3. the layout of the Premises following completion of the proposed Alterations; and
- (B) obtain all necessary consents for the Alterations; and
- (C) provide all information required by the insurers of the Building in order to obtain their approval to the Alterations; and
- (D) carry out the Alterations with good quality materials, in a good workmanlike manner in accordance with the terms of all consents obtained for the Alterations and the requirements of the local planning and other authorities; and
- (E) comply with the requirements of any Statute which affects the Alterations or the manner in which they are carried out; and
- (F) comply with the Fit-Out Guide; and
- (G) where the consent of the Landlord is required for the Alterations, carry out the Alterations to the reasonable satisfaction in all respects of the Landlord; and
- (H) give to the Landlord full details of the reinstatement value of the Alterations, excluding any tenant's fixtures forming part of the Alterations, for insurance purposes; and
- (I) provide the Landlord with three sets of as built plans, drawings and specifications on completion of the Alterations.

10.5 Reinstatement

In respect of all Alterations, the Tenant shall comply with the reinstatement provisions in clauses 20.2 and 20.3 at the end of the Term.

10.6 Effect of consent to Alterations

In consenting to any Alterations, the Landlord shall not guarantee:

- (A) the structural stability of either the Alterations or the Premises as altered by them; or
- (B) the suitability of any materials to be used in the Alterations; or
- (C) the compatibility of the Alterations with the Services Systems; or
- (D) that the Premises as altered will comply with the requirements of any Statute.

10.7 Indemnity

The Tenant shall indemnify the Landlord in respect of any matter arising out of the execution, retention and use of any Alterations.

11. Use of the Premises

11.1 Authorised Use

The Tenant shall use the Premises only for the Authorised Use.

11.2 Prohibited uses

The Tenant shall not use the Premises:

- (A) for the wholesale or retail sale of goods or any sale by auction; or
- (B) for any religious, public or political meeting; or
- (C) for any offensive, noxious or noisy trade, business or occupation; or
- (D) for illegal or immoral purposes; or
- (E) for the sale or production of alcohol; or
- (F) for residential purposes; or
- (G) as a club, sex shop, amusement arcade, betting office, staff agency or employment agency; or
- (H) the preparation or cooking of food, other than in those areas reasonably designated by the Landlord or the Management Company for those purposes; or
- (I) outside the Business Hours, except in accordance with clause 11.5.

11.3 Restrictions on use

The Tenant shall not:

- (A) bring any Hazardous Material onto the Building; or
- (B) discharge anything into the Conduits which is or may become corrosive or harmful or cause any blockage or destruction of them; or
- (C) allow any animals (other than guide dogs) in the Building; or
- (D) obstruct the Office Common Parts or any means of escape or other facilities serving the Premises or the Building; or
- (E) do anything on the Premises or the Building or install or operate any machinery or equipment which may in the reasonable opinion of the Landlord be or become a legal nuisance, damage or unreasonable annoyance to the Landlord or the occupiers of the Building or any Adjoining Premises or be unduly noisy or cause vibration or electrical or other interference; or

- (F) overload the floors or suspend any excessive weight from the ceilings or walls of the Premises or overload the lifts serving the Premises; or
- (G) overload the electrical systems in the Premises or the Building or connect any equipment to the electrical systems except in accordance with the design and specification of the electrical systems; or
- (H) install or operate any machinery or mechanical equipment in or on the Premises, except for office machinery and equipment used by the Tenant in the normal operation of its business, without the prior written consent of the Landlord, such consent not to be unreasonably withheld or delayed; or
- (I) erect, exhibit or hang any signs, advertisements, placards, flags, posters or, aerials, flags, poles, masts or satellite dishes or any other thing whatsoever on the exterior of the Premises or any other parts of the Building unless permitted to do so under clause 3.1; or
- (J) store any refuse in the Premises except in suitable containers for that purpose with all refuse being removed from the Premises as often as is reasonably necessary.

11.4 Regulations

The Tenant shall:

- (A) comply with any requirements or regulations made in respect of the Utilities by the supply companies; and
- (B) comply with the lawful requirements and recommendations of the local fire officer in respect of the Premises and the Building or their use; and
- (C) operate the Services Systems within and exclusively serving the Premises in accordance with the manufacturer's operating guidelines and reasonable recommendations provided to the Tenant; and
- (D) comply with all reasonable regulations made by the Landlord or the Management Company at any time for the management of the Building and the exercise of the rights granted to the Tenant under clause 3.1.

11.5 Use of the Premises outside Business Hours

The Tenant may, on written notice to the Landlord, use the Premises and have access to them outside Business Hours subject to the Tenant observing the following terms and conditions:

- (A) the Tenant shall use only those parts of the Office Common Parts as the Landlord may reasonably designate for the purpose of obtaining access to the Premises; and
- (B) the Tenant shall comply with the proper and reasonable requirements of the Landlord regarding the use of the Office Common Parts and the security of the Building; and
- (C) the Tenant shall comply with clause 23.16.

11.6 Keyholders

The Tenant shall:

- (A) ensure that, at all times, the Landlord has details of the names, home numbers and home addresses of at least two keyholders of the Premises; and
- (B) instruct the keyholders to provide the Landlord on request with a set of keys and passes and any necessary codes for key-pads or other equipment necessary to enable the Landlord to obtain access to the Premises for security purposes or in an emergency.

11.7 No warranty as to use

The Landlord gives no warranty that the Authorised Use is or will remain a permitted use under the Planning Acts.

12. Assignment

12.1 No assignment of part

The Tenant shall not assign part only of the Premises.

12.2 Assignment of whole

The Tenant shall not assign the whole of the Premises unless:

- (A) the conditions specified (for the purposes of s.19(1A) Landlord and Tenant Act 1927) in clause 12.3 are fulfilled; and
- (B) the circumstances specified (for the purposes of s.19(1A) Landlord and Tenant Act 1927) in clause 12.4 do not apply; and
- (C) the Tenant obtains the prior written consent of the Landlord which shall not be unreasonably withheld or delayed.

12.3 Conditions for assignment

The Landlord shall not be required to consent to any assignment unless on or before the date of the assignment:

- (A) the Tenant has paid to the Landlord the whole of any arrears of Principal Rent; and
- (B) the Landlord receives a direct covenant by way of a guarantee and indemnity that the assignee will comply with the Tenant's Covenants from the date of the assignment until the date the assignee is released from its obligations under the 1995 Act from:
 - 1. the Tenant by way of an Authorised Guarantee Agreement; and
 - if reasonably required by the Landlord, any guarantor or guarantors who shall not be a Current Guarantor and who shall be resident or incorporated in the United Kingdom whose financial status shall be acceptable to the Landlord acting reasonably; and

- (C) the assignee enters into a direct covenant with the Landlord to comply with the Tenant's Covenants from the date of the assignment until the date the assignee is released from its obligations under the 1995 Act; and
- (D) if asked to by the Landlord, any guarantor of the Tenant enters into a direct covenant with the Landlord by way of a guarantee and indemnity that the Tenant will comply with its Authorised Guarantee Agreement; and
- (E) if reasonably required by the Landlord, a rent deposit of not less than six months' Principal Rent, together with Value Added Tax (if any), shall be provided to the Landlord on such terms and for such period as the Landlord shall reasonably require.

12.4 Prohibited assignments

The Landlord shall not be required to consent to any assignment if the assignee is:

- (A) a Current Guarantor; or
- (B) a Group Company of the Tenant unless the Landlord is provided with evidence reasonably satisfactory to it that the covenant strength of the assignee is not less than that of the Tenant at the date of the assignment disregarding any Authorised Guarantee Agreement given by the Tenant to the Landlord; or
- (C) a company incorporated in or an individual resident in a jurisdiction outside the United Kingdom in respect of which there no applicable treaty for the mutual enforcement of civil judgments unless the Landlord is reasonably satisfied that a judgment obtained in England and Wales against the assignee can be enforced in the relevant jurisdiction; or
- (D) in the reasonable opinion of the Landlord, not of sufficient financial standing to comply with the Tenant's Covenants; or (E) a person enjoying diplomatic or sovereign immunity.

12.5 General provisions

On any assignment of this Lease:

- (A) any guarantee and indemnity to be given to the Landlord shall be in the form of schedule 5 incorporating such changes as the Landlord may reasonably require to meet the particular circumstances of the assignment; and
- (B) the Landlord shall prepare all documents containing covenants to be given to it; and
- (C) clause 12.2 shall operate without prejudice to the right of the Landlord to withhold consent in any other circumstances where such withholding of consent would be reasonable or to impose any further conditions upon the grant of consent where it is reasonable to do so.

13. Underletting

13.1 Defined terms

In this clause 13 and in schedule 6, unless the contrary intention appears:

"Access Areas"

means any corridors, lobbies, landings, staircases and lifts within or outside the Premises which are intended to be used in common by the Tenant and any undertenant.

"Permitted Parts"

means any part of the Premises which has or will have a means of access from any Access Areas and the extent of which has been approved by the Landlord, such approval not to be unreasonably withheld or delayed.

"Underlet Premises"

means those parts of the Premises let by any underlease.

13.2 Restrictions on underletting

The Tenant shall not underlet the whole or any part of the Premises except as permitted under this clause 13.

13.3 Underletting of whole

The Tenant shall not underlet the whole of the Premises without the prior written consent of the Landlord which shall not be unreasonably withheld or delayed if the conditions set out in this clause 13 are satisfied.

13.4 Underletting of part

The Tenant may underlet a Permitted Part if:

- (A) following the grant of the underlease, there will be no more than two occupiers of the whole or any part of the Premises; and
- (B) the conditions set out in this clause 13 are satisfied; and
- (C) the Tenant obtains the prior written consent of the Landlord, such consent not to be unreasonably withheld or delayed.

13.5 Undertenant's covenants

Before the grant of any underlease, the Tenant shall procure that the undertenant enters into direct covenants with the Landlord in the form set out in Part 2 of schedule 6.

13.6 Undertenant's guarantor

If reasonably required by the Landlord, before the grant of any underlease a guarantor or guarantors resident or incorporated in the United Kingdom whose financial standing has been approved by the Landlord acting reasonably shall enter in to a direct covenant with the Landlord guaranteeing that the undertenant shall comply with the covenants given to the Landlord under clause 13.5.

13.7 General provisions

On any underletting:

- (A) before the undertenant enters into the tenancy created by the underlease, or, if earlier, before the undertenant becomes contractually bound to enter into the tenancy created by the underlease:
- (1) the Tenant shall procure that the underlease is validly excluded from the security of tenure provisions of ss24 to 28 (inclusive) of the 1954 Act in accordance with the Order; and
- (2) the Tenant shall produce to the Landlord adequate evidence of such valid exclusion as referred to in Clause 14.7(A)(1); and
- (B) the underlease shall contain the provisions set out in Part 1 of schedule 6 and
- (C) any guarantee and indemnity to be given to the Landlord shall be in the form of schedule 5 incorporating such changes as the Landlord may reasonably require to meet the particular circumstances of the underletting and on the basis that paragraph 4 of schedule 5 shall only apply if the Landlord is the direct landlord of the undertenant when the guarantee and indemnity is enforced; and
- (D) the Landlord shall prepare all documents containing covenants to be given to it.

13.8 Observing the terms of the underlease

The Tenant shall:

- (A) not vary the terms of the underlease except to permit a change of use in accordance with the terms of the underlease; and
- (B) enforce the undertenant's covenants and not waive any breach of them; and
- (C) not grant any licence, consent or permission under the terms of the underlease without the prior written consent of the Landlord, which shall not be unreasonably withheld or delayed where the Landlord could not unreasonably withhold or delay its consent to a similar application made under the terms of this Lease; and
- (D) procure that the yearly rent payable is reviewed under each underlease in accordance with its terms and in connection with each rent review it shall:
 - 1. not agree the amount of the reviewed rent without the prior written consent of the Landlord, such consent not to be unreasonably withheld or delayed; and
 - 2. not agree the identity of any third party appointed to determine the reviewed rent or the capacity in which that party acts without the prior written consent of the Landlord, such consent not to be unreasonably withheld or delayed; and
 - 3. incorporate as part of its submissions or representations to any third party appointed to determine the reviewed rent such submissions or representations as the Landlord may reasonably require; and
 - 4. keep the Landlord informed about the progress of the rent review; and

- (E) procure that the rents reserved by any underlease are not commuted in whole or in part or payable more than one quarter in advance; and
- (F) not agree to any reduction in any rent payable under any underlease; and
- (G) not exercise any right of re-entry under the underlease or accept the surrender of the whole of any Underlet Premises without the prior written consent of the Landlord, such consent not to be unreasonably withheld or delayed; and
- (H) not accept the surrender of part only of any Underlet Premises.

14. Sharing occupation, etc.

14.1 Restrictions imposed

Except where there is an assignment or underletting permitted on the terms of this Lease, the Tenant shall not share occupation of the whole or any part of the Premises, part with the possession of them or hold this Lease on trust for any other person.

14.2 Group Companies

Clause 14.1 shall not prevent the Tenant sharing the occupation of the whole or any part of the Premises with any Group Company of the Tenant provided that:

- (A) the sharing of occupation shall cease if that company ceases to be a Group Company of the Tenant; and
- (B) no relationship of landlord and tenant shall be created between the Tenant and that company; and
- (C) prior to occupation being granted, the Tenant shall give written notice to the Landlord of the identity of that company and its relationship to the Tenant.

15. Charging

15.1 Part of the Premises

The Tenant shall not charge or create any other security interest over part only of the Premises.

15.2 The whole of the Premises

The Tenant shall not charge or create any other security interest over the whole of the Premises without the prior written consent of the Landlord, such consent not to be unreasonably withheld or delayed.

16. Notification of dispositions

Within one month after any assignment, charge or underletting of the Premises or the assignment of any underlease or the grant of any sub-underlease, however remote, the Tenant shall give notice in writing to the Landlord of that disposition together with certified copies of all the documents giving effect to it (and in the case of an underlease containing an agreement excluding the effect of §§24 to 28 (inclusive) of the 1954 Act, certified copies of the notices, declarations or statutory declarations required to be respectively served and made under the 1954 Act as amended by the Order), and shall pay the Landlord a reasonable fee of not less than £35 for its registration.

16.1 Notice of interests

The Tenant shall on request by the Landlord give the Landlord full details of:

- (A) all underleases of the Premises, however remote, including the rents payable under them and such other details as the Landlord may reasonably require in respect of them; and
- (B) all Group Companies of the Tenant or any undertenant (however remote) in occupation of the Premises and their relationship to the Tenant or undertenant, as the case may be.

17. <u>Legislation</u>

17.1 Planning

The Tenant shall comply with the Planning Acts in so far as they relate to the Premises and the use of the Common Parts and shall not:

- (A) make any application for planning permission or any application for a determination that planning permission is not required for any Alterations or any change of the Authorised Use without the prior written consent of the Landlord. Consent shall not be unreasonably withheld or delayed if the application relates to a matter for which the Landlord cannot unreasonably withhold or delay its consent under this Lease; and
- (B) carry out any Alterations or change the Authorised Use until all necessary planning permissions have been obtained and approved in writing by the Landlord. Subject to clause 17.2, the Landlord shall not unreasonably withhold or delay its approval where it has given consent to the application for planning permission being made.

17.2 Approval of permissions

The Landlord shall be entitled to:

- (A) impose reasonable conditions to the giving of its approval; and
- (B) withhold its approval to any planning permission if any condition contained in or omitted from it or its duration would, in the reasonable opinion of the Landlord, have or be likely to have an adverse effect on the value of the Landlord's interest in the Building or any Adjoining Premises at any time during or after the end of the Term.

17.3 Statutes

The Tenant shall comply with every Statute which affects the Premises and the Common Parts or their use and occupation and shall carry out and maintain at its own cost and expense all works and arrangements required under any Statute whether imposed on the Tenant or any other person so far as they relate to the Premises.

17.4 CDM Regulations

The Tenant shall comply in all respects with the CDM Regulations in so far as they relate to the Premises and, without limitation, shall at its own cost and expense:

- (A) notify the Health and Safety Executive of any Alterations; and
- (B) maintain and update the Health and Safety File as necessary and provide copies of it to the Landlord and any person entitled to inspect it on request by those persons; and
- (C) where there is more than one client in relation to a project, make a written election to be treated as the only client for the purposes of the CDM Regulations and provide the Landlord with a copy of the election, (and, where the Landlord is a client, the Landlord hereby consents to such election).

17.5 Statutory notices

The Tenant shall give to the Landlord a copy of every notice, order, direction, licence, consent or permission relating to the Premises made or given under any Statute within seven days of its receipt, or sooner in cases of emergency. If required by the Landlord, the Tenant shall at the joint cost of the parties cost and expense make or join the Landlord in making such objections, applications or representations against or in respect of them as the Landlord shall reasonably require save where it would be against the interest of the Tenant's use of the Premises or business carried out at the Premises in accordance with the terms of this Lease.

18. Third party rights

18.1 Loss of existing rights benefiting the Premises

The Tenant shall not stop up, darken or obstruct any window or other opening belonging to the Premises or do any other act or thing which may lead to any rights benefiting the Premises or the Building being lost.

18.2 Creation of new rights over the Premises

The Tenant shall not permit or allow any encroachment to be made which may lead to rights being acquired by any person over the Premises or the Building.

18.3 Notification

The Tenant shall notify the Landlord of any of the matters referred to in this clause 18 as soon as the same come to the notice of the Tenant and shall take such action as the Landlord reasonably requires to prevent any rights being acquired over the Premises or the Building or any rights benefiting the Premises or the Building lost.

19. Title matters

19.1 Covenants restricting use

The Tenant shall not enter into any covenant in favour of any person other than the Landlord relating to the Premises or require any assignee or undertenant to give covenants which would restrict the use of the Premises to a greater extent than the restrictions on use contained in this Lease.

19.2 Existing title matters

The Tenant shall comply with the conditions, covenants, restrictions and other matters contained or referred to in the deeds and documents specified in schedule 4

20. End of the Term

20.1 Return of the Premises

At the end of the Term, the Tenant shall return the Premises to the Landlord:

- (A) with vacant possession; and
- (B) repaired, cleaned and maintained in accordance with the Tenant's Covenants.

20.2 Reinstatement at the end of the Term

Subject to clause 20.3, at the end of the Term the Tenant shall:

- (A) remove any buildings or other works in respect of which any planning permission, bye-law or other consent may have been granted for a limited period only; and
- (B) remove any moulding, sign or painting of the name or business of the Tenant or any undertenant or other occupiers of the Premises and all tenant's fixtures and chattels; and
- (C) remove any Hazardous Material unless it was present at the time the Tenant first took possession of the Premises; and
- (D) reinstate all Alterations made to the Premises and restore the Premises to the state and condition in which the Tenant first took possession of them being to the Category A specification as specified in the Base Specification.

The Tenant shall carry out all works on the same terms as those contained in clause 10.4 as if the works to be carried out were Alterations for which the consent of the Landlord has been obtained and the Tenant shall make good all physical damage to the Building caused in the exercise of these obligations to the Landlord's reasonable satisfaction.

20.3 Waiver of reinstatement

Clause 20.2 shall not apply to the extent that the Landlord notifies the Tenant in writing at any time before the last six months of the Term that it does not require any works of reinstatement specified in that notice to be carried out. If the Landlord serves notice under this clause 20.3, the Tenant shall not carry out the works specified in that notice.

20.4 Removal of Tenant's effects

The Tenant irrevocably appoints the Landlord as its agent to store and dispose of any tenant's fixtures and chattels left by the Tenant on the Premises for more than twenty one days after the end of the Term. The Landlord shall be entitled to dispose of these items on such terms as it sees fit without being liable to the Tenant except to account for the net proceeds of sale after deducting any costs of storage and any other proper expenses reasonably incurred by the Landlord.

20.5 Service charge payments

At the end of the Term, the Tenant shall pay to the Landlord all unexpended amounts, if any, collected from any undertenants of the Premises by way of service charge together with any interest earned on such moneys. In respect of all moneys actually received by the Landlord:

- (A) the Landlord shall indemnify the Tenant from and against any lawful claims from any undertenants in respect of the unexpended moneys; and
- (B) this clause 20.5 shall be taken into account in determining any liability of the Tenant in respect of any breach of the Tenant's Covenants.

20.6 Exclusion of compensation

Subject to the 1954 Act, the Tenant shall not be entitled to any compensation under any Statute or otherwise at the end of the Term.

20.7 Land Registry forms

(A) Obligation to register Lease at Land Registry

The Tenant will:

- 1. use its reasonable endeavours to procure that the Tenant is registered at HM Land Registry as proprietor of the Lease as soon as reasonably possible; and
- 2. use its reasonable endeavours to procure that all rights granted or reserved by the Lease are properly noted against the affected titles; and
- 3. deliver to the Landlord, within 10 working days of registration, official copies of the registered title evidencing that the Tenant is the registered proprietor of the Lease.

20.8 EPC Regulations

- (A) Not to effect any alteration nor replace any fittings lighting or any machinery plant or other equipment or item which consumes energy at the Building without ensuring that any such activity:
 - 1. does not adversely affect the EPC Rating at the date of this Lease; and
 - 2. is consistent with the EPC Policy provided to the Tenant.
- (B) To allow the Landlord and the Management Company full access to the Premises and to the Tenant's records, data and documents which relate to energy consumption at the Building for the purpose of the Landlord's and the Management Company's development, implementation and revision of the EPC Policy and the production of any EPC and any Recommendation Report and any DEC from time to time.

- (C) To take all reasonably practicable steps to co-operate with the Landlord or the Management Company or their representatives in the preparation of any of the Energy Documents.
- (D) To adopt such reasonable practicable recycling, waste control and disposal arrangements, cleaning methods energy use and other reasonable strategies as conform with the EPC Policy and provide such reasonable evidence of the same to the Landlord or the Management Company upon reasonable request.
- (E) Not to commission nor suffer nor permit the commissioning of any EPC for the Premises save where it is required to do so to comply with Statute and in the event that an EPC shall be required to be commissioned, the Tenant will not do so (nor suffer or allow the same to be done) without previously informing the Landlord and allowing the Landlord and the Management Company (should the Landlord so request) to be a full participant in the exercise (on the basis that the Landlord and the Management Company shall have no right to obstruct its prompt conclusion).
- (F) To take all such reasonable actions as requested by the Landlord or the Management Company to conform with the proposals made in any Recommendation Report.

PART 6: LANDLORD'S COVENANTS AND MANAGEMENT COMPANY'S COVENANTS

21. <u>Landlord's obligations</u>

21.1 Quiet enjoyment

Subject to clause 21.2 the Tenant shall be entitled quietly to enjoy the Premises throughout the Term without any interruption by the Landlord or by any person lawfully claiming under or in trust for the Landlord.

21.2 No derogation from grant

The exercise by the Landlord or the Management Company or any other person of any right reserved in this Lease shall not be in derogation of the Landlord's grant nor be a breach of the Landlord's obligation in clause 21.1.

21.3 Guarantee of the Management Company's Covenants

- (A) The Landlord covenants with the Tenant that the Management Company or the Landlord will comply with the Management Company's Covenants.
- (B) At any time during the Term the Landlord may on giving written notice to the Tenant and the Management Company determine that it will undertake or exercise the Management Company's Covenants with effect from such date as shall be specified in that notice.

21.4 Limitation on liability

Nether the Landlord nor the Management Company shall be liable to the Tenant or to any persons deriving title under the Tenant or its or their respective licensees, agents or employees in respect of any loss or damage caused by or arising from any breach of:

(A) the Landlord's Covenants, save for its obligations in clause 7; or

- (B) the Management Company's Covenants; or
- (C) obligations on the part of the Landlord and/or the Management Company express or implied under Statute unless and until the Tenant has notified the Landlord or the Management Company of the breach and the Landlord or the Management Company has failed to remedy it within a reasonable time. If the Landlord or the Management Company fails to remedy the breach within a reasonable time, the Landlord or the Management Company shall be liable to compensate the Tenant only for loss or damage sustained by the Tenant after such reasonable time has elapsed.

21.5 Exclusion of liability

Neither the Landlord nor the Management Company shall be liable to the Tenant or to any persons deriving title under the Tenant or its or their respective licensees, agents or employees in respect of any loss or damage which is not attributable to the wilful default or negligence of the Landlord caused by or arising from:

- (A) any Insurance Event where the Landlord complies with its obligations under clause 7; or
- (B) any act or omission of any employee of the Landlord or the Management Company other than in the performance of duties imposed by the Landlord or the Management Company; or
- (C) any act or omission of any other tenant of the Landlord or any person deriving title under such a tenant or any licensee; or
- (D) any approval or consent given in respect of any drawings, plans, specifications or other documents prepared by or on behalf of the Tenant relating to any Alterations; or
- (E) any failure, interruption or delay in the performance of the Landlord's Covenants and the Management Company's Covenants from any cause or circumstance beyond the control of the Landlord or the Management Company including, without limitation, mechanical breakdown, failure, malfunction, shortages of fuel or materials or labour disputes; or
- (F) any failure, interruption or delay in the supply of any of the Services from any necessary maintenance, servicing, repair or replacement of the Conduits, any Services Systems or any other plant, equipment or machinery used in the provision of the Services.

PART 7: SUPERIOR LEASES

22. Superior Leases

22.1 Tenant's obligations

The Tenant shall comply with the obligations contained in the Superior Leases which the tenants of the Superior Leases covenant to comply with insofar as they are applicable to the Premises, except for the obligation on the tenants of the Superior Leases to pay the rents reserved by them.

22.2 Landlord's obligations

The Landlord shall:

- (A) pay the rents reserved by the Superior Leases; and
- (B) comply with the obligations on the part of the tenants of the Superior Leases which are not the responsibility of the Tenant under this Lease; and
- (C) at the request and cost of the Tenant on a full indemnity basis, including reasonable security for costs paid in advance if the Landlord reasonably requires this, use all reasonable endeavours to procure that the Superior Landlords comply with the obligations on the part of the landlords contained in the Superior Leases.

22.3 Consents

Without prejudice to the terms of this Lease, where the consent or approval of the Landlord is required to any act or thing:

- (A) it shall be a condition precedent to the grant of that consent or approval that, if required under the Superior Leases, the consent or approval of Superior Landlords is obtained; and
- (B) where the Landlord is under an obligation not unreasonably to withhold or delay its consent or approval, the Landlord shall, at the cost of the Tenant on a full indemnity basis, apply for and use all reasonable endeavours to obtain the consent or approval of the Superior Landlords where this is required under the Superior Leases; and
- (C) references to the consent or approval of the Superior Landlords shall include the consent or approval of any person from whom the Superior Landlords are under an obligation to obtain consent or approval.

PART 8: SERVICES AND SERVICE CHARGE

23. Services and Service Charge

23.1 Defined Terms

In this clause 23 and schedule 7 unless the contrary intention appears:

"Accounting Period"

means the period from and including 1 January in any year to and including 31 December in the same year. The Management Company acting reasonably may from time to time, by notice in writing to the Tenant, change:

- (A) the date on which each Accounting Period commences; or
- (B) the duration of the Accounting Period.

"Advance Service Charge"

means, in respect of each Accounting Period, the reasonable sum notified by the Management Company to the Tenant that is payable by the Tenant on account of the Service Charge for that Accounting Period.

"Balancing Payment"

means, in respect of each Accounting Period, the sum payable by or to the Tenant, as the case requires, equal to respectively the amount by which the Service Charge for that Accounting Period either exceeds or is less than the aggregate of the Advance Service Charge paid for that Accounting Period and any sums paid under clause 23.6 during that Accounting Period.

"Items of Expenditure"

means the items of expenditure set out in Part 44 of schedule 7.

"Service Charge Expenditure"

means, in respect of each Accounting Period, the aggregate of the following sums after:

- (A) the Service Costs for that Accounting Period whether or not incurred prior to the Term Commencement Date; and
- (B) any Service Costs included under clause 23.9; and
- (C) such sum or sums by way of reasonable provision for the Service Costs which the Management Company expects will be incurred within the period of three years immediately after the expiration of that Accounting Period.

"Service Charge Proportion"

means the proportion reasonably determined by the Management Company of the Service Charge Expenditure as a fair apportionment of the Service Charge Expenditure as between the Landlord (in respect of any Lettable Areas which are unlet), the Tenant and the other tenants and occupiers of the Lettable Areas.

"Service Charge Statement"

means a statement showing for the relevant Accounting Period:

- (A) the Service Charge Expenditure; and
- (B) the Service Charge; and
- (C) the Advance Service Charge paid by the Tenant; and
- (D) any sums paid by the Tenant under clause 23.6; and
- (E) the Balancing Payment.

"Service Costs"

means, in respect of each Accounting Period, the proper and reasonable expenditure incurred by or on behalf of the Management Company during that Accounting Period in providing the Services and in respect of the Items of Expenditure.

23.2 Provision of the Services

The Landlord shall provide the Services during Business Hours for the benefit of the Building in accordance with the principles of good estate management.

23.3 Service Charge

The Service Charge payable by the Tenant for each Accounting Period shall be the Service Charge Proportion of the Service Charge Expenditure.

23.4 Service Charge Proportion

In determining the Service Charge Proportion, the Management Company:

(A) may allocate to the Tenant or to the tenants or occupiers of any other Lettable Areas the whole or a part, as the Management Company reasonably determines, of the cost of providing or incurring any particular Services or Items of Expenditure; and

- (B) shall not increase or alter the Service Charge Proportion by reason only that:
 - 1. any part of the Lettable Areas is unlet or let on terms which do not require the tenant or other occupier to pay a service charge; or
 - 2. any tenant or other occupier of the Lettable Areas is in default in the payment of any Service Charge Expenditure.

23.5 Advance Service Charge

The Advance Service Charge shall be payable by equal quarterly payments in advance on the Quarter Days and, if the Management Company requires, by banker's standing order. If the Tenant has not been notified of the Advance Service Charge for any Accounting Period before the commencement of that Accounting Period:

- (A) until it is notified the Tenant shall continue to pay the Advance Service Charge at the rate payable for the Accounting Period immediately before the relevant Accounting Period; and
- (B) within seven days of notification to the Tenant of the Advance Service Charge for the relevant Accounting Period, the Tenant shall pay to the Management Company any increase in that Advance Service Charge.

The first payment of Advance Service Charge for the period from and including the date of this Lease or, if earlier, the date the Tenant took occupation of the Premises to and including the day immediately before the next Quarter Day, shall be paid on the date of this Lease.

23.6 Additional payments

If from time to time the Management Company wishes or is required to incur any Service Costs and the money held by the Management Company on account of the Service Costs is insufficient for this purpose, the Management Company shall be entitled to demand a further reasonable sum on account of the Service Charge. The Tenant shall pay such sum within seven days of demand.

23.7 Service Charge Statement

As soon as reasonably practicable after the end of each Accounting Period the Management Company shall deliver to the Tenant the Service Charge Statement ,the Service Costs in which have been certified or audited, as the case requires, at the Management Company's discretion by a surveyor or accountant employed by the Management Company, the Landlord, any Group Company of the Landlord or the Management Company or the Landlord's or the Management Company's managing agents.

23.8 Balancing Payment

If the Balancing Payment shown on a Service Charge Statement is payable by:

(A) the Tenant to the Management Company, the Tenant shall pay the Balancing Payment to the Management Company within twenty one days of the delivery to the Tenant of the Service Charge Statement; or

(B) the Management Company to the Tenant, the Balancing Payment shall be set off against any sum payable by the Tenant under this clause 23 and, in the case of the final Service Charge Statement, any part of the Balancing Payment which has not been so set off shall be paid by the Management Company to the Tenant within fourteen days of the delivery to the Tenant of that Service Charge Statement.

23.9 Correction of Service Charge calculations

If the Management Company omits any Service Costs from the Service Charge Statement for the Accounting Period in which they were incurred, the Management Company shall be entitled to include those Service Costs in the Service Charge Statement for any subsequent Accounting Period.

23.10 Meter readings

In the calculation of the Service Costs and the Service Charge Proportion, the gas, electricity, water and any other services consumed in the Building shall be determined by readings taken from time to time by or on behalf of the Management Company from the appropriate check meters in the Building.

23.11 Disputes

If the Tenant has paid in full all sums which the Management Company requires it to pay under this clause 23, including any disputed sums, it may, subject to clause 23.12, by written notice served on the Management Company within one month of delivery to the Tenant of a Service Charge Statement specify anything in it which the Tenant disputes.

If the Management Company and the Tenant cannot resolve the dispute by agreement within two months of the service of the Tenant's notice it shall be resolved by the decision of an expert who shall be appointed and act in accordance with clause 23.13.

Time is of the essence in relation to the service of a notice under this clause 23.11.

23.12 Amount of Service Charge Expenditure

The Tenant shall not be entitled to dispute the Service Charge Expenditure shown in a Service Charge Statement by reason that the Services or Items of Expenditure could have been provided or incurred at a cost less than that in fact incurred by the Management Company.

23.13 Appointment of expert

The expert shall be of at least ten years professional standing and appointed by agreement between the Management Company and the Tenant or, in the absence of agreement, by the President for the time being of the Royal Institution of Chartered Surveyors on the application of the Management Company or the Tenant. If he is unable or unwilling to make the appointment, it shall be made by the Vice President or the next senior officer willing and able to do so. If there is no such officer available, it shall be made by an officer of such other professional body as is nominated by the Management Company.

Clause 6.4 shall apply to the expert in the same way that it applies to the Valuer but his costs shall be paid by the Tenant unless the expert decides otherwise and he shall give his decision in writing to the Management Company and the Tenant.

23.14 Expert's decision

If in resolving a dispute pursuant to clause 23.13 the expert substitutes a different Balancing Payment for the Balancing Payment shown in a Service Charge Statement:

- (A) an appropriate payment or set off, as the case requires, shall be made between the Management Company and the Tenant in accordance with clause 23.8(A) or 23.8(B) so that they are in the position they should have been in had the different Balancing Payment been substituted for that shown in the relevant Service Charge Statement before its delivery to the Tenant; and
- (B) the payment or set off shall be made within fourteen days of the delivery of the expert's decision to the relevant party.

23.15 Variation to Services

If the Management Company reasonably determines that this is desirable for the better use and enjoyment of the Building by its occupiers or for the more efficient management of the Building it shall from time to time be entitled in its discretion to withhold, add to, extend, vary or make any alteration in the Services or the manner in which they are provided and shall notify the Tenant accordingly.

23.16 Extension in the Business Hours

If the Tenant uses the Premises outside the Business Hours under clause 11.5, the Tenant shall pay to the Management Company within seven days of demand and in addition to the Service Charge the whole or a part, as reasonably determined by the Management Company, of the costs and expenses attributable to the provision of any Services outside the Business Hours and any associated Items of Expenditure.

23.17 Continuing effect

This clause 23 shall continue in effect after the end of the Term but only in respect of any period before the end of the Term.

23.18 Environmental

Notwithstanding any other provision in this clause 23 or elsewhere in this Lease the Tenant shall have no liability by way of Service Charge or otherwise to make any payment or contribution towards the cost of compliance with any environmental regulations arising out of any Hazardous Material at the Building unless the presence of such Hazardous Material was caused by the Tenant.

PART 9: GUARANTOR'S OBLIGATIONS

24. Tenant's Guarantor

24.1

The Tenant's Guarantor covenants with the Landlord that the Tenant will comply with the Tenant's Covenants until the Tenant is released from them under the terms of the 1995 Act. This obligation is a guarantee and indemnity for the purposes of schedule 5 and incorporates its provisions.

PART 10: GENERAL PROVISIONS

25. Tenant's option to renew

25.1

If the Tenant wishes to enter into a further Lease of the Premises for a term of five years commencing on and including 22 June 2026 upon the same terms and conditions as this Lease (including the exclusion of sections 24 to 28 (inclusive) of the 1954 Act in accordance with section 38A of the 1954 Act) except as varied by the provisions of Schedule 8 (the <u>"Further Lease"</u>) the Tenant must first serve written notice on the Landlord indicating such intention (the <u>"Intention Notice"</u>) not more than twelve nor less than eight months before the Expiry of the Term.

25.2

Within twenty eight days of receipt of the Intention Notice the Landlord will:

- (A) serve notice on the Tenant in relation to the tenancy to be granted by the Further Lease in accordance with section 38A(3)(a) of the 1954 Act; and
- (B) serve notice on the Tenant's Guarantor in relation to the tenancy to be entered into by the Tenant's Guarantor pursuant to paragraph 4.2 of Schedule 5 of the Further Lease in accordance with section 38A(3)(a) of the 1954 Act.

25.3

If:

- (A) the Landlord has served notice on the Tenant in accordance with clause 25.2; and
- (B) the Tenant has made the appropriate declaration or statutory declaration confirming receipt of such notice in accordance with schedule 2 to the Order (as required to give effect to the agreement by the Landlord and the Tenant to exclude the provisions of sections 24 to 28 (inclusive) of the 1954 Act to be contained in the Further Lease in accordance with the provisions of section 38A of the 1954 Act); and
- (C) the Tenant's Guarantor has made the appropriate declaration or statutory declaration confirming receipt of such notice in accordance with schedule 2 to the Order (as required to give effect to the agreement by the Landlord and the Tenant's Guarantor to exclude the provisions of sections 24 to 28 (inclusive) of the 1954 Act to be contained in the Further Lease in relation to the tenancy to be entered into by the Tenant's Guarantor pursuant to paragraph 4.2 of Schedule 5 of the Further Lease in accordance with the provisions of section 38A of the 1954 Act); and

then the Tenant may elect to take the Further Lease by written notice to that effect (the "Option Notice") served on the Landlord not less than six months before the Expiry of the Term.

25.4

The Landlord and the Tenant agree to use all reasonable endeavours to procure the satisfaction of the conditions contained in clauses 25.2 and 25.3.

25.5

Subject always to the provisions of clause 25.6, the Option Notice will not be binding on the Landlord unless it: (A) recites the agreement of the Landlord and the Tenant that sections 24 to 28 (inclusive) of the 1954 Act will not apply to

the tenancy to be created by the Further Lease; and

(B) recites the agreement of the Landlord and the Tenant's Guarantor that sections 24 to 28 (inclusive) of the 1954 Act will not apply to the tenancy to be entered into by the Tenant's Guarantor pursuant to paragraph 4.2 of Schedule 5 of the Further Lease; and

- (C) contains confirmation by the Tenant that before the date of the Option Notice:
 - the Landlord served a notice (the "New 1954 Act Notice") on the Tenant in accordance with section 38A(3)(a) of the 1954 Act;
 - the Tenant (or a person duly authorised by the Tenant) made a declaration or statutory declaration (the "New 1954 Act Declaration") confirming receipt of the New 1954 Act Notice in accordance with schedule 2 to the Order; and
 - 3. that where the New 1954 Act Declaration was made by a person other than the Tenant, that person was duly authorised by the Tenant to make the New 1954 Act Declaration on the Tenant's behalf.
- (D) contains confirmation by the Tenant's Guarantor that before the date of the Option Notice:
 - 1. the Landlord served a notice (the "**New Guarantor 1954 Act Notice**") on the Tenant's Guarantor in accordance with section 38A(3)(a) of the 1954 Act;
 - 2. the Tenant's Guarantor (or a person duly authorised by the Tenant's Guarantor) made a declaration or statutory declaration (the "New Guarantor 1954 Act Declaration") confirming receipt of the New 1954 Act Notice in accordance with schedule 2 to the Order; and
 - 3. that where the New Guarantor 1954 Act Declaration was made by a person other than the Tenant's Guarantor, that person was duly authorised by the Tenant's Guarantor to make the New Guarantor 1954 Act Declaration on the Tenant's Guarantor's behalf.

25.6

PROVIDED:

- (A) there are no outstanding substantial breaches of the Tenant's Covenants of which the Landlord has given notice to the Tenant and given the Tenant a reasonable opportunity to remedy that breach; and
- (B) the Tenant has paid all the Rents due under this Lease up to and including the Expiry of the Term; and
- (C) the Option Notice has been validly served in accordance with the provisions of clause 25.3 and complies in all respects with the provisions of clause 25.5; and
- (D) any surety of the Tenant's obligations under this Lease joins in the Further Lease to covenant with the Landlord in the terms contained in schedule 5 (mutatis mutandis)

the Landlord will grant and the Tenant will take the Further Lease of the Premises on or (if earlier) not more than one month before the Expiry of the Term PROVIDED FURTHER that where either the Tenant and/or any surety of the Tenant's obligations under the Further Lease are incorporated under the laws of any jurisdiction other than England and Wales they have also in each case provided immediately prior to such grant a legal opinion (addressed to and in a form reasonably acceptable to the Landlord on the advice of the Landlord's solicitors and dated not more than 14 days prior to such grant) prepared by legal advisers duly qualified to practice in the relevant jurisdiction confirming (i) the validity of the execution of the Further Lease by the Tenant and/or any such surety and (ii) the enforceability of the Further Lease under the laws of the relevant jurisdiction.

25.7

At any time after the Expiry of the Term a party who is ready able and willing to complete the grant of the Further Lease may serve on the other a notice to complete the grant of the Further Lease in accordance with this clause 25 ("Completion Notice") and in connection with the service of any Completion Notice:

- (A) a party is ready able and willing to complete if it could be but for the default of the other party;
- (B) the parties are to complete the grant of the Further Lease within fourteen days of serving a Completion Notice (excluding the date on which the Completion Notice is served) as to which time shall be of the essence; and
- (C) if either party receives but fails to complete the grant of the Lease in accordance with a Completion Notice that has been validly served under this clause 25 this option will be terminated but without prejudice to any other right or remedy of either party against the other.

25.8

The Tenant will register this Option at HM Land Registry within three months from the date of this Lease.

26. Contractual rights of third parties

26.1

No person who is not a party to this Lease shall have any right under the Contracts (Rights of Third Parties) Act 1999 to enforce any term of this Lease.

27. Third party disputes

27.1

Notification by the Tenant

The Tenant shall promptly give notice to the Landlord of any dispute arising between it and any other tenants or occupiers of the remainder of the Building or the owners or occupiers of any Adjoining Premises which relates to any right or privilege, any party wall or to the use and enjoyment of the Common Parts.

27.2 Determination by the Landlord

Unless the Landlord is a party to the dispute, the resolution of any dispute notified under clause 27.1 shall, at the request of the Landlord, be carried out on behalf of the Tenant by the Landlord, but at the Tenant's cost and expense.

28. Assignment of reversion

28.1

The Tenant agrees not unreasonably to withhold or delay its consent to any release requested by the Landlord under s.6 or s.7 of the 1995 Act.

29. Notices

29.1

Any notice served under or in respect of this Lease may be served by posting it in a prepaid envelope and shall be deemed to have been served on the first working day after which it was posted and evidence showing that the envelope containing the notice was properly addressed, stamped and posted shall be sufficient proof of service. Notices shall be served:

(A) on the Landlord, the Management Company or any guarantor of the Tenant at its registered office or last known address; and

(B) on the Tenant at its registered office or at the Premises.

30. Law and jurisdiction

30.1 English law

This Lease shall be governed by and construed in accordance with English law.

30.2 Jurisdiction

The parties to this Lease:

(A) irrevocably agree that, subject to clause 30.2(B), the English courts shall have exclusive jurisdiction in relation to any legal action or proceedings arising out of or in connection with this Lease ("Proceedings") and waive any objection to Proceedings in such courts on the grounds of venue or on the grounds that Proceedings have been brought in an inappropriate forum; and

(B) agree that clause 30.2(A) operates for the benefit of the Landlord and accordingly the Landlord shall be entitled to take Proceedings in any other court or courts having jurisdiction.

30.3

Any Proceedings shall be validly served on the Landlord if sent to c/o Hondo Enterprises, 9 Newburgh Street, London W1F 7RL or to any other address for service in England or Wales as it may notify in writing to the other parties at any time.

30.4

Any Proceedings shall be validly served on the Tenant's Guarantor if sent to Oxford House, Bell Business Park, Aylesbury, Bucks HP19 8JR or to any other address for service in England or Wales as the Tenant's Guarantor may notify in writing to the other parties at any time.

31. Execution and delivery

31.1

The parties have executed this Lease as a deed but have not delivered it until the date of this Lease.

SCHEDULE 1 : THE PREMISES PART 1 : DEFINITION OF THE PREMISES

1. Identification of the Premises

The Premises shown edged red on Plan B comprising the Third Floor of the Building.

2. Areas included in the Premises

2.1

The Premises include:

- (A) all non-structural or non-loadbearing walls and columns within the Premises together with the inner half, severed vertically, of those dividing the Premises from other parts of the Building; and
- (B) the internal plaster surfaces and finishes of the walls of the Premises, including those on any structural or loadbearing walls and columns within or enclosing the Premises; and
- (C) all light fittings, raised floor systems, floor screeds and finishes; and
- (D) all carpets or other floor coverings provided or paid for by the Landlord; and
- (E) all internal windows, internal window frames and internal window furniture and glass in the internal windows; and
- (F) all curtains and blinds provided or paid for by the Landlord; and
- (G) all doors and doorframes in all walls within or enclosing the Premises and all door furniture and glass within the doors; and
- (H) all sanitary and hot and cold water apparatus and equipment, if any, in and exclusively serving the Premises; and
- (I) all other landlord's fixtures; and
- (J) all Alterations except to the extent that they comprise tenant's fixtures.

3. Areas excluded from the Premises

3.1

The Premises exclude:

- (A) the Main Structure and the Office Commons Parts:
- (B) all Conduits within or leading to the Premises (whether or not these exclusively serve the Premises); and
- (C) the Services Systems within the Premises; and
- (D) all tenant's fixtures and chattels; and
- (E) the external windows, external window frames and external window glass including those in any external cladding or overlooking any atrium or galleria.

49

PART 2: THE SERVICES SYSTEMS

The Services Systems include, without limitation:

- 1. All heating, cooling, air conditioning and ventilation units.
- 2. The Heat Installations.
- 3. Boilers and storage tanks.
- 4. Sprinkler systems, fire alarm systems and fire prevention equipment.
- 5. Intruder alarms and other security systems.
- 6. Lifts.
- 7. Emergency generator.
- 8. Uninterrupted power supply systems.
- 9. Window cleaning cradles and associated equipment.
- 10. Telephones and communication equipment.
- 11. All control systems for any of the Services Systems.

50

SCHEDULE 2: RIGHTS GRANTED

1.

Conduits

1 1

The right to the passage of Utilities through any Conduits which serve the Premises which do not form part of the public mains.

2. Rights of access

2.1

The right:

- (A) of access on foot only over the Office Common Parts for the purposes of access to and from the Premises; and
- (B) to use the passenger lifts in the Office Commons Parts for access to and from the Premises; and
- (C) to use any delivery area, loading bay and goods lifts which the Landlord has designated for use by the Tenant during the Business Hours only for the purposes only of loading and unloading and delivering goods and equipment to and from the Premises; and
- (D) to use the refuse disposal and collection areas within the Office Common Parts which the Landlord or the Management Company shall designate for use by the Tenant for the purpose of depositing refuse ready for collection; and
- (E) the right in emergency only to use any fire escape routes within the Building as the Landlord may designate; and
- (F) (at reasonable times and after giving reasonable written notice) to enter upon such other parts of the Building as may reasonably be necessary in order to make connections to any Conduits within or leading to the Premises and to inspect and repair such connections or otherwise complying with the Tenant's obligations under this Lease the Tenant doing as little damage as possible and making good all physical damage caused to the Building to the reasonable satisfaction of the Landlord and complying with the reasonable requirements of the Landlord and causing the minimum of inconvenience to the occupiers of such other parts of the Building.

3. Support

3.1

The right of support, shelter and protection for the Premises from the remainder of the Building.

4. Signage

4.1 In the entrance hall

The right to have the name of the Tenant and the location of the Premises displayed in such manner and in such place as the Landlord may reasonably designate in the main ground floor entrance hall of the Office Common Parts.

4.2 On the entrance doors

The exclusive right to display on the entrance doors of the Premises a sign or signs stating the Tenant's name and business of such nature and size and in such position as the Landlord may reasonably determine.

5. Basement Facilities

5.1

The non-exclusive right to use the cycle spaces and the shower and changing facilities provided by the Landlord currently in the basement of the Building subject to the right of the Landlord to provide alternative similar facilities elsewhere in the Building provided that such facilities shall be available for use at all times but operated on a first come first served basis.

SCHEDULE 3: RIGHTS RESERVED

1. Conduits and Services

1.1 To use the Conduits

The right to the passage of Utilities through any Conduits which may at any time be in, under or over the Premises for the benefit of the remainder of the Building and any Adjoining Premises.

1.2 To create new Conduits

The right to create any new services or easements in, under or over the Premises for the benefit of the remainder of the Building and any Adjoining Premises and to connect into and use any existing Conduits in, under or over the Premises.

2. Building and Adjoining Premises

2.1 Rights to use Building and Adjoining Premises

The right to build, alter, clean, decorate, demolish, develop, rebuild, renew, repair and carry out any other works upon and otherwise use or deal with the remainder of the Building and any Adjoining Premises in such manner and for any purpose as the person exercising the right thinks fit.

2.2 Exercise of the rights

In exercising the rights in paragraph 2.1 the person doing so may:

- (A) oversail cranes over the Premises for the duration of the works being carried out; and
- (B) build on or into any external wall of the Premises; and
- (C) erect scaffolding against any external wall of the Premises for the duration of the works being carried out

PROVIDED THAT the Landlord shall ensure that in the exercise of the rights referred to in paragraph 2.1 there is caused as little inconvenience or disturbance to the Tenant or any undertenant or other lawful occupier of the Premises as is reasonably practicable in the circumstances and that such exercise shall not prohibit or unreasonably interfere with or prevent the use of the Premises for the Authorised Use.

2.3No claims

The Tenant shall not be entitled to claim against the Landlord or any other person for any obstruction to the passage of light and air to the Premises or for any nuisance arising from the exercise of the rights in this paragraph 2.

3. Entry on to the Premises

3.1 Rights reserved

Subject to paragraph 3.2 below, the right to enter onto and remain upon the Premises with or without plant, machinery and equipment:

- (A) to exercise the rights reserved in this schedule 3; and
- (B) to carry out any works in respect of any easements or services benefiting the Premises, the remainder of the Building or any Adjoining Premises; and
- (C) to view the state and condition of the Premises or the remainder of the Building including, where necessary, opening up any part of the floors, ceilings and walls of the Premises; and
- (D) to remedy any breach of the Tenant's Covenants; and
- (E) to carry out any repairs, decoration, cleaning or other works which should or may be carried out by the Landlord whether under this Lease or otherwise or in connection with the provision of the Services; and
- (F) to carry out any repairs, decoration, cleaning or other works to the remainder of the Building and any Adjoining Premises; and
- (G) in connection with any requirements of the insurers; and
- (H) to value the Building whether for insurance purposes or otherwise; and
- (I) to prepare for the disposal of the Landlord's interest in the Premises or the Building or, in the last six months of the Term, the re-letting of the Premises; and
- (J) to affix on the exterior of the Premises notices for the sale of the Building or, in the last six months of the Term, the reletting of the Premises so long as this does not create a material obstruction of the access of light and air to the Premises.

3.2 Exercise of the rights

On the exercise of any rights of entry on to the Premises, the person entering shall give reasonable prior notice to the Tenant unless the rights need to be exercised in an emergency, cause as little damage as reasonably possible and as little inconvenience as reasonably practicable in the exercise of the rights and make good any physical damage caused to the Premises and the Tenant's fixtures and fittings in the exercise of those rights.

4. Support

4.1

The right of support, shelter and protection from the Premises for the remainder of the Building and any Adjoining Premises.

5. Fire escapes

5.1

The right in common with the Tenant in cases of emergency only to use any fire escape routes within the Premises designated by the Landlord for use as a means of escape in case of fire.

6. Superior Leases

6.1

The rights excepted and reserved to the Superior Landlords pursuant to the Superior Leases to the extent they relate to the Premises.

SCHEDULE 4 : TITLE MATTERS 1. The entries in the property and charges registers of the title numbers set out in clause LR2.1 as existing at the date of this Lease.
56

SCHEDULE 5: GUARANTEE PROVISIONS

1. Defined terms

1.1

In this schedule 5, unless the contrary intention appears:

"Event of Default" means:

- (A) the disclaimer of this Lease by a trustee in bankruptcy or liquidator of the Tenant or by the Crown; or
- (B) if the Tenant is a company, the Tenant is dissolved, struck off the register of companies or otherwise ceases to exist; or
- (C) the ending of this Lease pursuant to clause 2.2.

"Guarantee"

means any guarantee and indemnity given to the Landlord in accordance with the terms of this Lease by:

- (A) the Tenant's Guarantor under clause 24.1; or
- (B) the Tenant in accordance with clause 12.3(B)(1); or
- (C) any guarantor or guarantors of an assignee of this Lease in accordance with clause 12.3(B)(2); or
- (D) any guarantor of the Tenant in accordance with clause 12.3(D); or
- (E) any quarantor or quarantors of an undertenant in accordance with clause 13.6; or
- (F) any undertenant on an assignment of their underlease in accordance with paragraph 2.2(B) of Part 2 of schedule 6; or
- (G) any other person in accordance with the terms of this Lease.

"Guarantor"

means the party giving the Guarantee.

"New Lease"

means a lease of the Premises:

- (A) for a term of years commencing on the date of the Event of Default and expiring on the date upon which the Term would have expired; and
- (B) reserving a yearly rent equal to the yearly rent reserved under this Lease at the date of the Event of Default; and
- (C) containing rent a review date which occurs on the Review Date; and
- (D) containing a rent review date on the first day of the term of the New Lease, but reviewing the yearly rent as at the date of the outstanding rent review, if there was an outstanding rent review under this Lease which had not been agreed or determined before the date of the Event of Default; and
- (E) containing redecoration dates which occur on the dates upon which the Premises were to be redecorated under this Lease; and

(F) otherwise containing the same terms and conditions as this Lease.

"Obligations"

means the obligations guaranteed by the Guarantor under its Guarantee.

"Principal"

means the party who has responsibility for complying with the Obligations but excludes the Guarantor.

2. Effect of the Guarantee

2.1 Guarantee and indemnity

The Guarantor covenants with the Landlord as primary obligor, and not merely as guarantor, that the Obligations will be complied with and also as a separate obligation to indemnify the Landlord against any breach of them.

2.2 Claims

The Guarantor acknowledges to the Landlord that the Landlord may make any claim against the Guarantor without first making demand of the Principal or exercising any other rights or enforcing any other security or guarantee which it may have in respect of the Obligations.

2.3 Continuing guarantee

The Guarantor acknowledges to the Landlord that the Guarantee is a continuing guarantee and shall not be released or varied by:

- (A) any Event of Default; or
- (B) the surrender of any part of the Premises; or
- (C) the variation of this Lease; or
- (D) any concession, time, indulgence or release given by the Landlord to the Principal or any co-guarantor; or
- (E) any other act or thing which would, but for this paragraph 2.3, release or vary the Guarantee.

3. Postponement of rights

3.1 Priority of claims

The Guarantor covenants with the Landlord that, unless and until all the Obligations have been complied with or otherwise discharged:

- (A) the Guarantor shall not claim any rights of subrogation against the Principal, prove or claim in competition to the Landlord in any liquidation, bankruptcy, arrangement, scheme of arrangement, composition with creditors, receivership or administration of or concerning the Principal, or take any guarantee, indemnity or other security or other right from the Principal in respect of all or any of the liabilities of the Guarantor under this Lease; and
- (B) any moneys which the Guarantor receives from any procedure or action of any of the kinds referred to in paragraph 3.1(A) shall be paid to the Landlord, and every guarantee, indemnity or other security or other right referred to in paragraph 3.1(A) shall be held on trust for the benefit of the Landlord.

3.2 Discharge conditional

The Guarantor acknowledges to the Landlord that, if any payment made by the Principal, Guarantor or any co-guarantor is ordered to be refunded under any law relating to bankruptcy, liquidation or insolvency, the Landlord may claim from the Guarantor as if such payment had not been made and any release, discharge or settlement between the Guarantor and the Landlord shall take effect subject to this condition.

3.3 Set-off, counterclaim and other deductions

The Guarantor covenants with the Landlord that the Guarantor shall make any payments due from it under the Guarantee in full, without any deduction or withholding by way of set-off, counterclaim, taxation or otherwise, except only those required by law, in which case the Guarantor shall pay such increased amount as is necessary to ensure that the Landlord receives, after all such deductions and withholdings, the full amount.

4. Event of Default

4.1 Landlord's rights

The Guarantor acknowledges to the Landlord that, at any time within the period of six months after the Landlord receives actual notice that an Event of Default has occurred, the Landlord may serve written notice on the Guarantor requiring the Guarantor to accept the grant of a New Lease as tenant. Where there is more than one Guarantor, the Landlord may serve the notice on any of them.

4.2 Guarantor's obligations

The Guarantor covenants with the Landlord that if the Landlord requires the Guarantor to accept the grant of a New Lease, the Guarantor shall:

- (A) execute and deliver to any superior landlord a counterpart of any licence to underlet which may be required for the grant of the New Lease. The licence to underlet shall contain such covenants as may be properly required of the Guarantor as undertenant; and
- (B) if required by the Landlord, do all things required for the purposes of validly excluding the provisions of ss24 to 28 (inclusive) of the 1954 Act (as amended by the Order) in relation to the New Lease; and
- (C) if required by the Landlord, join in any application and do all things reasonably required for the purposes of authorising an agreement to exclude the operation of ss24 to 28 Landlord and Tenant Act 1954 in relation to the New Lease; and
- (D) accept the grant of the New Lease and execute and deliver to the Landlord a counterpart of the New Lease; and
- (E) pay the costs and disbursements of the Landlord's solicitors and other professional advisers on a full indemnity basis arising from the grant of the New Lease and the enforcement of the Guarantee together with any irrecoverable Value Added Tax incurred by the Landlord.

4.3 Effect of the grant of a New Lease

The Guarantor acknowledges to the Landlord that the grant of a New Lease shall not prejudice any rights of the Landlord against the Guarantor or any co-guarantor in respect of any liability under the Guarantee or any other guarantee or security held by the Landlord in respect of the Obligations in so far as that liability relates to any period prior to the date of the Event of Default.

4.4 No New Lease

The Guarantor covenants with the Landlord that if an Event of Default occurs and for any reason the Landlord does not require the Guarantor to accept the grant of a New Lease under paragraph 4.1, the Guarantor shall:

- (A) pay to the Landlord on demand an amount equal to all sums which would have been payable under this Lease but for the Event of Default; and
- (B) as a separate indemnity, indemnify the Landlord against the failure of the Tenant to comply with the Tenant's Covenants; in each case for the period commencing on the date of the Event of Default and ending on the date six months after the Event of Default or, if earlier, the date upon which the Landlord re-lets the Premises.

SCHEDULE 6: PROVISIONS FOR UNDERLETTING PART 1: TERMS OF THE UNDERLEASE

1. General terms

1 1

Except as required in this Part 1 of schedule 6, the Tenant shall grant each underlease on the same terms, including the right of re-entry, and subject to the same covenants and conditions as this Lease.

2. Fine or premium

2.1

No underlease shall be granted for a fine or premium.

3. Rents

3.1 Yearly rent

The rent payable shall be the rent reasonably obtainable for the Underlet Premises in the open market.

3.2 Rent review

Each underlease to be granted for a term exceeding 5 years shall contain provisions for the upwards only review of rent on each 5th anniversary of the term commencement date of the relevant underlease on the same terms as clause 5 with such amendments as the Landlord may approve, such approval not to be unreasonably withheld or delayed where the proposed amendments reflect the market practice at the date of the underlease.

3.3 Service charge rent

Each underlease shall contain provisions requiring the undertenant to pay by way of additional rent the whole (in the case of an underletting of whole) or a fair and proper proportion (in the case of an underletting of a Permitted Part) of the Service Charge payable by the Tenant under this Lease.

4. Underlease term

4.1

The term of the underlease shall expire no later than three days before the date on which the Term expires.

5. Tenant's Covenants

5.1

Each underlease shall contain a covenant by the undertenant to comply with the Tenant's Covenants, other than the payment of rent and, in the case of an underletting of a Permitted Part, to the extent only that they apply to the undertenant's use and occupation of the Underlet Premises.

6. Assignment and charging

6.1 Dealings with part

Each underlease shall contain an absolute prohibition on the undertenant assigning or charging part only of the Underlet Premises.

6.2 Dealings with the whole

Each underlease shall contain a covenant that the undertenant shall not assign or charge the whole of the Underlet Premises without the prior written consent of the Landlord and the Tenant, such consent not to be unreasonably withheld or delayed.

6.3 Terms for assignment

Each underlease shall contain restrictions on its assignment in the form set out in clause 12.

7. Subletting

7.1 Subletting of the whole

Each underlease shall contain an absolute prohibition on the undertenant subletting the whole of the Underlet Premises unless the Underlet Premises comprise the whole of the Premises, when the undertenant shall be entitled to sublet the whole of them by a sublease which excludes the provisions of ss24 to 28 (inclusive) of the 1954 Act.

7.2 Subletting of part

Each underlease shall contain an absolute prohibition on the undertenant subletting part only of the Underlet Premises.

7.3 Terms of subletting

Where permitted under this paragraph 7, each underlease shall contain:

- (A) a covenant by the undertenant not to sublet the whole of the Underlet Premises without the prior written consent of the Landlord and the Tenant, such consent not to be unreasonably withheld or delayed; and
- (B) provisions prohibiting the creation of any sublease of the whole of the Underlet Premises except by a sublease which complies with Part 1 of this schedule 6 but so that no subtenant of that undertenant can create any further subletting of the whole or any part of the premises sublet to them; and
- (C) a covenant by the undertenant to procure that any permitted subtenant gives a covenant to the Landlord before the grant of any sublease on similar terms to the covenants set out in Part 2 of this schedule 6; and
- (D) an absolute covenant by the subtenant not to assign part only of the Underlet Premises and a further covenant not to assign the whole of the Underlet Premises without the prior written consent of the Landlord, such consent not to be unreasonably withheld or delayed.

8. Sharing of occupation

8.1

Each underlease may contain provisions entitling the undertenant to share the occupation of the Underlet Premises with a Group Company of the undertenant on the same terms as those set out in clause 14.2.

PART 2: UNDERTENANT'S COVENANTS

1. Defined terms

1.1

In Part 2 of this schedule 6, unless the contrary intention appears:

"Liability Period"

means the period from and including the date the underlease is granted or assigned, as the case may be, until the date on which the undertenant or assignee, as the case may be, is released from its obligations to observe the Undertenant's Covenants under the 1995 Act.

"Undertenant's Covenants"

mean the obligations to be complied with by the undertenant of any underlease.

2. To observe terms of this Lease, etc

2.1 Undertenant's Covenants

The undertenant shall comply with the Undertenant's Covenants and the Tenant's Covenants (except the covenant to pay rent and, in the case of an underlease of part, to the extent only that they apply to the Underlet Premises) during the Liability Period and indemnify the Landlord on a full indemnity basis against any breach of them occurring during that period.

2.2 Assignment

On any assignment of the underlease, the undertenant shall:

(A) procure that the assignee gives the Landlord a direct covenant on the terms of this paragraph 2; and

(B) give the Landlord an Authorised Guarantee Agreement guaranteeing that its assignee will comply with the Undertenant's Covenants and the Tenant's Covenants (except the covenant to pay rent and, in the case of an underlease of a Permitted Part, to the extent only that they apply to the Underlet Premises) from the date of the assignment until the date when the assignee is released from its obligations under the 1995 Act.

SCHEDULE 7 : SERVICES AND ITEMS OF EXPENDITURE PART 1 : THE BUILDING SERVICES

- Inspecting, repairing, maintaining, decorating and, where appropriate, treating, cleaning, washing down, servicing
 and, when in the reasonable opinion of the Landlord necessary, altering, renewing, replacing and rebuilding:
 (A) the Retained Parts; and
 - (B) all other plant, machinery and equipment and any Conduits which are not adopted as part of the public mains, except those which are the individual responsibility of the Tenant or any other tenant or occupier of the Building.
- 2. Providing lighting to the Retained Parts where appropriate at such times as the Landlord reasonably considers necessary or desirable.
- 3. Providing an emergency power supply to the Building as the Landlord may from time to time determine by means of the emergency generator forming part of the Services Systems.
- 4. Providing a lift service in the Retained Parts at such times as the Landlord reasonably considers necessary or desirable
- 5. Providing a supply of hot and cold water to the lavatories in the Retained Parts at such times as the Landlord considers necessary or desirable.
- 6. Providing Heat and air conditioning to the Retained Parts at such times as the Landlord considers necessary or desirable using and within the capacity of the existing Service Systems.
- 7. Providing and installing in the main entrance to the Building and in such other places as the Landlord deems appropriate name-boards of such design, size and character as the Landlord shall determine.
- 8. Providing security guards for the Retained Parts at such times as the Landlord may reasonably determine.
- 9. Providing security systems for the Building including, without limitation, alarm systems, internal telephone systems, closed circuit television systems, and any traffic barriers, car park and traffic control and security systems.
- 10. Procuring the disposal of refuse from the Building including its collection and compaction and the provision of receptacles and plant and equipment for these purposes.
- 11. Cleaning the exterior of all windows and window frames in the Building, save where the responsibility of any particular tenant, and cleaning other external part of the Building and providing and maintaining facilities and equipment for these purposes.
- 12. Implementing rodent or other pest control in the Building.
- 13. Installing inspecting, maintaining, operating, repairing, and, as and when in the reasonable opinion of the Landlord necessary, renewing and replacing fire detectors, alarms, fire prevention and fighting equipment and systems in the Building.

PART 2: THE OFFICE SERVICES

- Inspecting, repairing, maintaining, decorating and, where appropriate, treating, cleaning, washing down, servicing
 and, when in the reasonable opinion of the Landlord necessary, altering, renewing, replacing and rebuilding:

 (A) the Office Common Parts: and
 - (B) such of the Services Systems as are intended to serve the office parts of the Building; and
 - (C) all other plant, machinery and equipment and any Conduits which are not adopted as part of the public mains, except those which are the individual responsibility of the Tenant or any other tenant or occupier of the Building.
- 2. Providing lighting to the Office Common Parts during Business Hours.
- 3. Providing a lift service in the Office Common Parts during Business Hours.
- 4. Providing a supply of hot and cold water to the lavatories in the Office Common Parts during Business Hours.
- 5. Providing Heat and air conditioning to the Premises and the Office Common Parts at such times as the Landlord considers necessary or desirable using and within the capacity of the existing Service Systems.
- 6. Providing and installing in the main entrance to the office part of the Building and in such other places as the Landlord deems appropriate name-boards of such design, size and character as the Landlord shall determine.
- 7. Providing security guards for the Office Common Parts 24 hours a day seven days a week.
- 8. Providing security systems for the Office Common Parts including, without limitation, alarm systems, internal telephone systems, closed circuit television systems, and any traffic barriers, car park and traffic control and security systems.
- 9. Procuring the disposal of refuse from the Office Common Parts including its collection and compaction and the provision of receptacles and plant and equipment for these purposes.
- 10. Cleaning the exterior canopy of the Building and providing and maintaining facilities and equipment for these purposes.
- 11. Installing inspecting, maintaining, operating, repairing, and, as and when in the reasonable opinion of the Landlord necessary, renewing and replacing fire detectors, alarms, fire prevention and fighting equipment and systems in the Office Common Parts.
- 12. Providing and maintaining any furniture architectural or ornamental features or murals and any horticultural displays plants and flowers in the Office Common Parts.
- 13. Providing one or more commissionaires or receptionists in the ground floor entrance hall of the office part of the Building
- 14. Maintaining a house internal telephone system within the Office Common Parts.

- 15. Maintaining a house satellite system within the Office Common Parts.
- 16. Providing musical and other entertainment in the ground floor entrance hall of the office part of the Building.
- 17. Any other services relating to the office parts of the Building or any part of them which the Management Company from time to time reasonably considers appropriate and which shall be in keeping with the principles of good estate management.

PART 3: THE CHP SERVICES

1. Providing Heat to the Premises and maintaining, insuring, repairing and renewing (as necessary) and keeping in good working order the Heat Installations provided that there shall be no obligation to provide the Heat to the Premises if the Heat Charge or any part of it remains unpaid seven days from demand.

PART 4: ITEMS OF EXPENDITURE

- 1. The payment of all existing and future rates (including water rates), taxes, duties, charges, assessments, impositions and outgoings whatsoever payable in respect of all or any part of the Retained Parts.
- 2. All payments whether direct or indirect to any local or other competent authority, person or body towards or in connection with the carrying out of all or any of the Services.
- 3. The cost of compliance with any Statute or any directions or requirements or recommendations of any competent authority or of any insurers of the Building.
- 4. The cost of abating any nuisance to the Building and the cost of removing any obstruction of the Building.
- 5. The cost of taking any steps which the Landlord reasonably deems desirable or appropriate in making representations against or otherwise contesting the operation of the provisions of any Statute or any directions or requirements of a competent authority relating to the Retained Parts.
- 6. The cost of maintenance and inspection contracts for the repair and maintenance of the Services Systems and any other plant, equipment and machinery in the Building and contracts for landscaping, planting and cultivation.
- 7. The cost of inspecting, repairing, maintaining, renewing, replacing and, where appropriate, lighting, marking, decorating and cleaning roads, paths, yards, party walls, fences or other structures or any other easements or services or other areas used or available to be used in common by all or any of the occupiers of the Building and the occupiers of any Adjoining Premises.
- 8. The cost of any policy or policies of insurance for insuring the Services Systems and any other plant, equipment and machinery in the Building against sudden and unforeseen damage and breakdown.

- 9. The cost of any policy or policies of insurance for insuring against occupier's and third party liability insofar as the same relate to the delivery of the Services.
- 10. The cost of any policy or policies of insurance for insuring loss of Service Charge following an Insurance Event.
- 11. The costs of preparing, submitting and settling any insurance claim relating to the Building not charged under clause 7.4(D).
- 12. The cost of carpeting, re-carpeting or otherwise covering such parts of the Office Common Parts as the Landlord shall from time to time reasonably consider desirable or appropriate.
- 13. The cost of providing in the main entrance hall and lift lobbies of the Building desks, tables, chairs and other furniture and fittings.
- 14. The cost of providing:
 - (A) floodlighting for the Building; and
 - (B) floral and other special or seasonal decorations for the office part of the Building; and
 - (C) any plants, shrubs, trees or garden or grassed areas in the Building.
- 15. The cost of the fuel, oil, gas, electricity or other energy supplies not included in the Heat Charge or the Energy Charge and needed in running or operating any of the Services or the Services Systems, except where provided for the exclusive use of a particular tenant or tenants.
- 16. The cost of annual servicing and repair and maintenance charges in relation to the Heat Installations together with an element of cost to be kept in a specific reserve fund for the future renewal and replacement of the Heat Installations.
- 17. The cost of small power to the Premises and the other Lettable Areas.
- 18. Such proportion of the CRC Net Costs as is fairly and reasonably attributable to the Building.
- 19. The cost of, where appropriate, toilet requisites and hygiene services in the lavatories in the Office Common Parts including the supply, maintenance, repair and renewal of receptacles, plant and equipment for these purposes.
- 20. The cost of providing tenants and occupiers of the office parts of the Building with any of a help desk, call centre and/or information centre service.

- 21. The cost of employing such staff and other personnel as the Landlord shall reasonably consider necessary or desirable for the carrying out of the Services or in connection with any of the other Items of Expenditure set out in this schedule 7, including:
 - (A) the payment of fees, expenses, salaries, gratuities, bonuses, annuities, redundancy payments, pensions, pension contributions, social security and national insurance contributions and other statutory levies or emoluments in respect of such staff or personnel; and
 - (B) the cost of providing uniforms, protective clothing, tools, appliances, materials and equipment for the proper performance of the duties of such staff or personnel; and
 - (C) the cost of providing for such staff or personnel residential or other accommodation, including a management office, such cost to include (without limitation) all outgoings and all utility charges and the cost of repairs, renewals and redecoration in respect of such accommodation and the actual or a fair and reasonable notional rent for such accommodation.
- 22. The fees, charges, expenses and commissions payable (on a full indemnity basis) to any solicitor, auditor, accountant, surveyor, valuer, architect, engineer, managing agent or other professional whom the Landlord and/or the Management Company may from time to time employ in connection with the maintenance or management of the Building including the cost of causing to be prepared and audited the Service Charge Statement.
- 23. The Landlord's and/or the Management Company's own management and administration expenses and, where management of the Building is undertaken by the Landlord, the Management Company or any Group Company of the Landlord or the Management Company, a sum equal to ten per centum of Service Charge Expenditure in the Accounting Period by way of a management fee.
- 24. The costs, on a full indemnity basis, of enforcing, or attempting to enforce:
 - (A) any term, covenant or condition or exercising, or attempting to exercise, any right of re-entry contained in any lease, underlease, licence or agreement relating to all or any part of the Building unless for the non-payment of yearly rent; and
 - (B) any warranty or other obligation owed to the Landlord, whether in contract or in tort, relating to the design or construction of the Building less any sums actually recovered by the Landlord from any third party on account of such costs.
- 25. Interest, commission, banking charges and fees in respect of any moneys included in Service Charge Expenditure borrowed to finance the provision of any of the Services or the Items of Expenditure.
- 26. The cost of providing any of the Services outside the Business Hours where the Landlord reasonably considers it appropriate to do so having regard to the level of occupation of the Building outside the Business Hours.
- 27. The costs incurred by the Landlord in complying with the tenant covenants under the Superior Leases.
- 28. The cost of any other works, facilities or services of any kind whatsoever which the Landlord reasonably considers necessary or desirable for the benefit of the Building or any part of it or the tenants or occupiers of the Building or in the interests of good estate management.

SCHEDULE 8: PROVISIONS REFERRED TO IN CLAUSE 25

The Further Lease entered into pursuant to clause 25 will be on the same terms and conditions as this Lease except that:

- 1. It will not contain provisions equivalent to clause 2.4, clause 25 and this schedule 8
- 2. The Term to be described at clause LR6 will be "5 years from and including the Term Commencement Date to and including 21 June 2031"
- 3. The entry at clause LR9.1 shall be deleted and replaced with "None".
- 4. The definition of "Break Date" shall be deleted from the Particulars.
- 5. The definition of "**Principal Rent**" in the Particulars will be the Principal Rent payable under this Lease as at the expiry of the Term (or if payment has been suspended or restricted the Principal Rent which would have been payable had there been no suspension or restriction) subject to review under clause 5.
- 6. The definition of the "Rent Commencement Date" and the "Review Date" in the Particulars will be the Term Commencement Date.
- 7. The definition of the "Term Commencement Date" in the Particulars will be 22 June 2026.
- 8. The following new definition will be inserted into clause 1.1:

"Previous Lease"

- : The previous lease of the Premises dated 22 June 2016 between (1) AG Commercial St. I.B.V. (2) DXI Limited (3) 8x8 Inc. and (4) One Commercial Street Management Company Limited.
- 9. Paragraph (E) in the definition of "Notional Lease" in clause 5.1 shall be deleted and replaced with the following: "(E) for a term equal to five years commencing on the Review Date; and"
- 10. Paragraph (G) in the definition of "Notional Lease" in clause 5.1 shall be deleted and replaced with the following: "(G) otherwise on the terms of this Lease, as at the Review Date, but excluding this clause 5, the amount of the Principal Rent payable and any rent free period or period of reduced or concessionary rent, capital payment or any other inducement of whatever nature allowed or given to the Tenant at the commencement of or otherwise in relation to this Lease."
- 11. Clause 20.2(D) shall be deleted and replaced with the following:
 - "(D) reinstate all Alterations made to the Premises and restore the Premises to the state and condition in which the Tenant under the Previous Lease first took possession of them whether such works were carried out during or prior to the Term during the term of years granted by the Previous Lease by the Tenant, any undertenant or their respective predecessors in title (whether of this Lease or the Previous Lease)"

APPENDIX 1 : BASE BUILD SPECIFICATION RELAY BUILDING OUTLINE CATEGORY A SPECIFICATION

This specification has been prepared for the Landlord to outline the basebuild specification to the CAT A Office Fitout. The specification is intended to be a description for the space for the purposes of a Lease. It is not intended to act as a detailed design or specification covering all aspects of the refurbishment and should be used as a general description.

The occupation density of the floor plates is 1 person per 8m² (175 people) which reflects the Shell & Core Outline Specification.

1 FLOORING

 Part raised access floor system on 600 x 600mm accessible tiles left unfinished complete as part of the original basebuild works.

2 CEILINGS

Exposed existing concrete ceiling soffit sealed with Keim protective sealant.

3 LIGHTING

The office lighting is complete with a completely programmable and addressable lighting control system c/w presence detection and daylight dimming control.

- All Office luminaires utilise high frequency ballasts and low energy fluorescent and compact fluorescent or LED lamp sources in compliance with Part L of the Building Regulations.
- The lighting installation is in align with CIBSE LG3 and LG7 code of practice taking cognisance of areas with computer use.
- Emergency lighting provision will be via self-contained battery backup luminaires with 3hrs battery backup. The emergency lighting installation will provide emergency operation for 3 hours on mains or local circuit failure. The system shall be designed to BS5266 and EN1838.
- Emergency test facilities are located within the floor distribution cupboard.

4 WALL FINISHES

Acoustic foam to the core walls around the office floor plates.

- RIMEX metal mirror effect panelling with back painted glass tiles to floor 5 of the office floor plate around the perimeter of the building. This was not installed to floors 1 4
- Plasterboard and painted finish to all other areas around the office floor plate.
- Black painted skirting and architraves

70

5 DOORS

- New doors were installed to the 5th floor. All doors leading to the lobbies and circulation space have been painted with good quality paint.
- Metal `corten' surrounds each of the doors entering the office floor plates.

6 SPRINKLER/FIRE PROTECTION

Therefore the staircase sizes are 1335mm per stair for ST 2 and ST 3 and therefore capable of catering for a maximum occupancy of 175 persons per floor on the office levels.

- The is a Fire Service Access route between ST2 and ST3 at level 05 and level 06
- Fire service access to the office levels are generally via ST2 and the fire fighting lift associated with this core.
- Both ST2 and ST3 are provided with wet riser fire mains served by tanks and pumps in the basement.
- A fire management room is provided on ground level of the office entrance.
- Smoke ventilation is provided to office stair lobbies
- ST2 is provided with natural ventilation via a 3msq cross-sectional area shaft. AOV's are provided from the shaft at each level to the lobby.
- The ventilation system is linked with the fire detection and alarm system within the lobbies. Upon activation of the lobby fire
 alarm and detection system, the damper to the floor level affected opens and the damper to the top of the natural ventilation
 shaft opens to allow smoke ventilation from the lobby space. Dampers to all other floors are expected to close. It is
 understood that the ventilation system is commissioned to the completed floors.
- The mechanical ventilation system to the ST3 lobby is provided with duty and standby smoke extract fans which are located on the 7th floor roof space.
- The mechanical extract shaft is located within the riser (Riser 32) which is next to the stair. A duct connects the extract riser to the ST3 lobby on each level. This extract system along with the duct to the lobby at each level is fire rated ductworks. Fire dampers are fitted to the system at each floor level.
- · A system of sprinkler protection is installed throughout all areas.
- Sprinkler heads are provided to achieve a design density of Ordinary Hazard Group II, in accordance with the requirements of British Standard 5306 Part 2 and the Landlord's insurance requirements.
- The base building fire alarm system provides coverage throughout all areas and shall comply with the requirements of British Standard 5839 Parts 1 LI standard.
- The base building public address system, which forms part of the fire alarm installation, provides the required audibility levels of 65dB(A) (or +5dB(A) above ambient noise levels) in all areas.

7 MECHANICAL

(A) Ventilation

- Fresh air is supplied evenly around the office floor plate. Fresh air is delivered to the back of the fan coil units, with the termination stopping 150mm short of the fan coil intake. Volume control dampers are provided to facilitate the full commissioning of the primary air system.
- Extract is provided by bell mouths terminating adjacent to each of the two ventilation risers. This doubles up as smoke clearance ventilation for the office floor plates
- Fresh air is provided at a rate of 10l/s/person for the designed occupancy with an occupation utilization of 93%.

(B) **Heating & Cooling**

- Heating and cooling is provided by 4-pipe fan coils to the guide lines with in the British Council for Offices criteria.
 The fan coils are exposed and where possible they shall be selected to meet the design duty when running at low speed.
- Perimeter FCUs are c/w heating and cooling coils. Internal FCU's are c/w cooling coils only.
- Each FCU is provided with local controller, isolation valves, flushing valves, flushing bypass, strainers, regulation/commissioning valves and 3-port control valves linked to the BMS.
- 2no UPVC condensate drainage mains complete with branches from each FCU are taken to the central drainage.
- Energy meters are provided for each CHW and LTHW service. These are linked to the Landlords central billing system.

8 ELECTRICAL

• No below floor power distribution is provided. There is sufficient power available from 2 statutory meters, one located in the north riser and the other in the south riser. Each supply is 102kVA.

9 STATUTORY SIGNS/TENANT SIGNS

• Internal signs shall comply with statutory requirements and the landlords base build specification signage.

10 WINDOW COVERING

N/A

11 PASSENGER LIFT LOBBY

- · Walls finished with mirror panelling with stainless steel trim
- · Floor finished with stone tiling
- · Recessed linear lighting with PIR control
- · Access Control to office doors

12 DESIGN CRITERA

The design criteria for the base build mechanical services installations are as follows:-

a. External Design Conditions

Summer 28°Cdb. 22°wb

Winter -4°C db, 100% saturated

Heat Rejection 34°C db, 21°C wb

b. Internal Comfort Conditions

Offices Fan coil solution

Summer 24°C db + 2°C

Winter 21°C db + 2°C

c. Fresh Air Quantities

10l/s/person for offices

d. Supply Air Quantity

Offices 10l/s/person for offices

e. Smoke Exhaust

Office 3 air changes on fire initiation with override control by the Fire Brigade to 6 air changes per hour mechanical exhaust. System sized to exhaust the single largest floor.

f. Load Densities for Cooling

In addition to the solar and transmission loads the air conditioning equipment will be designed to meet the following internal loads.

Offices: Lighting 12W/m2

g. Small Power

Offices at risers 25 W/m2

h. Noise and Vibration Control

The base-building will be designed to achieve the following limiting noise levels, from engineering services installations, when the office fit-out is installed.

Area Noise level

Open plan offices NR38

All noise levels quoted above are applicable when the base building plant is running under normal operation and greater than 1m from an enclosing surface (floor, wall or ceiling).

The above figures do not include noise breakout from standby generators and other emergency plant.

The design criteria for the electrical services installations are as follows:-

Load Densities:

Lighting 12 watts/m2

Sockets and Small Power 25 watts/m2

Lighting Levels:

Room or Area	Illumination level (Lux)	Uniformity			
General Circulation Corridors	200	0.8			
Stairs	150				
WC's	150				
Offices	350-400	0.8 at task plane/workstations			
General Stores	100				
Risers	200				
Reception , Break out areas	200-500	0.8			
Reception Desk	300	0.8			
Emergency Lighting	BS5266	BS5266			

13 FIXTURES AND FITTINGS (5th floor only)

- New Belfast sinks
- New wall mounted 3 hole basin trap
- · New soap dispensers, toilet brush and double toilet roll holders
- · Back to wall WC, Ideal Standard
- · Back to wall Urinal, Duravit
- Compact flushing devise
- · New mirrors over each hand basin with pre-made cut mirrors to create window affect

EXECUTED AS A DEED on behalf of)	
AG COMMERCIAL ST. I.B.V. a)	
company incorporated in the Netherlands)	
by Eurostrat Netherlands Manager, L.L.C.,)	
Jean-Baptiste Garcia and Robert Tieskens)	
being persons who, in accordance with)	
the laws of that territory, are acting)	
under the authority of the company)	
Name of Francistant Notice of the Management I. I. O. Discotor		
Name: Eurostrat Netherlands Manager, L.L.C., Director		
Name: Eurostrat Netherlands Manager, L.L.C., Director		
Name: Jean-Baptiste Garcia, Director		
Name: Jean-Baptiste Garcia, Director		
Name: Jean-Baptiste Garcia, Director		

75

EXECUTED as a deed by	
DXI LIMITED acting by	
, a director, in the	
presence of:	
Signature of witness: Name of witness:	
Address of witness:	
Occupation of witness:	
EXECUTED as a deed by	
8X8, INC. acting by	
, an officer, in the	
presence of:	
Signature of witness:	
Name of witness:	
Address of witness:	
Occupation of witness:	
EXECUTED as a deed by	
ONE COMMERCIAL STREET	
MANAGEMENT COMPANY	
LIMITED acting by	
Dimenton	
Director	
, a director, in the	
presence of: Signature of witness:	
Name of witness:	
Address of witness:	
Addices of Withess.	
Occupation of witness:	
76	

RULES 13a-14(a) AND 15d-14(a) UNDER THE SECURITIES EXCHANGE ACT OF 1934 AS ADOPTED PURSUANT TO

SECTION 302 OF THE SARBANES-OXLEY ACT OF 2002

- I, Vikram Verma, certify that:
- 1. I have reviewed this quarterly report on Form 10-Q of 8x8, Inc.;
- 2. Based on my knowledge, this report does not contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements made, in light of the circumstances under which such statements were made, not misleading with respect to the period covered by this report;
- 3. Based on my knowledge, the financial statements, and other financial information included in this report, fairly present in all material respects the financial condition, results of operations and cash flows of the registrant as of, and for, the periods presented in this report;
- 4. The registrant's other certifying officer and I are responsible for establishing and maintaining disclosure controls and procedures (as defined in Exchange Act Rules 13a-15(e) and 15d-15(e)) and internal control over financial reporting (as defined in Exchange Act Rules 13a-15(f) and 15d-15(f)) for the registrant and have:
 - a. Designed such disclosure controls and procedures, or caused such disclosure controls and procedures to be designed under our supervision, to ensure that material information relating to the registrant, including its consolidated subsidiaries, is made known to us by others within those entities, particularly during the period in which this report is being prepared;
 - b. Designed such internal control over financial reporting, or caused such internal control over financial reporting to be designed under our supervision, to provide reasonable assurance regarding the reliability of financial reporting and the preparation of financial statements for external purposes in accordance with generally accepted accounting principles;
 - Evaluated the effectiveness of the registrant's disclosure controls and procedures and presented in this report our conclusions
 about the effectiveness of the disclosure controls and procedures as of the end of the period covered by this report based on
 such evaluation; and
 - d. Disclosed in this report any change in the registrant's internal control over financial reporting that occurred during the registrant's most recent fiscal quarter (the registrant's fourth fiscal quarter in the case of an annual report) that has materially affected, or is reasonably likely to materially affect, the registrant's internal control over financial reporting; and
- 5. The registrant's other certifying officer and I have disclosed, based on our most recent evaluation of internal control over financial reporting, to the registrant's auditors and the audit committee of the registrant's board of directors (or persons performing the equivalent functions):
 - a. All significant deficiencies and material weaknesses in the design or operation of internal control over financial reporting which are reasonably likely to adversely affect the registrant's ability to record, process, summarize and report financial information; and
 - b. Any fraud, whether or not material, that involves management or other employees who have a significant role in the registrant's internal control over financial reporting.

July 28, 2016
/s/ VIKRAM VERMA
Vikram Verma
Chief Executive Officer

RULES 13a-14(a) AND 15d-14(a) UNDER THE SECURITIES EXCHANGE ACT OF 1934 AS ADOPTED PURSUANT TO

SECTION 302 OF THE SARBANES-OXLEY ACT OF 2002

- I, MaryEllen Genovese, certify that:
- 1. I have reviewed this quarterly report on Form 10-Q of 8x8, Inc.;
- 2. Based on my knowledge, this report does not contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements made, in light of the circumstances under which such statements were made, not misleading with respect to the period covered by this report;
- 3. Based on my knowledge, the financial statements, and other financial information included in this report, fairly present in all material respects the financial condition, results of operations and cash flows of the registrant as of, and for, the periods presented in this report;
- 4. The registrant's other certifying officer and I are responsible for establishing and maintaining disclosure controls and procedures (as defined in Exchange Act Rules 13a-15(e) and 15d-15(e)) and internal control over financial reporting (as defined in Exchange Act Rules 13a-15(f) and 15d-15(f)) for the registrant and have:
 - a. Designed such disclosure controls and procedures, or caused such disclosure controls and procedures to be designed under our supervision, to ensure that material information relating to the registrant, including its consolidated subsidiaries, is made known to us by others within those entities, particularly during the period in which this report is being prepared;
 - b. Designed such internal control over financial reporting, or caused such internal control over financial reporting to be designed under our supervision, to provide reasonable assurance regarding the reliability of financial reporting and the preparation of financial statements for external purposes in accordance with generally accepted accounting principles;
 - c. Evaluated the effectiveness of the registrant's disclosure controls and procedures and presented in this report our conclusions about the effectiveness of the disclosure controls and procedures as of the end of the period covered by this report based on such evaluation; and
 - d. Disclosed in this report any change in the registrant's internal control over financial reporting that occurred during the registrant's most recent fiscal quarter (the registrant's fourth fiscal quarter in the case of an annual report) that has materially affected, or is reasonably likely to materially affect, the registrant's internal control over financial reporting; and
- 5. The registrant's other certifying officer and I have disclosed, based on our most recent evaluation of internal control over financial reporting, to the registrant's auditors and the audit committee of the registrant's board of directors (or persons performing the equivalent functions):
 - All significant deficiencies and material weaknesses in the design or operation of internal control over financial reporting which are reasonably likely to adversely affect the registrant's ability to record, process, summarize and report financial information; and
 - b. Any fraud, whether or not material, that involves management or other employees who have a significant role in the registrant's internal control over financial reporting.

July 28, 2016
/s/ MARYELLEN GENOVESE
MaryEllen Genovese
Chief Financial Officer and Secretary

18 U.S. C. SECTION 1350, AS ADOPTED PURSUANT TO

SECTION 906 OF THE SARBANES-OXLEY ACT OF 2002

In connection with the Quarterly Report on Form 10-Q of 8x8, Inc. (the "Company") for the period ended June 30, 2016, as filed with the Securities and Exchange Commission on the date hereof (the "Report"), I, Vikram Verma, Chief Executive Officer of the Company, hereby certify, pursuant to 18 U.S.C. Section 1350, as adopted pursuant to Section 906 of the Sarbanes-Oxley Act of 2002, that:

- 1. The Report fully complies with the requirements of Section 13(a) or 15(d) of the Securities Exchange Act of 1934; and
- 2. The information contained in the Report fairly presents, in all material respects, the financial condition and results of operations of the Company.

/s/ VIKRAM VERMA Vikram Verma

Chief Executive Officer

July 28, 2016

This certification accompanies this Report pursuant to § 906 of the Sarbanes-Oxley Act of 2002 and shall not, except to the extent required by the Sarbanes-Oxley Act of 2002, or otherwise required, be deemed filed by the Company for purposes of § 18 of the Securities Exchange Act of 1934, as amended.

18 U.S. C. SECTION 1350, AS ADOPTED PURSUANT TO

SECTION 906 OF THE SARBANES-OXLEY ACT OF 2002

In connection with the Quarterly Report on Form 10-Q of 8x8, Inc. (the "Company") for the period ended June 30, 2016, as filed with the Securities and Exchange Commission on the date hereof (the "Report"), I, MaryEllen Genovese, Chief Financial Officer and Secretary of the Company, hereby certify, pursuant to 18 U.S.C. Section 1350, as adopted pursuant to Section 906 of the Sarbanes-Oxley Act of 2002, that:

- 1. The Report fully complies with the requirements of Section 13(a) or 15(d) of the Securities Exchange Act of 1934; and
- 2. The information contained in the Report fairly presents, in all material respects, the financial condition and results of operations of the Company.

/s/ MARYELLEN GENOVESE

MaryEllen Genovese Chief Financial Officer and Secretary

July 28, 2016

This certification accompanies this Report pursuant to § 906 of the Sarbanes-Oxley Act of 2002 and shall not, except to the extent required by the Sarbanes-Oxley Act of 2002, or otherwise required, be deemed filed by the Company for purposes of § 18 of the Securities Exchange Act of 1934, as amended.

Document and Entity 3 Months Ended Information - shares Jun. 30, 2016 Jul. 25, 2016

Document And Entity Information

Entity Registrant Name 8X8 INC /DE/ Entity Central Index Key 0001023731

<u>Document Type</u> 10-Q

Document Period End Date Jun. 30, 2016

Amendment Flag false
Current Fiscal Year End Date --03-31

Is Entity a Well-known Seasoned Issuer? Yes
Is Entity a Voluntary Filer? No
Is Entity's Reporting Status Current? Yes

Entity Filer Category Large Accelerated Filer

Entity Common Stock, Shares Outstanding 89,680,865

Document Fiscal Period Focus Q1
Document Fiscal Year Focus 2017

CONDENSED CONSOLIDATED

BALANCE SHEETS - USD

(\$)

\$ in Thousands

Cur	rrent	ass	ets:

Cash and cash equivalents	\$ 27,466	\$ 33,576
Short-term investments	139,607	129,274
Accounts receivable, net	11,538	11,070
Inventory	463	520
Deferred cost of goods sold	717	634
Deferred tax asset	0	5,382
Other current assets	5,096	5,444
Total current assets	184,887	185,900
Property and equipment, net	13,015	12,375
Intangible assets, net	19,531	21,464
Goodwill	45,931	47,420
Non-current deferred tax asset	48,532	43,189
Other assets	3,751	3,104
<u>Total assets</u>	315,647	313,452
Current liabilities:		
Accounts payable	12,649	10,954
Accrued compensation	9,631	10,063
Accrued warranty	340	326
Accrued taxes	5,390	5,200
Accrued outside commissions	2,070	2,186
<u>Deferred revenue</u>	2,120	1,925
Other accrued liabilities	3,226	4,080
Total current liabilities	35,426	34,734
Non-current liabilities	3,107	3,258
Non-current deferred revenue	134	154
<u>Total liabilities</u>	38,667	38,146
Commitments and contingencies (Note 6)	
Stockholders' equity:		
Common stock	89	89
Additional paid-in capital	394,100	389,260
Accumulated other comprehensive loss	(6,822)	(4,184)
Accumulated deficit	(110,387)	(109,859)
Total stockholders' equity	276,980	275,306
Total liabilities and stockholders' equity	\$ 315,647	\$ 313,452

Jun. 30, 2016 Mar. 31, 2016

3 Months Ended

Jun. 30, 2016 Jun. 30, 2015

CONDENSED CONSOLIDATED STATEMENTS OF OPERATIONS - USD (\$) shares in Thousands, \$ in

shares in Thousands, \$ in Thousands

Condensed	Consolidated	Statements	Of O	nerations
Conuchiscu	Consonuateu	Statements		oci ations

Service revenue	\$ 55,296	\$ 44,168
Product revenue	4,745	3,724
Total revenue	60,041	47,892
Operating expenses:		
Cost of service revenue	10,235	8,459
Cost of product revenue	5,505	4,382
Research and development	6,710	5,080
Sales and marketing	31,691	23,824
General and administrative	6,801	6,068
<u>Total operating expenses</u>	60,942	47,813
<u>Income (loss) from operations</u>	(901)	79
Other income, net	410	234
Income (loss) before provision for income taxes	(491)	313
<u>Provision for income taxes</u>	37	785
Net loss	\$ (528)	\$ (472)
Net loss per share:		
Basic	\$ (0.01)	\$ (0.01)
<u>Diluted</u>	\$ (0.01)	\$ (0.01)
Weighted average number of shares:		
Basic	89,434	88,233
<u>Diluted</u>	89,434	88,233

3 Months Ended

CONDENSED CONSOLIDATED STATEMENTS OF **COMPREHENSIVE**

INCOME (LOSS) - USD (\$)

Jun. 30, 2016 Jun. 30, 2015

\$ in Thousands

Statement of	'Com	<u>orehensive</u>	Income [Abstract

Net loss	\$ (528)	\$ (472)
Other comprehensive income (loss), net of tax		
Unrealized gain (loss) on investments in securities	146	(48)
Foreign currency translation adjustment	(2,784)	1,478
Comprehensive income (loss)	\$ (3,166)	\$ 958

CONDENSED CONSOLIDATED

STATEMENTS OF CASH FLOWS - USD (\$)

\$ in Thousands

Jun. 30, 2016 Jun. 30, 2015

3 Months Ended

5 III 1 HOUSAHUS		
Cash flows from operating activities:		
Net loss	\$ (528)	\$ (472)
Adjustments to reconcile net loss to net cash provided by operating activites		
<u>Depreciation</u>	1,471	993
Amortization of intangible assets	960	546
Amortization of capitalized software	146	456
Net accretion of discount and amortization of premium on marketable securities	100	236
Stock-based compensation	5,051	3,022
<u>Deferred income tax (benefit) provision</u>	(44)	476
<u>Other</u>	190	74
Changes in assets and liabilities:		
Accounts receivable, net	(1,043)	(612)
<u>Inventory</u>	53	88
Other current and noncurrent assets	(508)	(470)
Deferred cost of goods sold	(120)	(53)
Accounts payable	1,137	933
Accrued compensation	(354)	725
Accrued warranty	14	3
Accrued taxes	240	492
<u>Deferred revenue</u>	211	(704)
Accrued outside commissions	(116)	199
Other current and noncurrent liabilities	(324)	(1,272)
Net cash provided by operating activities	6,536	4,660
Cash flows from investing activities:		
Purchases of property and equipment	(1,604)	(1,073)
Purchase of businesses, net of cash acquired	0	(23,434)
Cost of capitalized software	(707)	(471)
Proceeds from maturity of investments	17,025	7,820
Sales of investments - available for sale	15,324	22,620
Purchase of investments - available for sale	(42,625)	(34,409)
Net cash used in investing activities	(12,587)	(28,947)
Cash flows from financing activities:		
<u>Capital lease payments</u>	(182)	(54)
Payment of contingent consideration and escrow	(200)	0
Repurchase of common stock	(629)	(25)
Proceeds from issuance of common stock under employee stock plans	1,039	336
Net cash provided by financing activities	28	257
Effect of exchange rate changes on cash	(87)	218
Net decrease in cash and cash equivalents	(6,110)	(23,812)

Cash and cash equivalents at the beginning of the period	33,576	53,110
Cash and cash equivalents at the end of the period	27,466	29,298
Supplemental cash flow information		
Income taxes paid	87	52
Interest paid	\$ 7	\$ 6

DESCRIPTION OF BUSINESS AND SIGNIFICANT ACCOUNTING POLICIES

3 Months Ended

Jun. 30, 2016

- Note 1

Notes to Financial
Statements

DESCRIPTION OF
BUSINESS AND
SIGNIFICANT
ACCOUNTING POLICIES Note 1

1. DESCRIPTION OF BUSINESS AND SIGNIFICANT ACCOUNTING POLICIES

DESCRIPTION OF BUSINESS

8x8, Inc. (8x8 or the Company) is a provider of cloud-based, enterprise-class software solutions that transform the way businesses communicate and collaborate globally. The Company's integrated, "pure-cloud" offering combines global voice, conferencing, messaging and video with integrated workflows and big data analytics on a single platform to enable increased team productivity, better customer engagement and real-time insights into business performance.

BASIS OF PRESENTATION

The Company's fiscal year ends on March 31 of each calendar year. Each reference to a fiscal year in these notes to the consolidated financial statements refers to the fiscal year ended March 31 of the calendar year indicated (for example, fiscal 2017 refers to the fiscal year ended March 31, 2017).

The accompanying interim condensed consolidated financial statements are unaudited and have been prepared on substantially the same basis as our annual consolidated financial statements for the fiscal year ended March 31, 2016. In the opinion of the Company's management, these interim condensed consolidated financial statements reflect all adjustments (consisting only of normal recurring adjustments) considered necessary for a fair statement of our financial position, results of operations and cash flows for the periods presented. The preparation of financial statements in conformity with generally accepted accounting principles requires management to make estimates and assumptions that affect the reported amounts of assets and liabilities and disclosure of contingent assets and liabilities at the date of the condensed consolidated financial statements and the reported amounts of revenue and expenses during the reporting periods. Actual results could differ from these estimates.

The March 31, 2016 year-end condensed consolidated balance sheet data in this document were derived from audited consolidated financial statements and does not include all of the disclosures required by U.S. generally accepted accounting principles. These condensed consolidated financial statements should be read in conjunction with the Company's audited consolidated financial statements as of and for the fiscal year ended March 31, 2016 and notes thereto included in the Company's fiscal 2016 Annual Report on Form 10-K.

The results of operations and cash flows for the interim periods included in these condensed consolidated financial statements are not necessarily indicative of the results to be expected for any future period or the entire fiscal year.

PRINCIPLES OF CONSOLIDATION

The consolidated financial statements include the accounts of 8x8 and its subsidiaries. All material intercompany accounts and transactions have been eliminated.

SIGNIFICANT ACCOUNTING POLICIES

The significant accounting policies used in preparation of these condensed consolidated financial statements are disclosed in our Annual Report on Form 10-K for the fiscal year ended March 31, 2016 filed with the SEC on May 31, 2016, and there have been no changes to the Company's significant accounting policies during the three months ended June 30, 2016, except as described in the "Recently Adopted Accounting Pronouncements" section below.

RECENTLY ADOPTED ACCOUNTING PRONOUNCEMENTS

In April 2015, the Financial Accounting Standards Board (FASB) issued Accounting Standards Update (ASU) 2015-5, "Intangibles-Goodwill and Other-Internal-Use Software (Subtopic 350-40): Customer's Accounting for Fees Paid in a Cloud Computing Arrangement." This update provides guidance in evaluating whether a cloud computing arrangement includes a software license. If a cloud computing arrangement includes a software license, then the software license element of the arrangement should be accounted for as an acquisition of a software license. If a

cloud computing arrangement does not include a software license, the customer should account for the arrangement as a service contract. The guidance does not change generally accepted accounting principles for a customer's accounting for service contracts. This update is effective for annual periods, including interim periods within those annual periods, beginning after December 15, 2015. Therefore, the Company has prospectively adopted this new standard on April 1, 2016. The adoption of this standard did not have a material impact on our consolidated financial statements.

In November 2015, the FASB issued ASU 2015-17, *Income Taxes - Balance Sheet Classification of Deferred Taxes (Topic 740)*. This ASU requires all deferred tax liabilities and assets to be presented in the balance sheet as noncurrent. As permitted, the Company early adopted this standard prospectively during the quarter ended June 30, 2016. The adoption of this standard resulted in reclassifying current deferred income tax assets to noncurrent deferred income tax assets and current deferred income tax liabilities to noncurrent deferred income tax liabilities. No prior periods were retrospectively adjusted.

RECENT ACCOUNTING PRONOUNCEMENTS

In May 2014, the FASB issued ASU No. 2014-09, *Revenue from Contracts with Customers*, which requires an entity to recognize the amount of revenue to which it expects to be entitled for the transfer of promised goods or services to customers. The ASU will replace most existing revenue recognition guidance in U.S. GAAP when it becomes effective. The new standard will become effective for the Company on April 1, 2018. The standard permits the use of either the retrospective or cumulative effect transition method. The Company has not yet selected a transition method.

In April 2016, the FASB issued ASU No. 2016-10, *Revenue from Contracts with Customers (Topic 606) - Identifying Performance Obligations and Licensing*. The ASU amends the guidance in ASU 2014-09 related to identifying performance obligations and accounting for licenses of intellectual property.

In May 2016, the FASB issued ASU No. 2016-12, Revenue from *Contracts with Customers (Topic 606): Narrow-Scope Improvements and Practical Expedients*. The amendment provides clarifying guidance in certain narrow areas and adds some practical expedients.

The Company is currently assessing the impact of this pronouncement and its amendments to its consolidated financial statements.

FAIR VALUE MEASUREMENTS - Note 2

3 Months Ended Jun. 30, 2016

Cash and Cash Equivalents
[Abstract]
FAIR VALUE
MEASUREMENTS - Note 2

2. FAIR VALUE MEASUREMENTS

Cash, cash equivalents, and available-for-sale investments, and contingent consideration were (in thousands):

As of June 30, 2016	A	amortized Costs	Į	Gross Inrealized Gain	Gross Unrealized Loss	I	Estimated Fair Value		Cash and Cash Equivalents		Short-Term
Cash	\$	22,440	\$	-	\$ - 5	\$	22,440	\$	22,440	\$	_
Level 1: Money market funds		5,026		-	-		5,026		5,026		-
Mutual funds		2,000		-	(175)		1,825		-		1,825
Subtotal Level 2:	_	29,466		-	(175)	-	29,291		27,466	-	1,825
Commercial paper		15,660		8	-		15,668		-		15,668
Corporate debt		88,620		143	(10)		88,753		-		88,753
Municipal securities		1,001		-	(1)		1,000		-		1,000
Asset backed securities		27,349		32	(1)		27,380		-		27,380
Mortgage backed securities		1,989		-	(14)		1,975		-		1,975
Agency		2,000		5	-		2,005		-		2,005
International government securities		1,000		1	-	_	1,001	ı	-		1,001
Subtotal	_	137,619		189	(26)		137,782	ī	-		137,782
Total assets	\$	167,085	\$	189	\$ (201) 5	\$	167,073	\$	27,466	\$	139,607
Level 3: Contingent consideration	\$	-	\$	-	\$ - 5	\$_	216	\$	-	\$_	<u>-</u>

Total | | - | | - | | 216 | | - | |

As of March 31, 2016	A	Amortized Costs	. 1	Gross Unrealized Gain		Gross Unrealized Loss]	Estimated Fair Value		Cash and Cash Equivalents		hort-Term
Cash	\$	18,596	\$	-	\$	-	\$	18,596	\$	18,596	\$	-
Level 1: Money market funds		14,980		-		-		14,980		14,980		-
Mutual funds	_	2,000		-	ı	(187)	-	1,813		-		1,813
Subtotal	_	35,576		-	ì	(187)	_	35,389	_	33,576	_	1,813
Level 2:												
Commercial paper		6,794		2		-		6,796		-		6,796
Corporate debt		85,164		78		(28)		85,214		-		85,214
Municipal securities		1,007		-		(1)		1,006		-		1,006
Asset backed securities		24,614		7		(11)		24,610		-		24,610
Mortgage backed securities		2,045		-		(17)		2,028		-		2,028
Agency bond		6,805		1		-		6,806		-		6,806
International government securities		1,000		1		-		1,001		-		1,001
Subtotal		127,429		89		(57)	-	127,461	•	-		127,461
Total assets	\$	163,005	\$	89	\$	(244)	\$	162,850	\$	33,576	\$	129,274
Level 3: Contingent consideration	\$	-	\$	-	\$	-	\$	341	\$	- :	\$	-
Total liabilities	\$	-	\$	-	\$	-	\$	341	\$	- ;	\$	-

Contractual maturities of investments as of June 30, 2016 are set forth below (in thousands):

Estimated
Fair Value
\$ 79,772

Due within one year

Due after one year 59,835

Total \$ 139,607

Contingent Consideration and Escrow Liability

The Company's contingent consideration liability and escrow liability, included in other accrued liabilities and noncurrent liabilities on the consolidated balance sheets, was associated with the Quality Software Corporation (QSC) acquisition made in the first quarter of fiscal 2016. Amounts held in escrow were measure at fair value using present value computations. The contingent consideration was measured at fair value using a probability weighted average of the potential payment outcomes that would occur should certain contract milestones be reached. There is no market data available to use in valuing the contingent consideration; therefore, the Company developed its own assumptions related to the achievement of the milestones to evaluate the fair value of the liability. As such, the contingent consideration is classified within Level 3 as described below.

The items are classified as Level 3 within the valuation hierarchy, consisting of contingent consideration and escrow liability related to the QSC acquisition, were valued based on an estimate of the probability of success of the milestones being achieved and present value computations, respectively. The table below presents a roll-forward of the contingent consideration and escrow liability valued using a Level 3 input (in thousands):

]	ee Months Ended e 30, 2016	N	Year Ended March 31, 2016		
Balance at beginning of period	\$	341	\$	-		
Purchase price contingent consideration		-		541		
Contingent consideration payments		(125)		(200)		
Balance at end of period	\$	216	\$	341		

INVENTORIES - Note 3

3 Months Ended Jun. 30, 2016

Notes to Financial Statements

INVENTORIES - Note 3

3. INVENTORIES

		June 30, 2016]	March 31, 2016
Inventory (in thousands)				
Work-in-process	\$	3	\$	76
Finished goods	_	460		444
Total	\$	463	\$	520

INTANGIBLE ASSETS AND GOODWILL - Note 4

3 Months Ended Jun. 30, 2016

Goodwill and Intangible Assets Disclosure [Abstract] INTANGIBLE ASSETS AND 4. INTANGIBLE ASSETS AND GOODWILL GOODWILL - Note 4

The carrying value of intangible assets consisted of the following (in thousands):

June 30, 2016				March 31, 2016					
	Gross			Net	-	Gross		Net	
(Carrying	A	ccumulated	Carrying	. (Carrying	Accumulated	Carrying	
1	Amount	A	mortization	Amount		Amount	Amortization	Amount	
\$	18,033	\$	(5,215)	\$ 12,818	\$	18,640	\$ (4,622)	\$ 14,018	
	0.716		(5.215)	4 501		0.003	(4 947)	5,146	
	9,/10		(3,213)	4,501		9,993	(4,047)	3,140	
	2 117			2 117		2 205		2,205	
	2,117		-	2,117		2,203	-	2,203	
	95		-	95		95	-	95	
_		_							
\$	29,961	\$	(10,430)	\$ 19,531	\$	30,933	\$ (9,469)	\$ 21,464	
_		=			: :				
	\$	Carrying	Gross Carrying A	Gross Accumulated Amount Amortization \$ 18,033 \$ (5,215) 9,716 (5,215) 2,117 - 95 -	Gross Net Carrying Amount Accumulated Amount Carrying Amount \$ 18,033 \$ (5,215) \$ 12,818 9,716 (5,215) 4,501 2,117 - 2,117 95 - 95	Gross Net Carrying Accumulated Carrying Amount Amortization Amount \$ 18,033 \$ (5,215) \$ 12,818 9,716 (5,215) 4,501 2,117 - 2,117 95 - 95	Gross Net Gross Carrying Amount Accumulated Amount Carrying Amount Amount Amount \$ 18,033 \$ (5,215) \$ 12,818 \$ 18,640 9,716 (5,215) 4,501 9,993 2,117 - 2,117 2,205 95 - 95 95	Gross Net Accumulated Amount Gross Carrying Accumulated Amount Carrying Amount Amount Accumulated Amount Amount Amount Amount Amount Accumulated Amount Amount Amount \$ 18,033 \$ (5,215) \$ 12,818 \$ 18,640 \$ (4,622) 9,716 (5,215) 4,501 9,993 (4,847) 2,117 - 2,117 2,205 - 95 - 95 95 -	

At June 30, 2016, annual amortization of intangible assets, based upon our existing intangible assets and current useful lives, is estimated to be the following (in thousands):

	_	Amount
Remaining 2017	\$	2,758
2018		3,411
2019		3,162
2020		3,162
2021		2,796
Thereafter	_	2,030
Total	\$	17,319

The following table provides a summary of the changes in the carrying amounts of goodwill by reporting segment (in thousands):

	Α	Americas	Europe	Total
Balance as of March 31, 2016	\$	25,729 \$	3 21,691 \$	47,420
Foreign currency translation	_	_	(1,489)	(1,489)
Balance as of June 30, 2016	\$	25,729	3 20,202 \$	45,931

RESEARCH, DEVELOPMENT AND SOFTWARE COSTS - Note 5

Jun. 30, 2016

3 Months Ended

Research and Development
[Abstract]
RESEARCH,
DEVELOPMENT AND
SOFTWARE COSTS - Note 5

5. RESEARCH, DEVELOPMENT AND SOFTWARE COSTS

In the first three months of fiscal 2017 and 2016, the Company expensed all research and development costs in accordance with ASC 985-20, *Costs of Software to be Sold, Leased or Marketed* (ASC 985-20).

The Company accounts for computer software developed or obtained for internal use in accordance with ASC 350-40, *Internal Use Software* (ASC 350-40). Capitalized costs are classified as either long-term assets or property and equipment on the consolidated balance sheets.

Other Long-Term Assets

In the first three months of fiscal 2017, the Company capitalized \$0.7 million as other long-term assets. In the first three months of fiscal 2016, the Company capitalized \$0.5 million as other long-term assets. At June 30, 2016 and March 31, 2016, total completed capitalized software development costs included in other long-term assets was approximately \$1.7 million and \$0, respectively. At June 30, 2016 and March 31, 2016, accumulated amortization costs related to capitalized software in other long term assets was approximately \$0.1 million and \$0, respectively.

Property and Equipment

In the first three months of fiscal 2017, the Company capitalized \$0.3 million as property and equipment. The Company did not capitalize any costs in property and equipment in accordance with ASC 350-40 in the first three months of fiscal 2016. At June 30, 2016 and March 31, 2016, total completed capitalized software costs included in property and equipment was approximately \$1.2 million. At June 30, 2016 and March 31, 2016, accumulated amortization costs related to capitalized software in property and equipment was approximately \$0.3 million and \$0.2 million, respectively.

COMMITMENTS AND CONTINGENCIES - Note 6

3 Months Ended Jun. 30, 2016

Notes to Financial
Statements
COMMITMENTS AND
CONTINGENCIES - Note 6

6. COMMITMENTS AND CONTINGENCIES

Leases

The Company leases its headquarters facility in San Jose, California under an operating lease agreement that expires in October 2019. The lease is an industrial net lease with monthly base rent of \$130,821 for the first 15 months with a 3% increase each year thereafter, and requires us to pay property taxes, utilities and normal maintenance costs. The Company also leases facilities for office space under non-cancelable operating leases for its various domestic and international locations

The Company has entered into a series of noncancelable capital lease agreements for office equipment bearing interest at various rates. Assets under capital lease at June 30, 2016 totaled \$1.6 million with accumulated amortization of \$0.6 million.

Guarantees

Indemnifications

In the normal course of business, the Company may agree to indemnify other parties, including customers, lessors and parties to other transactions with the Company, with respect to certain matters such as breaches of representations or covenants or intellectual property infringement or other claims made by third parties. These agreements may limit the time within which an indemnification claim can be made and the amount of the claim. In addition, the Company has entered into indemnification agreements with its officers and directors.

It is not possible to determine the maximum potential amount of the Company's exposure under these indemnification agreements due to the limited history of prior indemnification claims and the unique facts and circumstances involved in each particular agreement. Historically, payments made by the Company under these agreements have not had a material impact on the Company's operating results, financial position or cash flows. Under some of these agreements, however, the Company's potential indemnification liability might not have a contractual limit.

Product Warranties

The Company accrues for the estimated costs that may be incurred under its product warranties upon revenue recognition. Changes in the Company's product warranty liability, which is included in cost of product revenues in the consolidated statements of income (loss), were as follows (in thousands):

Three Months Ended

	June 30,				
		2016		2015	
Balance at beginning of period	\$	326	\$	339	
Accruals for warranties		114		98	
Settlements		(87)		(83)	
Adjustments		(13)		(12)	
Balance at end of period	\$	340	\$	342	

Minimum Third Party Customer Support Commitments

In the third quarter of 2010, the Company amended its contract with one of its third party customer support vendors containing a minimum monthly commitment of approximately \$0.4 million effective April 1, 2010. The agreement requires a 150-day notice to terminate. At June 30, 2016, the total remaining obligation upon a termination of the contract was \$2.2 million. *Minimum Third Party Network Service Provider Commitments*

The Company has entered into contracts with multiple vendors for third party network service which expire on various dates in fiscal 2017 through 2020. At June 30, 2016, future minimum annual payments under these third party network service contracts were as follows (in thousands): Year ending March 31:

Remaining 2017 \$ 1,961

2018	981
2019	32
2020	8
Total minimum payments	\$ 2,982

Legal Proceedings

The Company, from time to time, is involved in various legal claims or litigation, including patent infringement claims that can arise in the normal course of the Company's operations. Pending or future litigation could be costly, could cause the diversion of management's attention and could upon resolution, have a material adverse effect on the Company's business, results of operations, financial condition and cash flows.

On February 22, 2011, the Company was named a defendant in Bear Creek Technologies, Inc. v. 8x8, Inc. et al., along with 20 other defendants. On August 17, 2011, the suit was dismissed without prejudice as to the Company under Rule 21 of the Federal Rules of Civil Procedure. On August 17, 2011, Bear Creek Technologies, Inc. refiled its suit against the Company in the United States District Court for the District of Delaware. Further, on November 28, 2012, the U.S. Patent & Trademark Office initiated a Reexamination proceeding with a Reexamination Declaration explaining that there is a substantial new question of patentability, based on four separate grounds and affecting each claim of the patent which is the basis for the complaint filed against us. On March 26, 2013, the USPTO issued a first Office Action in the Reexamination, with all claims of the '722 patent being rejected on each of the four separate grounds raised in the Request for Reexamination. On July 10, 2013, the Company filed an informational pleading in support of and joining a motion to stay the proceeding in the District Court; the District Court granted the motion on July 17, 2013, based on the possibility that at least one of the USPTO rejections will be upheld and considering the USPTO's conclusion that Bear Creek's patent suffers from a defective claim for priority. On March 24, 2014, the USPTO issued another Office Action in which the rejections of the claims were maintained. On August 15, 2014, the USPTO issued a Right of Appeal Notice, as the USPTO maintained all rejections of the patent claims. On September 15, 2014, Bear Creek Technologies, Inc. filed a Notice of Appeal of this decision with the Patent Trial and Appeal Board. The case is currently on appeal. The Company believes that it has meritorious defenses to these claims and is presenting a vigorous defense, but the Company cannot estimate potential liability in this case at this early stage of litigation. On April 16, 2015, the Company was named as a defendant in Slocumb Law Firm v. 8x8, Inc. The Slocumb Law Firm has alleged that it purchased certain business services from the Company that did not perform as advertised or expected, and has asserted causes of actions for fraud, breach of contract, violations of the Alabama Deceptive Trade Practices Act and negligence. On May 7, 2015, the Company filed a motion with the U.S. District Court for the Middle District of Alabama, seeking an order compelling the Slocumb Law Firm to arbitrate its claims against the Company in Santa Clara County, California pursuant to a clause mandating arbitration of disputes set forth in the terms and conditions to which Slocumb Law Firm agreed in connection with its purchase of business services from the Company. No briefing schedule or hearing date for the motion has been set as of this time. Discovery has not yet commenced in the case. The Company intends to vigorously defend against Slocumb Law Firm's claims.

State and Municipal Taxes

From time to time, the Company has received inquiries from a number of state and municipal taxing agencies with respect to the remittance of sales, use, telecommunications, excise, and income taxes. Several jurisdictions currently are conducting tax audits of the Company's records. The Company collects or has accrued for taxes that it believes are required to be remitted. The amounts that have been remitted have historically been within the accruals established by the Company.

STOCK-BASED COMPENSATION - Note 7

3 Months Ended Jun. 30, 2016

Three Months Ended

Equity [Abstract]
STOCK-BASED
COMPENSATION - Note 7

7. STOCK-BASED COMPENSATION

The following table summarizes stock-based compensation expense (in thousands):

	Three Months End			Bilaca
		Jui),	
		2016		2015
Cost of service revenue	\$	360	\$	219
Cost of product revenue		-		-
Research and development		887		531
Sales and marketing		1,915		1,197
General and administrative	_	1,889		1,075
Total stock-based compensation expense related to employee				
stock options and employee stock purchases, pre-tax		5,051		3,022
Tax benefit	_	-	_	-
Stock-based compensation expense related to employee				
stock options and employee stock purchases, net of tax	\$_	5,051	\$_	3,022

Stock Options, Stock Purchase Right and Restricted Stock Unit Activity

Stock Option activity under all the Company's stock option plans for the three months ended June 30, 2016, is summarized as follows:

		Weighted Average
	Number of	Exercise Price
	Shares	Per Share
Outstanding at March 31, 2016	4,793,266 \$	6.29
Granted	53,976	11.31
Exercised	(191,855)	2.18
Canceled/Forfeited	(1)	2.58
Outstanding at June 30, 2016	4,655,386 \$	6.52
Vested and expected to vest at June 30, 2016	4,655,386 \$	6.52
Exercisable at June 30, 2016	3,014,307 \$	5.45

Stock Purchase Right activity for the three months ended June 30, 2016 is summarized as follows:

		Weighted Average Grant-Date	Weighted Average Remaining	
	Number of Shares	Fair Market Value	Contractual Term (in Years)	
Balance at March 31, 2016 Granted Vested	82,171 S - (9,475)	6.30	0.76	

Forfeited	_ _	-	
Balance at June 30, 2016	72,696 \$	6.31	0.45

Restricted Stock Unit activity for the three months ended June 30, 2016 is summarized as follows:

	Number of Shares	Weighted Average Grant Date Fair Value	Weighted Average Remaining Contractual Term (in Years)
Balance at March 31, 2016	4,544,799 \$	8.08	1.67
Granted	300,942	11.40	
Vested	(311,949)	8.02	
Forfeited	(101,193)	8.48	
Balance at June 30, 2016	4,432,599 \$	8.32	1.53

The following table summarizes stock options outstanding and exercisable at June 30, 2016:

	Options Outstanding			Options Exercisable					
	Shares		Weighted Average Exercise Price Per Share	Weighted Average Remaining Contractual Life (Years)	-	Shares		Weighted Average Exercise Price Per Share	Aggregate Intrinsic Value
\$0.55 to \$1.27	933,625	\$	1.12	1.5	\$12,596,169	933,625	\$	1.12	\$12,596,169
\$1.39 to \$6.86	1,477,028	\$	5.83	6.8	12,968,088	1,050,590	\$	5.43	9,643,718
\$7.52 to \$8.93	973,650	\$	8.28	8.8	6,165,143	271,615	\$	8.22	1,735,944
\$9.21 to \$9.74	983,835	\$	9.62	7.2	4,905,183	666,290	\$	9.63	3,319,570
\$10.50 to \$11.31	287,248	\$	11.02	8.5	1,031,611	92,187	\$	11.10	323,576
	4,655,386				\$ <u>37,666,194</u>	3,014,307			\$ <u>27,618,977</u>

As of June 30, 2016, there was \$32.9 million of unamortized stock-based compensation expense related to unvested stock options and awards which is expected to be recognized over a weighted average period of 2.22 years.

Unamortized stock-based compensation expense related to shares issued as part of an acquisition was approximately \$1.8 million, which will be recognized over a weighted average period of 2.92 years.

Assumptions Used to Calculate Stock-Based Compensation Expense

The fair value of each of the Company's option grants has been estimated on the date of grant using the Black-Scholes pricing model with the following assumptions:

Three Months Ended				
Jun	e 30,			
2016	2015			
51%	53%			

Expected volatility

Expected dividend yield	-	-
Risk-free interest rate	1.40%	1.59%
Weighted average expected option term	5.75 years	5.25 years
Weighted average fair value of options granted	\$ 5.45 \$	4.17

Stock Repurchases

In February 2015, the Company's board of directors authorized the Company to purchase up to \$20.0 million of its common stock from time to time until February 29, 2016 (the "2015 Repurchase Plan"). Share repurchases, if any, will be funded with available cash. Repurchases under the 2015 Repurchase Plan may be made through open market purchases at prevailing market prices or in privately negotiated transactions. The timing, volume and nature of share repurchases are subject to market prices and conditions, applicable securities laws and other factors, and are at the discretion of the Company's management. Share repurchases under the 2015 Repurchase Plan may be commenced, suspended or discontinued at any time. This tranche of shares authorized for repurchase expired in February 2016.

In October 2015, the Company's board of directors authorized the Company to purchase an additional \$15.0 million of its common stock from time to time until October 20, 2016 under the 2015 Repurchase Plan. There were no stock repurchases for the period ended June 30, 2016. The remaining authorized repurchase amount at June 30, 2016 was \$15.0 million.

INCOME TAXES - Note 8

3 Months Ended **Jun. 30, 2016**

Notes to Financial
Statements
INCOME TAXES - Note 8

8. INCOME TAXES

For the three months ended June 30, 2016, the Company recorded a provision for income taxes of \$37,000, which was primarily attributable to income from operations. For the three months ended June 30, 2015, the Company recorded a provision for income taxes of \$0.8 million which was primarily attributable to income from operations.

The effective tax rate is calculated by dividing the income tax provision by net income before income tax expense.

At March 31, 2016, the Company had a liability for unrecognized tax benefits of \$2.9 million, all of which, if recognized, would decrease the company's effective tax rate. The Company does not believe that there has been any significant change in the unrecognized tax benefits in the three-month period ended June 30, 2016, and does not expect the remaining unrecognized tax benefit to change materially in the next 12 months. To the extent that the remaining unrecognized tax benefits are ultimately recognized, they will have an impact on the effective tax rate in future periods.

The Company is subject to taxation in the U.S., California and various other states and foreign jurisdictions in which it has or had a subsidiary or branch operations or it is collecting sales tax. All tax returns from fiscal 2013 to fiscal 2016 may be subject to examination by the Internal Revenue Service, California and various other states. As of July 22, 2016, there were no active federal or state income tax audits. Returns filed in foreign jurisdictions may be subject to examination for the fiscal years 2011 to 2016.

NET INCOME (LOSS) PER SHARE - Note 9

3 Months Ended Jun. 30, 2016

Notes to Financial
Statements
NET INCOME (LOSS) PER
SHARE - Note 9

9. NET INCOME (LOSS) PER SHARE

The following is a reconciliation of the weighted average number of common shares outstanding used in calculating basic and diluted net income (loss) per share (in thousands, except share and per share data):

	Three Months Ended June 30,			
		2016		2015
Numerator:				
Net loss available to common stockholders	\$	(528)	\$	(472)
Denominator:				
Common shares	_	89,434	_	88,233
Denominator for basic calculation		89,434		88,233
Employee stock options		-		-
Stock purchase rights		_		-
Denominator for diluted calculation	_	89,434	_	88,233
Net loss per share				
Basic	\$	(0.01)	\$	(0.01)
Diluted	\$	(0.01)	\$	(0.01)

The following shares attributable to outstanding stock options and restricted stock purchase rights were excluded from the calculation of diluted earnings per share because their inclusion would have been antidilutive (in thousands):

	Three Months Ended June 30,		
	2016	2015	
Employee stock options	1,528	2,447	
Stock purchase rights	1,602	70	
Total anti-dilutive employee stock-based securities	3,130	2,517	

SEGMENT REPORTING -Note 10

3 Months Ended Jun. 30, 2016

Segment Reporting
[Abstract]
SEGMENT REPORTING Note 10

10. SEGMENT REPORTING

ASC 280, Segment Reporting, establishes annual and interim reporting standards for an enterprise's business segments and related disclosures about its products, services, geographic areas and major customers. Under ASC 280, the method for determining what information to report is based upon the way management organizes the operating segments within the Company for making operating decisions and assessing financial performance.

The Company manages its operations primarily on a geographic basis. The Chief Executive Officer, the Chief Financial Officer, and the Chief Technology Officer or the Company's Chief Operating Decision Makers (CODMs), evaluate performance of the Company and make decisions regarding allocation of resources based on geographic results. The Company's reportable segments are the Americas and Europe. The Americas segment is primarily North America. The Europe segment is primarily the United Kingdom. Each operating segment provides similar products and services.

The Company's CODMs evaluate the performance of its operating segments based on revenues and net income. Revenues are attributed to each segment based on the ordering location of the customer or ship to location. The Company allocates corporate overhead costs such as research and development, sales and marketing, general and administrative, amortization expense, stock-based compensation expense, and commitment and contingencies to the Americas segment. The Company's revenue distribution by geographic region (based upon the destination of shipments and the customer's service address) was as follows:

Three Months Ended June 30,

	2016	2015
Americas (principally US)	88%	88%
Europe	11%	9%
Asia Pacific	1%	3%
	100%	100%

Geographic area data is based upon the location of the property and equipment and is as follows (in thousands):

	June 30,		March 31,		
	 2016		2016		
Americas (principally US)	\$ 9,643	\$	9,165		
Europe	3,126		2,642		
Asia-Pacific	 246		568		
Total	\$ 13,015	\$	12,375		

The following table provides financial information by segment for the three month period ending June 30, 2016 and 2015 (in thousands):

Three Months Ended June 30.

		2016		2015	
Americas (principally US):			· ·		
Net revenue	\$	53,398	\$	43,588	
Net income	\$	1,466	\$	251	
Europe:					
Net revenue	\$	6,643	\$	4,304	

Net loss \$ (1,994) \$ (723)

Summary of Significant Accounting Policies (Policies)

3 Months Ended **Jun. 30, 2016**

Notes to Financial Statements
Fiscal Period

The Company's fiscal year ends on March 31 of each calendar year. Each reference to a fiscal year in these notes to the consolidated financial statements refers to the fiscal year ended March 31 of the calendar year indicated (for example, fiscal 2017 refers to the fiscal year ended March 31, 2017).

Use of Estimates

The accompanying interim condensed consolidated financial statements are unaudited and have been prepared on substantially the same basis as our annual consolidated financial statements for the fiscal year ended March 31, 2016. In the opinion of the Company's management, these interim condensed consolidated financial statements reflect all adjustments (consisting only of normal recurring adjustments) considered necessary for a fair statement of our financial position, results of operations and cash flows for the periods presented. The preparation of financial statements in conformity with generally accepted accounting principles requires management to make estimates and assumptions that affect the reported amounts of assets and liabilities and disclosure of contingent assets and liabilities at the date of the condensed consolidated financial statements and the reported amounts of revenue and expenses during the reporting periods. Actual results could differ from these estimates.

Basis of Accounting

The March 31, 2016 year-end condensed consolidated balance sheet data in this document were derived from audited consolidated financial statements and does not include all of the disclosures required by U.S. generally accepted accounting principles. These condensed consolidated financial statements should be read in conjunction with the Company's audited consolidated financial statements as of and for the fiscal year ended March 31, 2016 and notes thereto included in the Company's fiscal 2016 Annual Report on Form 10-K.

The results of operations and cash flows for the interim periods included in these condensed consolidated financial statements are not necessarily indicative of the results to be expected for any future period or the entire fiscal year.

The significant accounting policies used in preparation of these condensed consolidated financial statements are disclosed in our Annual Report on Form 10-K for the fiscal year ended March 31, 2016 filed with the SEC on May 31, 2016, and there have been no changes to the Company's significant accounting policies during the three months ended June 30, 2016, except as described in the "Recently Adopted Accounting Pronouncements" section below.

Principles of Consolidation

PRINCIPLES OF CONSOLIDATION

The consolidated financial statements include the accounts of 8x8 and its subsidiaries. All material intercompany accounts and transactions have been eliminated.

Recently Adopted Accounting Pronouncements

Recently Adopted Accounting RECENTLY ADOPTED ACCOUNTING PRONOUNCEMENTS

In April 2015, the Financial Accounting Standards Board (FASB) issued Accounting Standards Update (ASU) 2015-5, "Intangibles-Goodwill and Other-Internal-Use Software (Subtopic 350-40): Customer's Accounting for Fees Paid in a Cloud Computing Arrangement." This update provides guidance in evaluating whether a cloud computing arrangement includes a software license. If a cloud computing arrangement includes a software license, then the software license element of the arrangement should be accounted for as an acquisition of a software license. If a cloud computing arrangement does not include a software license, the customer should account for the arrangement as a service contract. The guidance does not change generally accepted accounting principles for a customer's accounting for service contracts. This update is effective for annual periods, including interim periods within those annual periods, beginning after

December 15, 2015. Therefore, the Company has prospectively adopted this new standard on April 1, 2016. The adoption of this standard did not have a material impact on our consolidated financial statements.

In November 2015, the FASB issued ASU 2015-17, *Income Taxes - Balance Sheet Classification of Deferred Taxes (Topic 740)*. This ASU requires all deferred tax liabilities and assets to be presented in the balance sheet as noncurrent. As permitted, the Company early adopted this standard prospectively during the quarter ended June 30, 2016. The adoption of this standard resulted in reclassifying current deferred income tax assets to noncurrent deferred income tax assets and current deferred income tax liabilities to noncurrent deferred income tax liabilities. No prior periods were retrospectively adjusted.

Recent Accounting Pronouncements

RECENT ACCOUNTING PRONOUNCEMENTS

In May 2014, the FASB issued ASU No. 2014-09, *Revenue from Contracts with Customers*, which requires an entity to recognize the amount of revenue to which it expects to be entitled for the transfer of promised goods or services to customers. The ASU will replace most existing revenue recognition guidance in U.S. GAAP when it becomes effective. The new standard will become effective for the Company on April 1, 2018. The standard permits the use of either the retrospective or cumulative effect transition method. The Company has not yet selected a transition method.

In April 2016, the FASB issued ASU No. 2016-10, *Revenue from Contracts with Customers* (*Topic 606*) - *Identifying Performance Obligations and Licensing*. The ASU amends the guidance in ASU 2014-09 related to identifying performance obligations and accounting for licenses of intellectual property.

In May 2016, the FASB issued ASU No. 2016-12, Revenue from *Contracts with Customers (Topic 606): Narrow-Scope Improvements and Practical Expedients*. The amendment provides clarifying guidance in certain narrow areas and adds some practical expedients.

The Company is currently assessing the impact of this pronouncement and its amendments to its consolidated financial statements.

Research, Development and Software Costs

RESEARCH. DEVELOPMENT AND SOFTWARE COSTS

In the first three months of fiscal 2016, the Company expensed all research and development costs in accordance with ASC 985-20, *Costs of Software to be Sold, Leased or Marketed* (ASC 985-20).

The Company accounts for computer software developed or obtained for internal use in accordance with ASC 350-40, *Internal Use Software* (ASC 350-40).

Warranty Expense

WARRANTY EXPENSE

The Company accrues for the estimated costs that may be incurred under its product warranties upon revenue recognition.

<u>Indemnifications</u>

Indemnifications

In the normal course of business, the Company may agree to indemnify other parties, including customers, lessors and parties to other transactions with the Company, with respect to certain matters such as breaches of representations or covenants or intellectual property infringement or other claims made by third parties. These agreements may limit the time within which an indemnification claim can be made and the amount of the claim. In addition, the Company has entered into indemnification agreements with its officers and directors.

It is not possible to determine the maximum potential amount of the Company's exposure under these indemnification agreements due to the limited history of prior indemnification claims and

the unique facts and circumstances involved in each particular agreement. Historically, payments made by the Company under these agreements have not had a material impact on the Company's operating results, financial position or cash flows. Under some of these agreements, however, the Company's potential indemnification liability might not have a contractual limit.

Segment Reporting

SEGMENT REPORTING

ASC 280, *Segment Reporting*, establishes annual and interim reporting standards for an enterprise's business segments and related disclosures about its products, services, geographic areas and major customers. Under ASC 280, the method for determining what information to report is based upon the way management organizes the operating segments within the Company for making operating decisions and assessing financial performance.

The Company manages its operations primarily on a geographic basis. The Chief Executive Officer, the Chief Financial Officer, and the Chief Technology Officer or the Company's Chief Operating Decision Makers (CODMs), evaluate performance of the Company and make decisions regarding allocation of resources based on geographic results. The Company's reportable segments are the Americas and Europe. The Americas segment is primarily North America. The Europe segment is primarily the United Kingdom. Each operating segment provides similar products and services.

The Company's CODMs evaluate the performance of its operating segments based on revenues and net income. Revenues are attributed to each segment based on the ordering location of the customer or ship to location. The Company allocates corporate overhead costs such as research and development, sales and marketing, general and administrative, amortization expense, stockbased compensation expense, and commitment and contingencies to the US segment.

FAIR VALUE MEASUREMENTS (Tables)

3 Months Ended Jun. 30, 2016

Fair Value Measurements Tables

Fair Value Measurements, Recurring and Nonrecurring (Tables) Cash, cash equivalents, and available-for-sale investments, and contingent consideration were (in thousands):

As of June 30, 2016		Gross Unrealized Gain	Gross Unrealized Loss	Estimated Fair Value		Short-Term Investments
Cash	\$ 22,440	\$ -	\$ -	\$ 22,440	\$ 22,440	\$ -
Level 1: Money market funds Mutual funds	5,026 2,000	-	- (175)	5,026 1,825	5,026	1,825
Subtotal Level 2:	29,466		(175)	29,291	27,466	1,825
Commercial paper	15,660	8	-	15,668	-	15,668
Corporate debt	88,620	143	(10)	88,753	-	88,753
Municipal securities	1,001	-	(1)	1,000	-	1,000
Asset backed securities	27,349	32	(1)	27,380	-	27,380
Mortgage backed securities	1,989	-	(14)	1,975	-	1,975
Agency bond	2,000	5	-	2,005	-	2,005
International government securities	1,000	1	-	1,001	-	1,001
Subtotal	137,619	189	(26)	137,782	-	137,782
Total assets Level 3:	\$ 167,085	\$ 189	\$ (201)	\$ 167,073	\$ 27,466	\$ 139,607
Contingent consideration	\$ -	\$ -	\$ -	\$ 216	\$ -	\$ -
Total liabilities	\$ -	\$	\$	\$ 216	\$	\$

As of March 31, 2016	A	mortized Costs	Į	Gross Inrealized Gain	-	Gross Unrealized Loss		Estimated Fair Value		Cash and Cash Equivalents		Short-Term
Cash	\$	18,596	\$	-	\$	_	\$	18,596	\$	18,596	\$	_
Level 1: Money market funds Mutual		14,980 2,000		-		(187)		14,980		14,980		1,813
funds		2,000				(107)		1,813		-		1,613
Subtotal Level 2:		35,576	_	-		(187)	·	35,389	•	33,576	_	1,813
Commercial paper		6,794		2		-		6,796		-		6,796
Corporate debt		85,164		78		(28)		85,214		-		85,214
Municipal securities		1,007		-		(1)		1,006		-		1,006
Asset backed securities		24,614		7		(11)		24,610		-		24,610
Mortgage backed securities		2,045		-		(17)		2,028		-		2,028
Agency bond		6,805		1		-		6,806		-		6,806
International government securities		1,000		1		-		1,001		-		1,001
Subtotal	_	127,429	_	89		(57)		127,461	-	-		127,461
Total assets Level 3:	\$ _	163,005	\$	89	\$	(244)	\$	162,850	\$	33,576	\$ =	129,274
Contingent consideration	\$	-	\$	-	\$	-	\$	341	\$	- :	\$ _	-
Total liabilities	\$_	-	\$	-	\$	-	\$	341	\$	- 5	\$ =	-

Contractual maturities of investments as of June 30, 2016 are set forth below (in thousands):

	E	Estimated		
	F	air Value		
Due within one year	\$	79,772		
Due after one year		59,835		

Total \$___139,607

Rollforward of contingent consideration liabilities (Tables)

The table below presents a roll-forward of the contingent consideration and escrow liability valued using a Level 3 input (in thousands):

	I	ee Months Ended e 30, 2016	Year Ended March 31, 2016	
Balance at beginning of period	\$	341	\$	-
Purchase price contingent consideration		-		541
Contingent consideration payments		(125)		(200)
Balance at end of period	\$	216	\$	341

Inventories (Tables)

3 Months Ended Jun. 30, 2016

<u>Inventories Tables</u> <u>Inventories (Tables)</u>

	_	June 30, 2016	March 31, 2016		
Inventory (in thousands)					
Work-in-process	\$	3	\$ 76		
Finished goods	_	460	 444		
Total	\$	463	\$ 520		

Intangible Assets (Tables)

3 Months Ended Jun. 30, 2016

Intangible Assets Tables

<u>Carrying values of intangible assets</u>

The carrying value of intangible assets consisted of the following (in thousands):

	June 30, 2016					March 31, 2016				
	(Gross Carrying Amount	Accumulated Amortization		Net Carrying Amount	Gross Carrying Amount	Accumulated Amortization	Net Carrying Amount		
Technology	\$	18,033	\$ (5,215)	\$	12,818	\$ 18,640	\$ (4,622)	\$ 14,018		
Customer relationships		9,716	(5,215)		4,501	9,993	(4,847)	5,146		
Trade names/ domains		2,117	-		2,117	2,205	-	2,205		
In-process research and development		95	-		95	95	-	95		
Total acquired identifiable intangible assets	\$	29,961	\$ (10,430)	\$	19,531	\$ 30,933	\$ (9,469)	\$ 21,464		

<u>Finite-lived intangible assets - future amortization expense</u>

At June 30, 2016, annual amortization of intangible assets, based upon our existing intangible assets and current useful lives, is estimated to be the following (in thousands):

		Amount
Remaining 2017	\$	2,758
2018		3,411
2019		3,162
2020		3,162
2021		2,796
Thereafter	_	2,030
Total	\$	17,319

Goodwill (Tables)

3 Months Ended Jun. 30, 2016

Goodwill Tables

Carrying amounts of goodwill The following table provides a summary of the changes in the carrying amounts of goodwill by reporting segment (in thousands):

	A	mericas	Europe	Total
Balance as of March 31, 2016	\$	25,729 \$	21,691 \$	47,420
Foreign currency translation			(1,489)	(1,489)
Balance as of June 30, 2016	\$	25,729 \$	20,202 \$	45,931

Commitments And Contingencies (Tables)

Commitments And Contingencies Tables

Product warranties

3 Months Ended Jun. 30, 2016

Changes in the Company's product warranty liability, which is included in cost of product revenues in the consolidated statements of income (loss), were as follows (in thousands):

Three Months Ended June 30.

		2016		2015
Balance at beginning of period	\$	326	\$	339
Accruals for warranties		114		98
Settlements		(87)		(83)
Adjustments		(13)		(12)
Balance at end of period	\$	340	\$	342

Minimum third party network service provider commitments

At June 30, 2016, future minimum annual payments under these third party network service contracts were as follows (in thousands):

Year ending March 31:

Remaining 2017	\$ 1,961
2018	981
2019	32
2020	8
Total minimum payments	\$ 2,982

Distribution of Stock-Based Compensation Plan Expense (Tables)

<u>Distribution Of Stock-based Compensation Plan Expense</u> <u>Tables</u>

<u>Schedule Of Stock-Based Compensation Expense By</u> <u>Statement Of Operations Line Item</u>

3 Months Ended Jun. 30, 2016

The following table summarizes stock-based compensation expense (in thousands):

•	Three Months Ended June 30,		
	2016		2015
Cost of service revenue	\$ 360	\$	219
Cost of product revenue	-		-
Research and development	887		531
Sales and marketing	1,915		1,197
General and administrative	1,889		1,075
Total stock-based compensation expense related to employee stock options and employee stock purchases, pre-tax	5,051		3,022
Tax benefit	-		
Stock-based compensation expense			
related to employee			
stock options and employee stock purchases, net of tax	\$ 5,051	\$	3,022

Stock-Based Compensation And Employee Purchase Plan (Tables)

3 Months Ended Jun. 30, 2016

Option Grants

<u>Disclosure Of Share-Based Compensation</u> <u>Arrangements By Share-Based Payment</u> Award Stock Option activity under all the Company's stock option plans for the three months ended June 30, 2016, is summarized as follows:

	Number of Shares	Average Exercise Price Per Share
Outstanding at March 31, 2016	4,793,266	6.29
Granted	53,976	11.31
Exercised	(191,855)	2.18
Canceled/Forfeited	(1)	2.58
Outstanding at June 30, 2016	4,655,386	6.52
Vested and expected to vest at June 30, 2016	4,655,386	6.52
Exercisable at June 30, 2016	3,014,307	5.45

Schedule of Share-based Payment Award, Stock Options, Valuation Assumptions The fair value of each of the Company's option grants has been estimated on the date of grant using the Black-Scholes pricing model with the following assumptions:

	June 30,			
	2016	2015		
Expected volatility	51%	53%		
Expected dividend yield	-	-		
Risk-free interest rate	1.40%	1.59%		
Weighted average expected option term	5.75 years	5.25 years		
Weighted average fair value of options granted	\$ 5.45 \$	4.17		

Stock Purchase Rights

<u>Disclosure Of Share-Based Compensation</u> <u>Arrangements By Share-Based Payment</u> <u>Award</u> Stock Purchase Right activity for the three months ended June 30, 2016 is summarized as follows:

Weighted	Weighted
Average	Average
Grant-	Domoining
Date	Remaining

Three Months Ended

Weighted

	Number of	Fair Market	Contractual
	Shares	Value	Term (in Years)
Balance at March 31, 2016	82,171	6.30	0.76
Granted	-	-	
Vested	(9,475)	6.23	
Forfeited		-	
Balance at June 30, 2016	72,696	6.31	0.45

Restricted Stock Units

Disclosure Of Share-Based Compensation Arrangements By Share-Based Payment Award Restricted Stock Unit activity for the three months ended June 30, 2016 is summarized as follows:

		Weighted Average	Weighted Average Remaining
	Number of	Grant Date	Contractual
	Shares	 Fair Value	Term (in Years)
Balance at March 31, 2016	4,544,799	\$ 8.08	1.67
Granted	300,942	11.40	
Vested	(311,949)	8.02	
Forfeited	(101,193)	8.48	
Balance at June 30, 2016	4,432,599	\$ 8.32	1.53

Stock Options Outstanding And Exercisable (Tables)

Stock Options Outstanding
And Exercisable Tables
Summary Of Outstanding And
Exercisable Stock Options

3 Months Ended Jun. 30, 2016

The following table summarizes stock options outstanding and exercisable at June 30, 2016:

	Options Outstanding					Op	ti	ons Exerc	isable
	Shares	1	Average	Weighted Average Remaining Contractual Life (Years)		Shares		Weighted Average Exercise Price Per Share	Aggregate Intrinsic Value
\$0.55 to \$1.27	933,625	\$	1.12	1.5	\$12,596,169	933,625	\$	1.12	\$12,596,169
\$1.39 to \$6.86 \$7.52	1,477,028	\$	5.83	6.8	12,968,088	1,050,590	\$	5.43	9,643,718
to \$8.93 \$9.21	973,650	\$	8.28	8.8	6,165,143	271,615	\$	8.22	1,735,944
to \$9.74 \$10.50	983,835	\$	9.62	7.2	4,905,183	666,290	\$	9.63	3,319,570
to \$11.31	287,248	\$	11.02	8.5	1,031,611 \$37,666,194		\$	11.10	323,576 \$27,618,977

Net Income (Loss) Per Share (Tables)

Net Income Loss Per Share Tables

Net Income (Loss) Per Share

3 Months Ended Jun. 30, 2016

The following is a reconciliation of the weighted average number of common shares outstanding used in calculating basic and diluted net income (loss) per share (in thousands, except share and per share data):

,	Three Months Ended June 30,			
	_	2016		2015
Numerator:				
Net loss available to common stockholders	\$	(528)	\$	(472)
Denominator:				
Common shares	_	89,434	_	88,233
Denominator for basic calculation		89,434		88,233
Employee stock options		-		-
Stock purchase rights		-		-
Denominator for diluted calculation	_	89,434	_	88,233
Net loss per share				
Basic	\$	(0.01)	\$	(0.01)
Diluted	\$	(0.01)	\$	(0.01)

Excluded from Computation of Earnings Per Share

Schedule of Antidilutive Securities The following shares attributable to outstanding stock options and restricted stock purchase rights were excluded from the calculation of diluted earnings per share because their inclusion would have been antidilutive (in thousands):

	Three Months Ended June 30,		
	2016	2015	
Employee stock options	1,528	2,447	
Stock purchase rights	1,602	70	
Total anti-dilutive employee stock-based securities	3,130	2,517	

Segment Information (Tables)

3 Months Ended Jun. 30, 2016

Segment Information Tables

Schedule of Revenue from External Customers

The Company's revenue distribution by geographic region (based upon the and Long-Lived Assets, by Geographical Areas destination of shipments and the customer's service address) was as follows:

,	Three	Months	Ended
		June 30	

	June 3	υ,
	2016	2015
Americas (principally US)	88%	88%
Europe	11%	9%
Asia Pacific	1%	3%
	100%	100%

Geographic area data is based upon the location of the property and equipment and is as follows (in thousands):

	June 30, 2016		March 31, 2016
Americas (principally US)	\$ 9,643	\$	9,165
Europe	3,126		2,642
Asia-Pacific	 246	_	568
Total	\$ 13,015	\$	12,375

Operating Segment Information (Tables)

Operating Segment Information Tables

Schedule of Segment Reporting Information, by Segment

3 Months Ended Jun. 30, 2016

The following table provides financial information by segment for the three month period ending June 30, 2016 and 2015 (in thousands):

Three Months Ended June 30,

	5 time 2 5,			
		2016		2015
Americas (principally US):				
Net revenue	\$	53,398	\$	43,588
Net income	\$	1,466	\$	251
Europe:				
Net revenue	\$	6,643	\$	4,304
Net loss	\$	(1,994)	\$	(723)

Description of the Business (Narrative) (Details)

3 Months Ended Jun. 30, 2016

Description Of Business Narrative Details

Fiscal Year End Date

--03-31

FAIR VALUE MEASUREMENTS (Details)

- USD (\$)

Jun. 30, 2016 Mar. 31, 2016 Jun. 30, 2015 Mar. 31, 2015

•	•		1
•	ın	Thousan	ИC
r D		i iivusaii	us

\$ III Thousands				
Amortized Costs	\$ 167,085	\$ 163,005		
Gross Unrealized Gains	189	89		
Gross Unrealized Loss	(201)	(244)		
Estimated Fair Value	167,073	162,850		
Cash and cash equivalents	27,466	33,576	\$ 29,298	\$ 53,110
Short-term marketable investments	139,607	129,274		
Liabilities, Fair Value Disclosure	216	341		\$ 0
Aavailable-for-sale investments due within one yea	<u>r</u> 79,772			
Aavailable-for-sale investments due after one year	59,835			
Level 1				
Amortized Costs	29,466	35,576		
Gross Unrealized Gains	0	0		
Gross Unrealized Loss	(175)	(187)		
Estimated Fair Value	29,291	35,389		
Cash and cash equivalents	27,466	33,576		
Short-term marketable investments	1,825	1,813		
Level 2				
Amortized Costs	137,619	127,429		
Gross Unrealized Gains	189	89		
Gross Unrealized Loss	(26)	(57)		
Estimated Fair Value	137,782	127,461		
Cash and cash equivalents	0	0		
Short-term marketable investments	137,782	127,461		
Level 3				
Amortized Costs	0	0		
Gross Unrealized Gains	0	0		
Gross Unrealized Loss	0	0		
Estimated Fair Value	216	341		
Cash and cash equivalents	0	0		
Short-term marketable investments	0	0		
Liabilities, Fair Value Disclosure	216	341		
Cash				
Amortized Costs	22,440	18,596		
Gross Unrealized Gains	0	0		
Gross Unrealized Loss	0	0		
Estimated Fair Value	22,440	18,596		
Cash and cash equivalents	22,440	18,596		
Short-term marketable investments	0	0		
Money Market Funds Level 1				
Amortized Costs	5,026	14,980		

Gross Unrealized Gains	0	0
Gross Unrealized Loss	0	0
Estimated Fair Value	5,026	14,980
Cash and cash equivalents	5,026	14,980
Short-term marketable investments	0	0
Mutual Funds Level 1		
Amortized Costs	2,000	2,000
Gross Unrealized Gains	0	0
Gross Unrealized Loss	(175)	(187)
Estimated Fair Value	1,825	1,813
Cash and cash equivalents	0	0
Short-term marketable investments	1,825	1,813
Commercial Paper Level 2		
Amortized Costs	15,660	6,794
Gross Unrealized Gains	8	2
Gross Unrealized Loss	0	0
Estimated Fair Value	15,668	6,796
Cash and cash equivalents	0	0
Short-term marketable investments	15,668	6,796
Corporate Debt Level 2		
Amortized Costs	88,620	85,164
Gross Unrealized Gains	143	78
Gross Unrealized Loss	(10)	(28)
Estimated Fair Value	88,753	85,214
Cash and cash equivalents	0	0
Short-term marketable investments	88,753	85,214
Municipal Securities Level 2		
Amortized Costs	1,001	1,007
Gross Unrealized Gains	0	0
Gross Unrealized Loss	(1)	(1)
Estimated Fair Value	1,000	1,006
Cash and cash equivalents	0	0
Short-term marketable investments	1,000	1,006
Asset-backed Securities Level 2		
Amortized Costs	27,349	24,614
Gross Unrealized Gains	32	7
Gross Unrealized Loss	(1)	(11)
Estimated Fair Value	27,380	24,610
Cash and cash equivalents	0	0
Short-term marketable investments	27,380	24,610
Mortgage backed Securities Level 2		
Amortized Costs	1,989	2,045
Gross Unrealized Gains	0	0
Gross Unrealized Loss	(14)	(17)

Estimated Fair Value	1,975	2,028
Cash and cash equivalents	0	0
Short-term marketable investments	1,975	2,028
Agency Bond Level 2		
Amortized Costs	2,000	6,805
Gross Unrealized Gains	5	1
Gross Unrealized Loss	0	0
Estimated Fair Value	2,005	6,806
Cash and cash equivalents	0	0
Short-term marketable investments	2,005	6,806
International Government Securities Level 2		
Amortized Costs	1,000	1,000
Gross Unrealized Gains	1	1
Gross Unrealized Loss	0	0
Estimated Fair Value	1,001	1,001
Cash and cash equivalents	0	0
Short-term marketable investments	1,001	1,001
Contingent Consideration Level 3		
Amortized Costs	0	0
Gross Unrealized Gains	0	0
Gross Unrealized Loss	0	0
Estimated Fair Value	216	341
Cash and cash equivalents	0	0
Short-term marketable investments	0	0
Liabilities, Fair Value Disclosure	\$ 216	\$ 341

Fair Value Measurements	3 Months Ended 12 Months En		
(Contingent Consideration			
and Escrow Liability)	Inn 20 2016	May 21 2016	
(Details) - USD (\$)	Jun. 30, 2016	Mar. 31, 2016	
\$ in Thousands			
Contingent consideration liabilities [Roll Forwar	<u>rd]</u>		
Contingent considerations, beginning of period	\$ 341	\$ 0	
Purchase price contigent consideration	0	541	
Contingent consideration payments	(125)	(200)	
Contingent consideration, end of period	\$ 216	\$ 341	

Inventories (Details) - USD (\$) S in Thousands Inventories Details Work-in-process Finished goods Inventory, net \$ 460 444 Inventory, net \$ 463 \$ 520

Intangible Assets Schedule Of Intangibles (Details) -Jun. 30, 2016 Mar. 31, 2016 **USD (\$)** \$ in Thousands **Gross Carrying Amount** \$ 29,961 \$ 30,933 Accumulated Amortization (10,430)(9,469)**Net Carrying Amount** 19,531 21,464 **Technology Gross Carrying Amount** 18,033 18,640 **Accumulated Amortization** (5,215)(4,622)**Net Carrying Amount** 12,818 14,018 Customer relationships 9.993 **Gross Carrying Amount** 9,716 Accumulated Amortization (5,215)(4,847)4,501 **Net Carrying Amount** 5,146 Trade names/domains **Gross Carrying Amount** 2,117 2,205 **Accumulated Amortization** 0 2,117 2,205 **Net Carrying Amount** In-process R&D 95 95 **Gross Carrying Amount Accumulated Amortization** 0 0 \$ 95 \$ 95 **Net Carrying Amount**

Intangible Assets Schedule Of Future Amortization Of Intangibles (Details) \$ in Thousands

Jun. 30, 2016 USD (\$)

Intangible Assets Schedule Of Future Amortization Of Intangibles Details

Remaining 2017	\$ 2,758
2018	3,411
2019	3,162
<u>2020</u>	3,162
<u>2021</u>	2,796
Thereafter	2,030
<u>Total</u>	\$ 17,319

Changes in Carrying	3 Months Ended	
Amount of Goodwill by Location (Detail) \$ in Thousands	Jun. 30, 2016 USD (\$)	
Goodwill, beginning balance	\$ 47,420	
Foreign currency translation	(1,489)	
Goodwill, ending balance	45,931	
Americas		
Goodwill, beginning balance	25,729	
Foreign currency translation	0	
Goodwill, ending balance	25,729	
Europe		
Goodwill, beginning balance	21,691	
Foreign currency translation	(1,489)	
Goodwill, ending balance	\$ 20,202	

Software Development Costs (Narrative) (Details) Software Development -

3 Months Ended

Software Development - USD (\$)

Jun. 30, 2016 Jun. 30, 2015 Mar. 31, 2016

\$ in Millions

Software costs capitalized during the period	\$ 1.0	\$ 0.5	
Total capitalized software costs	4.7		
Accumulated software costs amortization	0.4		\$ 0.2
Property and equipment	1.5		1.2
Long-term assets charged to other assets	\$ 3.2		\$ 2.5

Commitments and Contingencies (Product Varranties) (Details) - USD

Warranties) (Details) - USD (\$)

Jun. 30, 2016 Jun. 30, 2015

3 Months Ended

\$ in Thousands

Balance at beginning of period	\$ 326	\$ 339
Accruals for warranties	114	98
Settlements	(87)	(83)
<u>Adjustments</u>	(13)	(12)
Balance at end of period	\$ 340	\$ 342

Commitments and Contingencies (Service Provider Contracts) (Details) \$ in Thousands

Jun. 30, 2016 USD (\$)

Commitments And Contingencies Service Provider Contracts Details

Remaining 2017	\$ 1,961
<u>2018</u>	981
2019	32
<u>2020</u>	8
Total minimum payments	\$ 2,982

Commitments and Contingencies

(CustomerSupport

Commitments) (Narrative)

Jun. 30, 2016 USD (\$)

(Details)
\$ in Millions

Third party customer support vendor minimum monthly commitment	\$ 0.4
Third party customer support vendor maximum obligation	\$ 2.2
Advance termination notice required, days	150
Property and equipment under capital lease	\$ 1.6
Accumulated depreciation related to assets under capital lease	\$ 0.6

Stock-based Compensation Stock-Based Compensation	3 Montl	hs Ended
Expense By Statement Of Operations (Details) - USD (\$) \$ in Thousands	Jun. 30, 2016	Jun. 30, 2015
Stock-based employee compensation expense related to employee stock options and	\$ 5,051	\$ 3,022
employee stock purchases, pre tax	0	
Tax benefit	0	0
Stock-based employee compensation expense related to employee stock options and	5,051	3,022
employee stock purchases, net of tax	,	
Cost of service revenue		
Stock-based employee compensation expense related to employee stock options and	360	219
employee stock purchases, pre tax		
<u>Cost of product revenue</u>		
Stock-based employee compensation expense related to employee stock options and	0	0
employee stock purchases, pre tax	V	O
Research and development		
Stock-based employee compensation expense related to employee stock options and	887	531
employee stock purchases, pre tax	007	331
Sales and marketing		
Stock-based employee compensation expense related to employee stock options and	1,915	1,197
employee stock purchases, pre tax	1,913	1,197
General and administrative		
Stock-based employee compensation expense related to employee stock options and	¢ 1 000	¢ 1 075
employee stock purchases, pre tax	\$ 1,889	\$ 1,075

Stock-based Compensation Option Activity (Details)	3 Months Ended Jun. 30, 2016 \$ / shares shares
Stock-based Compensation Option Activity Details	
Balance at beginning of period shares	4,793,266
Granted shares	53,976
Exercised shares	(191,855)
Cancelled/forfeited shares	(1)
Balance at end of period shares	4,655,386
Options, Vested and expected to vest shares	4,655,386
Options, Exercisable at end of period shares	3,014,307
Weighted-average exercise price of options outstanding, at beginning of period \$ / shares	\$ 6.29
Weighted-average exercise price of options granted during period \$ / shares	11.31
Weighted-average exercise price of options exercised during the period \$ / shares	2.18
Weighted-average exercise price of options forfeited, cancelled or expired during the period \$ shares	2.58
Weighted-average exercise price of options outstanding at end of period \$ / shares	6.52
Options, Vested and Expected to Vest, Weighted Average Exercise Price \$ / shares	6.52
Weighted-Average Exercise Prices, Exercisable at end of period \$ / shares	\$ 5.45

Stock-based Compensation Stock Purchase Right Activity (Details)	3 Months Ended Jun. 30, 2016 \$ / shares shares
Stock-based Compensation Stock Purchase Right Activity Details	
Balance at beginning of period shares	82,171
Granted shares	0
<u>Vested shares</u>	(9,475)
Forfeited shares	0
Balance at end of period shares	72,696
Weighted-average grant date fair market value, beginning balance \$ / shares	\$ 6.30
Weighted-average grant date fair market value of restricted stock rights granted \$ / shar	<u>es</u> 0
Weighted-average grant date fair market value, vested during period \$ / shares	6.23
Weighted-average grant date fair market value, forfeited during period \$ / shares	0
Weighted-average grant date fair market value, ending balance \$ / shares	\$ 6.31
Weighted-average remaining contractual term, in years, ending balance	162 days

3 Months Ended **Stock-based Compensation** Jun. 30, 2016 **Restricted Stock Unit** \$ / shares Mar. 31, 2016 **Activity (Details)** shares **Stock-based Compensation Restricted Stock Unit Activity Details** Balance at beginning of period | shares 4,544,799 Granted | shares 300,942 Vested | shares (311,949)Forfeited | shares (101,193)Balance at end of period | shares 4,432,599 RSU weighted-average remaining contractual term, in years 1.53 1.67 Beginning of period, weighted-average grant date fair value | \$ / shares \$ 8.08 Granted, weighted-average grant date fair value | \$ / shares 11.40 Vested, weighted-average grant date fair value | \$ / shares 8.02 Forfeited, weighted-average grant date fair value | \$ / shares 8.48 End of period, weighted-average grant date fair value | \$ / shares \$ 8.32

Stock-based Compenstaion Stock Options Outstanding	3 Months Ended	
And Exercisable (Details) - USD (\$)	Jun. 30, 2016	Mar. 31, 2016
Options Outstanding, Number of Shares	4,655,386	4,793,266
Options Outstanding, Weighted-Average Exercise Price Per Share	\$ 6.52	\$ 6.29
Options Exercisable, Number of Shares	3,014,307	•
Options Exercisable, Weighted-Average Exercise Price Per Share	\$ 5.45	
\$0.55 - \$1.27	+	
Range of Exercise Prices, Minimum	0.55	
Range of Exercise Prices, Maximum	\$ 1.27	
Options Outstanding, Number of Shares	933,625	
Options Outstanding, Weighted-Average Exercise Price Per Share	\$ 1.12	
Options Outstanding, Weighted-Average Remaining Contractual Term (in years	1 year 180 days	
Options Outstanding, Aggregate Intrinsic Value	\$ 12,596,169	
Options Exercisable, Number of Shares	933,625	
Options Exercisable, Weighted-Average Exercise Price Per Share	\$ 1.12	
Options Exercisable, Aggregate Intrinsic Value	\$ 12,596,169	
<u>\$1.39 - \$6.86</u>		
Range of Exercise Prices, Minimum	\$ 1.39	
Range of Exercise Prices, Maximum	\$ 6.86	
Options Outstanding, Number of Shares	1,477,028	
Options Outstanding, Weighted-Average Exercise Price Per Share	\$ 5.83	
Options Outstanding, Weighted-Average Remaining Contractual Term (in years	6) 6 years 288 days	
Options Outstanding, Aggregate Intrinsic Value	\$ 12,968,088	
Options Exercisable, Number of Shares	1,050,590	
Options Exercisable, Weighted-Average Exercise Price Per Share	\$ 5.43	
Options Exercisable, Aggregate Intrinsic Value	\$ 9,643,718	
<u>\$7.52 - \$8.93</u>		
Range of Exercise Prices, Minimum	\$ 7.52	
Range of Exercise Prices, Maximum	\$ 8.93	
Options Outstanding, Number of Shares	973,650	
Options Outstanding, Weighted-Average Exercise Price Per Share	\$ 8.28	
Options Outstanding, Weighted-Average Remaining Contractual Term (in years	3) 8 years 288 days	
Options Outstanding, Aggregate Intrinsic Value	\$ 6,165,143	
Options Exercisable, Number of Shares	271,615	
Options Exercisable, Weighted-Average Exercise Price Per Share	\$ 8.22	
Options Exercisable, Aggregate Intrinsic Value	\$ 1,735,944	
<u>\$9.21 - \$9.74</u>		
Range of Exercise Prices, Minimum	\$ 9.21	
Range of Exercise Prices, Maximum	\$ 9.74	
Options Outstanding, Number of Shares	983,835	
Options Outstanding, Weighted-Average Exercise Price Per Share	\$ 9.62	
Options Outstanding, Weighted-Average Remaining Contractual Term (in years	7 years 72 days	

Options Outstanding, Aggregate Intrinsic Value	\$ 4,905,183
Options Exercisable, Number of Shares	666,290
Options Exercisable, Weighted-Average Exercise Price Per Share	\$ 9.63
Options Exercisable, Aggregate Intrinsic Value	\$ 3,319,570
<u>\$10.50 - \$11.31</u>	
Range of Exercise Prices, Minimum	\$ 10.50
Range of Exercise Prices, Maximum	\$ 11.31
Options Outstanding, Number of Shares	287,248
Options Outstanding, Weighted-Average Exercise Price Per Share	\$ 11.02
Options Outstanding, Weighted-Average Remaining Contractual Term (in years	8 years 180 days
Options Outstanding, Aggregate Intrinsic Value	\$ 1,031,611
Options Exercisable, Number of Shares	92,187
Options Exercisable, Weighted-Average Exercise Price Per Share	\$ 11.10
Options Exercisable, Aggregate Intrinsic Value	\$ 323,576
<u>\$0.55 - \$11.31</u>	
Range of Exercise Prices, Minimum	\$ 0.55
Range of Exercise Prices, Maximum	\$ 11.31
Options Outstanding, Number of Shares	4,655,386
Options Outstanding, Aggregate Intrinsic Value	\$ 37,666,194
Options Exercisable, Number of Shares	3,014,307
Options Exercisable, Aggregate Intrinsic Value	\$ 27,618,977

Stock-based Compensation Assumptions Used In Black-	3 Months Ended		
Scholes Model (Details) - Option Grants - \$ / shares	Jun. 30, 2016	Jun. 30, 2015	
Expected volatility	51.00%	53.00%	
Expected dividend yield	0.00%	0.00%	
Risk-free interest rate	1.40%	1.59%	
Weighted average expected option term, in years	5 years 9 months	s 5 years 3 months	
Weighted average fair value of options granted, per share	\$ 5.45	\$ 4.17	

Stock-based Compensation (Narrative) (Details) \$ in Millions

3 Months Ended Jun. 30, 2016 USD (\$)

Stock-Based Awards

<u>Unamortized stock-based compensation expense related to unvested stock awards</u> \$ 32.9

Weighted average period of recognition for unrecognized compensation costs (in years) 2 years 80 days

As Part of Acquisition

<u>Unamortized stock-based compensation expense related to unvested stock awards</u> \$ 1.8

Weighted average period of recognition for unrecognized compensation costs (in years) 2 years 331 days

Stock Repurchases (Detail) - 3 Months		Ended	
USD (\$)	Jun. 30, 2016	Mar. 31, 2016	
February 2015			
Stock repurchased and retired during period, shares	0		
Stock repurchased and retired during period, value	[1] \$ 0		
Stock repurchased, weighted average price per share	\$ 0		
Stock repurchase program, authorized amount		\$ 20,000,000	
Stock repurchase program expiration date	Oct. 20, 2016		
October 2015			
Stock repurchase program, authorized amount		15,000,000	
Total Plan 2015			
Stock repurchase program, remaining authorized repurchase amount		\$ 15,000,000	

[1] Amount excludes commission fees.

Income Taxes (Narrative) 3 Months Ended (Details) Jun. 30, 2016 \$ in Millions USD (\$)

Unrecognized tax benefits \$ 2.9

Federal

Tax years open for examination 2013

State

Tax years open for examination 2013

Foreign

Tax years open for examination 2011

Net Income (Loss) Per Share (Details) - USD (\$)

3 Months Ended

\$ / shares in Units, shares in Jun. 30, 2016 Jun. 30, 2015 Thousands

Numerator:

Net loss available to common stockholde	<u>rs</u> \$ (528)	\$ (472)
Denominator:		
Common shares	89,434	88,233
Denominator for basic calculation	89,434	88,233
Employee stock options	0	0
Stock purchase rights	0	0
Denominator for diluted calculation	89,434	88,233
Net loss per share:		
Basic	\$ (0.01)	\$ (0.01)
<u>Diluted</u>	\$ (0.01)	\$ (0.01)

Net Income (Loss) Per Share (Options and Rights	3 Months Ended	
Excluded) (Details) - shares	Jun. 30, 2016	Jun. 30, 2015
shares in Thousands	ŕ	ŕ
Anti-dilutive shares	3,130	2,517
Employee stock options		
Anti-dilutive shares	1,528	2,447
Stock purchase rights		
Anti-dilutive shares	1,602	70

Segment Reporting Revenue and Property and Equipment by Geographic Area (Details) - USD (\$)

\$ in Thousands

3 Months Ended

Jun. 30, 2016 Jun. 30, 2015 Mar. 31, 2016

<u>s</u> 100%	100%	
0	0	
\$ 13,015		\$ 12,375
<u>s</u> 88%	88%	
\$ 9,643		9,165
<u>s</u> 11%	9%	
\$ 3,126		2,642
<u>s</u> 1%	3%	
\$ 246		\$ 568
	\$ 13,015 \$ 88% \$ 9,643 \$ 3,126	0 0 0 \$ 13,015 88% \$ 89,643 9% \$ 3,126 \$ 3%

Operating Segment (Details)	3 Months Ended		
- USD (\$) \$ in Thousands	Jun. 30, 2016 Jun. 30, 2015		
Net revenue	\$ 60,041	\$ 47,892	
Net income (loss)	(528)	(472)	
Americas			
Net revenue	53,398	43,588	
Net income (loss)	1,466	251	
<u>Europe</u>			
Net revenue	6,643	4,304	
Net income (loss)	\$ (1,994)	\$ (723)	